IDENTIFYING FRAILTY IN CANCER SURVIVORS: PATTERNS OF CANCER FOLLOW-UP CARE AND IMPLICATIONS FOR PERSONALIZED SURVIVORSHIP MODELS

by

Sarah Marie Murnaghan

Submitted in partial fulfilment of the requirements for the degree of Master of Science

at

Dalhousie University Halifax, Nova Scotia January 2023

Dalhousie University is located in Mi'kma'ki, the ancestral and unceded territory of the Mi'kmaq. We are all Treaty people.

© Copyright by Sarah Marie Murnaghan, 2023

Dedication

For the patients and families who touched my heart and opened my eyes to the reasons we need to strive for better.

For grammie and grampy Dickieson, my love for you both lit fire to a larger passion.

Also, for my little brother-the future is bright.

List of Tables vi
List of Figuresvii
Abstractviii
List of Abbreviations and Symbols Usedix
Acknowledgements xii
Chapter 1: Introduction1
Chapter 2: Background5
2.1 Cancer Incidence and Survivorship5
2.1.1 Guidelines for Follow-up Care
2.2 Aging and Frailty14
2.3 Measures of Frailty15
 2.3.1 Phenotypic Approach to Frailty
Care 21
2.5 Identifying Frailty in Cancer Survivors23
2.6 Future Models of Care are Personalized27
Chapter 3: Objectives
Chapter 4: Methods
4.1 Data and Study Population32
4.2 Variables
4.2.1 Key Measures
4.3 Analysis

Table of Contents

4.3.1 Objective 1 Analysis	
4.5.2 Objective 2 Analysis	
4.4 Data Access and Ethics	
Chapter 5: Results	
5.1 Participants Eligible for Analyses	
5.2 Objective 1 Results	
5.2.1 Frailty Prevalence	
5.2.3 Frailty Rules	
5.3 Objective 2 Results	
5.3.1 Total Cancer-related Follow-up Care	
5.5.2 Proportion of Cancer-related FCF Follow-up Care	
5.4 Tables	
5.4.1 Frailty Tables 5.4.2 Health Care Use Tables	
Chapter 6: Discussion	
6.1 Comparing Findings to Literature	
6.2 Limitations	
6.3 Strengths, Research and Policy Implications	
6.4 Future Research Directions	
6.5 Conclusion	
6.6 Knowledge Translation	
References	
Appendix A: Follow-up Guideline Chart	
Appendix B: MSI Physician Billings/CIHI DAD Censoring Proced	ure Codes 118
Appendix C: MSI Physician Billings Diagnostic Codes Cancer-rela	ated Follow-up
Visits	

Appendix D: Distribution of the Total Visit Count for the Cancer Survivor	
Cohort	27
Appendix E: Distribution of the Proportion of PCP Visits (Continuous) for the Cancer Survivor Cohort	28

List of Tables

Table 1. Prevalence of frailty in the cancer survivor cohort by patient characteristics (Objective 1.1)
Table 2. Patient characteristics amongst the cancer survivor cohort (<i>Objective 1.2</i>)64
Table 3. Logistic regression model for the association between patient characteristics and frailty (<i>Objective 1.2</i>)
Table 4. Frailty identification amongst cancer survivors by specific rule (<i>Objective 1.3</i>)
Table 5. Median number of total cancer-related follow-up visits during the follow-up period by patient characteristics (<i>Objective 2.1</i>)
Table 6. Negative binomial (GLM) regression for the association between patient characteristics and the annual rate of cancer-related follow-up visits (<i>Objective 2.1</i>)
Table 7. Provider type seen by all, non-frail, and frail cancer survivors during the follow-up period (<i>Objective2.2</i>)
Table 8. The percentage of cancer survivors' cancer-related follow-up care attributable to PCPs (<i>Objective 2.2</i>)
Table 9. Univariate multinomial models for the association between patientcharacteristics and the proportion PCP visits (low, medium, high) (Objective2.2)
Table 10. Multivariate partial proportional odds for the association between patient characteristics and the proportion of PCP visits (low, medium, high) (Objective 2.2)

List of Figures

Figure 1: Study timeline	34
Figure 2: Flow chart of the cancer survivor cohort used for analyses	53

Abstract

Improved cancer care has increased the number of survivors, straining Canada's cancer care system. As survivors age, they may become frail, resulting in complex needs better addressed through alternative care models. No research has examined health services use in frail cancer survivors or quantified frailty within a Canadian cancer survivor population.

This thesis estimated the burden of frailty amongst Nova Scotia cancer survivors diagnosed with stage I-III breast, colorectal, gynecologic, or prostate cancer between Jan 2006-Dec 2013, from the provincial Cancer Registry (n=10,176). Then, it characterized differences in survivors' patterns of follow-up care by frailty and other characteristics.

The prevalence of frailty amongst the survivor population was 17.7%. Compared to nonfrail cancer survivors, frail survivors had 58% greater odds of having a high (versus lowmedium) proportion of primary care provider visits (Odds Ratio 1.58, 95% CI 1.43-1.76). Future research should investigate how primary care models can best support frail survivors.

List of Abbreviations and Symbols Used

ADLs	Activities of Daily Living
AIC	Akaike's Information Criteria
APN	Advanced Practice Nurse
AYA	Adolescents and Young Adults with Cancer
AB	Alberta
BIC	Bayesian Information Criteria
BMI	Body Mass Index
BC	British Columbia
CIHI DAD	Canadian Institute for Health Information Discharge Abstracts Database
CA	Census Agglomeration
PCCF+	Census-Based Small Area Data/Postal Code Conversion File
СМА	Census Metropolitan Area
CFS	Clinical Frailty Scale
MIZ	CMA/CA Influenced Zone
CI	Confidence Interval
EFS	Edmonton Frail Scale
FRAIL	Fatigue, Resistance, Ambulation, Illness, Loss of weight (a measurement of
	frailty)
GLM	Generalized Linear Modelling
HR	Hazard Ratio
HDNS	Health Data Nova Scotia
IOM	Institute of Medicine

- IPR Insured Patient Registry
- IQR Interquartile Range
- IRR Incidence Rate Ratio
- ICD-9/10 International Classification of Diseases, Ninth and Tenth Revisions
- LPR Licensed Provider Registry
- LR Likelihood-Ratio test
- MB Manitoba
- MSI Medical Service Insurance
- NS Nova Scotia
- NSCR Nova Scotia Cancer Registry
- NSH Nova Scotia Health
- OR Odds Ratio
- OPIS Oncology Patient Information System
- ON Ontario
- ICES Ontario's Institute for Clinical Evaluative Sciences
- PCP Primary care provider (in the context of this study, PCP refers to either a family physician, general, or community medicine practitioner)
- QC Quebec
- REB Research Ethics Board
- SES Socioeconomic status
- SD Standard Deviation
- SACtype Statistical Area Classification

- TCPS 2
 Tri-Council Policy Statement Ethical Conduct for Research Involving

 Humans
 US

 US
 United States
- Nd= λR Little's Law
- χ^2 Chi-squared

Acknowledgements

To my supervisor, Dr. Robin Urquhart, you are a shining example of strength and kindness. Your gift for helping people find solutions to the problems that matter to them is inspiring. Thank you for your support throughout this non-linear journey. The numerous research opportunities you trusted me with the past few years have been invaluable to my Master of Science experience.

I would also like to express appreciation to my committee members who helped cultivate meaning in this project. To Dr. George Kephart, thank you for your patience and plenty of hours spent guiding me to think critically about this project; I watched this project blossom more after each conversation. To Dr. Ravi Ramjeesingh, I am grateful for your time, expertise, and positive attitude dedicated to this project.

Thank you to Cynthia and Margaret for guiding me through my first experience using administrative data. Your knowledge made the administrative processes much smoother. Thank you also to Devbani and Lynn for taking my sometimes lengthy and semi-formed questions about handling administrative data. Asking these questions was a crucial part of my growth, as well as that of the project; I appreciate you allowing me the space to do so. To the staff at Health Data Nova Scotia thank you for your patience as we worked together through many project amendments. This research is based upon data made available by Health Data Nova Scotia. Although this research is based on data obtained from the Nova Scotia Department of Health and Wellness, the observations and opinions

xii

expressed in this thesis are those of the authors and do not represent either Health Data Nova Scotia or the Department of Health and Wellness.

I entered the Department of Community Health and Epidemiology program to further my education, expand my skills, and obtain a degree. I gained much more than that. Michelle, your friendship and support throughout our time in the program and beyond is the surprising gift I gained; Thank you. To my family and friends, especially my parents and my brother (Tyler), thank you for your love and encouragement. To my husband (Kyle), your support and belief in me are unmatched.

I would not be at this point in my career, merging my clinical background with my interest in research if it were not for a few very special nurses. Stacey, Tracy, and Melissa, I met each of you at different points in my nursing career, but each of you inspired, taught, and encouraged me along the way. I will continue to take the lessons you have given me in my future endeavours.

Finally, I would like to thank my funding sources: The Canadian Centre for Applied Research in Cancer Control (Studentship), the Canadian Institutes of Health Research (Frederick Banting and Charles Best Canada Graduate Scholarship-Master's), the Maritime SPOR SUPPORT Unit (Master's Studentship), and the Province of Nova Scotia (Nova Scotia Graduate Scholarship). Supporting graduate work is crucial to the future of research, and I am honoured you funded my project; it has helped me greatly. Thank you also to the Nova Scotia Research Fund for funding the linkage of the original dataset.

xiii

Chapter 1: Introduction

Improved cancer screening and treatments have contributed to increased cancer survival (1). The aging population is a driving force for increased cancer incidence (2,3). Of those newly diagnosed, about 65% will become long-term survivors (3,4). Many cancer survivors experience lingering health issues due to their cancer and/or its treatment and require ongoing follow-up care (5). The Institute of Medicine (IOM) recommends cancer survivors receive follow-up care appointments with their health care provider to monitor for cancer recurrence and address the late and long-term effects of cancer treatment (6,7). Follow-up care in Canada has been largely oncologist-led despite evidence that alternative models of follow-up care (e.g., primary care-led, shared-care, or nurse-led) are safe and effective (8–11). Evidence for alternative care models largely comes from studies amongst survivors of breast, colorectal, or prostate cancer (8,9,12). There is a lack of evidence and consensus, particularly for survivors of other (e.g., gynecologic) and/or rare cancers (e.g., esophageal, soft tissue) regarding the appropriate follow-up pathway. Current cancer systems struggle to meet the increased demand for cancer services and deliver optimal care for newly diagnosed cancer patients and survivors (13).

Survivorship care is becoming increasingly complex as survivors are living longer, often have comorbid illness, and experience lingering health effects from cancer and its treatment. As Canada's population ages, frailty will become more common – both in the general population and in those who have survived cancer. Frailty is a vulnerable health state that leaves individuals susceptible to adverse health outcomes and reduced physical resiliency (14). Frail cancer survivors may have higher or more complex needs after

cancer treatment. Studied outside of cancer care, research has shown that health care often inadequately addresses the needs of frail people. Further, alternative care models (e.g., multidisciplinary) in other areas of care have enhanced the care of frail people (15,16).

Current data suggests our traditional models of follow-up care are not working for many survivors: cancer survivors are using more health care resources, yet there is no evidence that their needs are any better met (17,18). The current system must change as it neither meets survivors' needs nor is sustainable from a health system perspective (1). Alternative follow-up care models have been studied, prompting a recent shift to more personalized models of follow-up care. Personalized models consider the heterogeneity amongst cancer survivors by identifying subgroups best served by primary care and others that require ongoing care by their oncology teams (5,6,19,20). Thus, work must be done to better understand survivors' needs and design follow-up care models tailored to those needs. To our knowledge, no research has examined health services utilization in frail cancer survivors or studied the prevalence of frailty within a Canadian cancer survivor population.

This thesis builds upon the body of research aimed at designing more personalized models of care for cancer survivors. Chapter 2 begins with a review of the literature on cancer survivorship and frailty and concludes by identifying the knowledge gap and rationale of this thesis. Chapter 3 describes this study's objectives. The study had two main objectives. The first focused on frailty, and the second focused on health care

utilization. The two overarching objectives were first, to estimate the burden of frailty amongst NS cancer survivors and determine how frailty differs by survivor characteristics. Secondly, to identify patterns of cancer-related follow-up visits and how they differ between non-frail and frail cancer survivors, and patient characteristics.

Chapter 4 describes the methods used in this retrospective analysis of a population-based cohort. This research used a linked administrative dataset containing all those in Nova Scotia (NS) diagnosed with stage I-III breast, colorectal, gynecologic, or prostate cancer between January 01, 2006, and December 31, 2013 (n=20,901). The cohort was identified from the Nova Scotia Cancer Registry (NSCR) and linked, at the patient level, to various Health Data Nova Scotia (HDNS) databases. A survivorship cohort was identified (n=10,176). The follow-up period began 366 days after the index cancer diagnosis (i.e., the cancer diagnosed between 2006-2013) and ended on December 31, 2018. Frailty was assessed using decision rules to identify frail persons from two administrative health databases: Canadian Institute for Health Information Discharge Abstracts Database (CIHI DAD) and Medical Service Insurance (MSI) Physician Billings (14,21). The prevalence of frailty within the survivor cohort was determined. Descriptive statistics compared frail and non-frail cancer survivor characteristics. Logistic regression was used to describe variation in frailty prevalence by patient characteristics. For the study's second objective we described patterns of cancer survivor follow-up care (i.e., visit amounts and to whom) and differences between characteristics (i.e., clinical, demographic, and other characteristics). For example, we used numbers and percentages to detail visits by provider type (i.e., total, primary care, oncology, surgery) and amount. Negative binomial

regression was used to model the annual rate of all cancer-related physician visits during follow-up. Ordinal logistic regression was used to describe the proportion of all follow-up visits that were provided by primary care providers (PCPs) and how this differed by frailty and patient characteristics.

Chapter 5 includes the results of the study's analyses. Chapter 6 discusses the context of the study's results, their relation to the current body of literature, the study's strengths, limitations, and future implications. This research addresses an important gap in our understanding of health care use by a vulnerable segment of cancer survivors, which has not yet been studied, and thereby lays a foundation for designing more personalized models of follow-up care. Such knowledge provides a basis for future studies to design and test models of care to meet this subpopulation's needs best.

Chapter 2: Background

2.1 Cancer Incidence and Survivorship

Cancer broadly describes a group of diseases caused by an uncontrolled division of abnormal cells in the body. Cancer can originate in nearly any part of the human body. Cancers are generally referred to by their primary site (e.g., breast, prostate, etc.). Cancer has the propensity to affect many Canadians' lives: one in two Canadians will develop cancer at some point in their lifetime. In 2022, it was expected that 233,900 Canadians were newly diagnosed with cancer (22).

Many risk factors (e.g., age, alcohol consumption, diet, exposure to toxic substances, chronic inflammation, obesity, and tobacco use) contribute to cancer development, and these can vary by cancer (disease) site (23). When adjusted for age and population size, cancer risk has decreased (24). However, driven by the aging population and improved screening, incident cancer cases in the United States (US) are expected to rise to 79% by 2032 (13). Similarly, in Canada, given the growing and aging population, the number of cancer cases continues to increase (22). Age is a non-modifiable risk factor, meaning that health service planners and researchers should pay attention to its impact on cancer incidence as long as life expectancy increases.

From a global perspective, Canada is fortunate to have access to effective screening (e.g., colorectal cancer screening [fecal occult blood or fecal immunochemical testing for average-risk asymptomatic adults 50-74 years every two years], cervical cancer screening [papanicolau testing every three years for sexually active women aged 25-69 years] (25),

and cancer treatments (e.g., immune checkpoint inhibitors) that foster an increased life expectancy and increase cancer survival in its population (7). Canada is a leader in population-based cancer survival (26). Over the past 20 years, cancer survival has improved for most cancers (24). In the 1940s, only 25% of Canadians survived their cancer diagnoses; in 2019, the number of Canadians expected to survive long-term was 63% (3). Traditionally, cancer was considered a death sentence. Today, due to advanced treatments and improved screening, many experts believe patients treated for some types of cancers should be managed as having a chronic disease (27).

What is a cancer survivor? There are various definitions of cancer survivorship. Some consider survivorship to begin upon initial diagnosis, while others consider it to be when there are no more signs of cancer following treatment completion (7,28). Researchers use different definitions to commence monitoring of survivorship, although one-year post-diagnosis is commonly used (29–31). A long-term survivor of cancer is considered an individual five years post-diagnosis (32). Surveillance data on cancer survival in Canada are commonly presented as five-year or 10-year net survival (3,24).

This study examined cancer sites of particularly high survival, including breast, prostate, colorectal, and gynecologic cancers. There is variation in survival by cancer site. Cancers with the highest five- and 10-year net survival include thyroid, testis, and prostate (3,24). Breast, prostate, and colorectal cancer survivors combined constitute nearly half of all cancer survivors (7). The five-year net survival probabilities for breast and prostate are 80% or more, colorectal is between 50-79%, and gynecologic ranges from less than 50%

(ovarian), to between 50-79% (cervical), up to greater than 80% (uterine) (3,24). Considering their survival probabilities, it is a priority to understand and optimize the quality of life after cancer treatment in these populations (24).

After completing cancer treatment, many people describe living in a new normal versus returning to their pre-diagnosis normal. This is mainly due to the late and long-term physical and emotional effects of cancer and its treatment (33). Late effects are consequences of prior cancer treatment that manifest weeks, months (e.g., lymphoedema after axillary node removal, neuropathy after chemotherapy), or years after treatments are complete (e.g., heart disease following radiotherapy to the chest, bone loss or lung problems after chemotherapy and radiation) (7,20,34). Long-term effects include cancer therapy side effects that begin during treatment and continue after treatment completion. Many survivors live well beyond the end of treatment; therefore, late and long-term effects can significantly impact their life after cancer treatment. As a result of more complex cancer treatment combinations, late and long-term effects are becoming more common. Further, survivors may worry about the risk of secondary cancers posttreatment (e.g., breast cancer following radiotherapy for Hodgkin's lymphoma) (35). Cancer screening, diagnosis, and therapy differ by cancer site, and thus the aftermath of cancer depends on cancer site and treatment type. There are physical (e.g., fatigue, chronic pain), emotional (e.g., fear of cancer recurrence, depression, challenges in one's relationships), and practical (e.g., paying for medical bills, return to work) concerns (7). Cancer and its survival can shift relationship dynamics. Some individuals report closer intimate relationships, though this is not the case for all (36,37). The former concern

applies to both spousal relations and other family bonds, such as that between parents and adult children (38). To further describe some physical concerns, survivors may experience late effect changes in cardiovascular, endocrine, lymphatic, gastrointestinal, genitourinary, nervous, hematologic, hepatic, immune, pulmonary, or renal organ systems (7). As a result of these late and long-term effects, cancer survivors are at significant risk for morbidity, diminished quality of life, and premature mortality (5). Each survivor's challenges are uniquely influenced by their stage at diagnosis, treatment, age, genetic factors, comorbidities, and social circumstances (20). Ongoing monitoring is needed to investigate cancer recurrence and address late and long-term effects and their ramifications. This is recognized as follow-up care. The IOM strongly recommends lifelong health care for all survivors following the increased evidence of the late effects of cancer (6).

2.1.1 Guidelines for Follow-up Care

International, national, and provincial recommendations exist to guide follow-up care. Not every Canadian province has published follow-up care guidelines. NS has follow-up surveillance guidelines for breast, pancreatic, colon, and rectal cancers, which endorse follow-up by the PCP, where appropriate in partnership with the surgeon, and encourage active patient engagement in follow-up (39). Other examples include those published by the National Comprehensive Cancer Network (40–49), the American Society of Clinical Oncology (50–54), Alberta Health Services (55–67), Cancer Care Ontario (68–75), and Saskatchewan Cancer Agency (76–83) (Appendix A). Recommendations are unique to each type of cancer in terms of follow-up screening tests, examinations, and the

frequency of follow-up visits. There are also guidelines for special services, such as psychosocial oncology services (68). However, there are common elements all follow-up care should address, including preventing and managing post-treatment late and longterm effects, screening for recurrence and new primary cancers, managing non-cancerrelated pre-existing chronic conditions, and addressing a survivor's preventative health needs (7,31). Follow-up care (i.e., visits) examined in the context of this study were defined as cancer-related oncology, surgery, and PCP follow-up; therefore, visits of noncancer-related nature were excluded. The description of late and long-term effects is relatively new; thus, survivor follow-up care has primarily focused on screening for new primary cancers and monitoring for recurrence (7).

Despite guidelines, differences in follow-up care practice remain. For example, breast cancer survivors in NS have greater surgeon follow-up, while their British Columbia (BC) counterparts have greater primary care follow-up (30). Patterns of health care utilization in survivors across the country have revealed inconsistencies in care; this is likely related to the current models of cancer follow-up care (30,84–86). Health care settings and providers are important components of follow-up guidelines and play a meaningful role in the patient survivorship experience (87). Models of follow-up care (i.e., structure and delivery of follow-up care) are important for the appropriate implementation of survivorship guidelines.

2.1.2 Current Models of Cancer Survivor Follow-up Care

Existing cancer care systems were not designed or resourced considering the needs of the growing population of survivors. There are now more cancer survivors than ever before (29). The increased demand has strained the current cancer care system. In addition to treating active cancer cases, oncologists are treating survivors with late and long-term effects of cancer. As the number of cancer survivors grows, it is increasingly vital to reevaluate cancer care organization and delivery. The traditional survivorship follow-up care model in Canada has seen patients followed by oncologists. This thesis used the overarching terms oncology and oncologist to refer to both the area of practice and practitioners, encompassing medical oncology and radiation oncology. It will be difficult, not to mention costly, for the current system to continue as is. The cost of having patients see specialists such as oncologists rather than PCPs tends to be higher (5,29). Cancer care spending is rising; in Canada, costs rose from 2.9 billion in 2005 to 7.5 billion in 2012 (88). de Oliveira et al. (2018) described this increase in cost as largely due to hospitalbased care of cancer patients, including the receipt of chemotherapy and radiation. They included cancer clinic care and rehabilitation under the umbrella of hospital-based care; both of these sub-category costs also increased over the study period (88). Given the increasing cancer incidence and survival, these costs are likely to continue trending upward.

A 2018 study on medical oncologist workload in high-income countries surveyed 58 Canadian oncologists. They found the Canadian median workload per medical oncologist (175 new consults per year) aligned with Cancer Care Ontario's recommended upper

limit (160-175 new consults per year). In contrast, the median number of new patients per medical oncologist reported in other high-income countries was 125 per annum. Further, Canadian medical oncologists spent more time with each patient per day than their high-income country counterparts (89). It is crucial to consider the potential for employee burnout in oncology practitioners and its impacts on both providers and patients (90).

Efforts have been made within Canada to shift the practice of routine follow-up care to PCPs. Although, there is confusion amongst providers regarding the roles and responsibilities of follow-up care. The role of the PCP in follow-up care has been illdefined and largely informal (91). In 2015, Choosing Wisely, a national initiative to encourage the reduction of unnecessary tests and treatments, identified the provision of follow-up care in high-cost cancer centres as one of 10 tests/treatments in cancer care not supported by evidence (92). The evidence supporting this change in practice comes from randomized controlled trials that demonstrated follow-up care by PCPs was safe, effective, and equivalent to oncologist-led follow-up care (8-10). Informed by this literature, Cancer Care Ontario created cancer survivorship guidelines for follow-up models of care. In 2017, they released a guideline stating discharge from oncologist-led care to PCP-led care was deemed a reasonable option for breast and colorectal survivors based on evidence and for prostate survivors based on the expert opinion of its similar disease trajectory to the former cancer sites (75). Their 2019 updated guideline recommended the implementation of transfer from specialist to PCP care be focused on breast and colorectal cancer site groups, with expansion to other cancer sites over time (74). In the context of this research, there is a notable lack of evidence for follow-up care

by PCPs in gynecologic survivors. However, guidelines commonly cite five years of recurrence-free follow-up as a marker for transition to PCP follow-up for some types of cervical, endometrial, and vulvar gynecologic cancer survivors (71,72,79,80,82). Studies that examined follow-up care utilization in Canada showed that despite this evidence and PCPs' expressed interest in providing follow-up care (8,10,93), utilization patterns are varied, and survivor needs are still not met (29–31,94–96).

There are at least three influential contextual points to consider upon reviewing the literature on current patterns of follow-up care; these include cancer site, study province, and provider specialty. Cancer sites examined in published studies on patterns of health care utilization in survivors within Canada include colorectal (31,94,97), breast (30,95), Hodgkin Lymphoma (98), as well as one study multi-site investigation including breast, gynecologic, prostate, and colorectal (29). The majority of the studies that examine follow-up care were done in Ontario (ON) (30,95,98–100), two included Manitoba (MB) and BC, and four were done in NS (29-31,101). Three of these focused primarily on screening/preventative measures in follow-up care (94,99,101). The evidence shows a variation in PCP use across provinces for follow-up care (100). Discharge from the oncology clinic to primary care can vary by oncology provider (29). A cohort study of colorectal cancer survivors in NS found that almost 60% of survivors continued to receive follow-up care appointments with their oncologist four years post-diagnosis. The majority of survivors engaged in follow-up visits to multiple specialist physicians (31). Hodgson et al. (2010) found that most patients had visits with both a PCP and an oncologist in years two through five after their Hodgkin's Lymphoma diagnosis. Though

many physician visits were noted, screening recommendations were not widely met (96). Similarly, a cohort study done in ON by Grunfeld et al. (2010) found that most survivors received follow-up care from their oncologist and PCP (95). Further, Kendell et al. (2017) examined physician services use across BC, MB, ON, and NS, and found that during the follow-up period (one to five years post-diagnosis) across all provinces, 65-93% of breast cancer survivors received follow-up from both oncologists and PCPs (30). Interprovincial variations were observed, with greater surgeon follow-up in NS, greater PCP follow-up in BC, and greater medical oncologist follow-up in MB (30). Overall, these results point to a lack of change from specialist to PCP-led follow-up survivor care.

There are multiple reasons why care patterns vary by provider type (i.e., PCP, oncologist, surgeon), amongst provinces, and cancer sites. Different provincial policies, initiatives, recommendations, and resources to support PCP-led follow-up care may account for some of the variation (30). Oncologists may be skeptical of PCPs' ability to provide cancer follow-up care (102,103). Additionally, PCPs may lack confidence, support, or education and feel overwhelmed by their patient load. It may also be difficult for the oncologist to transfer care back to the PCP following treatment completion after forming a therapeutic relationship (104,105). Patient preference plays a role as well. A 2017 study of prostate and melanoma cancer survivors in the Netherlands found survivors more likely to prefer a specialist (such as an oncologist) were women, older, or lower/intermediately educated. Preference for follow-up care provider (specialist versus PCP) also varies based on PCP satisfaction (106). Health visits of cancer survivors may relate to specific cancer concerns, residual mental health effects, and other chronic

conditions. It is possible that for patients with varying concerns, seeing a specialist (i.e., surgeon or oncologist) alone for follow-up does not meet all their health care needs; therefore, they also seek PCP care. Another hurdle is the lack of primary care in provinces across Canada. Adding more responsibilities to an already busy field (i.e., primary care) may impede change (107). Furthermore, some studies done amidst the efforts for change in practice may lag in capturing the actual change in practice to date (29).

2.2 Aging and Frailty

The primary risk factor for cancer is age, and thus, older people make up the majority of people who get cancer. Of all diagnosed cancers, 9 in 10 will occur in Canadians aged 50 years and older (3,24). The aging population is of particular importance for the future of cancer care, as the fastest-growing age group in Canada is those aged 65 years and above (3,24). The proportion of Canadians over the age of 65 is expected to reach 25% in 2036 (108). Paralleling this trend, the incidence of cancer is expected to rise into the 2030s. As previously described, our aging population contributes to cancer's growing burden and increased need for cancer-related services. With more and older survivors, when planning to sustain cancer care services, this population's needs must be considered and understood.

Older people tend to have a greater accumulation of health conditions to be managed. Further, as people grow older, they are more likely to experience frailty (109). Approximately 25% of Canadians over age 65 are frail, increasing to over 50% in those

older than 85 (21). Although there are various definitions of frailty, it may be described as a vulnerable health state resulting from age-related physical (e.g., weakness, muscle wasting, slow gait, decreased endurance), cognitive (e.g., dementia), and/or psychosocial (e.g., decreased autonomy with activities of daily living [ADLs], social isolation) deficits that leaves one susceptible to adverse health outcomes such as falls, hospitalization, and mortality (14,15,110). Consequently, this high risk of acute complications (e.g., infection) often delays functional rehabilitation of adverse outcomes such as disability induced by procedures or hospitalization, and increases health costs (111). Frailty is common in older individuals and those with multiple chronic conditions, but it does not require either to exist (14). Frailty may be a precursor for developing other chronic diseases, or in turn, chronic conditions may contribute to frailty (111). In contrast to terms such as comorbidity or disability, frailty has distinct implications related to health needs. As frailty includes multi-system health issues present in one person, their care and needs are increasingly complex, and their care often requires coordination amongst several providers. Frequent, intense medical monitoring to prevent acute fluctuations in health status amongst frail patients may promote more effective rehabilitation and improve prognosis (111). However, the literature on frailty within cancer survivor care is limited.

2.3 Measures of Frailty

How should frailty be measured? There are many approaches to measuring frailty. Although there is no gold standard for frailty assessment, two prominent theories that

guide frailty measurement are the phenotypic approach and the deficit accumulation approach (112).

2.3.1 Phenotypic Approach to Frailty

The phenotypic approach to frailty is a widely recognized measure of frailty in both clinical and research settings. Its criteria include grip strength, weight loss, exhaustion, walking speed, and physical activity (113). Individuals are frail if they have poor performance in any three or more criteria; those that meet two are prefrail, and those that do not perform poorly in any are non-frail. The original measurement criteria proposed by Fried et al. (2001) have evolved (114). A systematic review conducted in 2015 identified 262 variations of the frailty phenotype criteria (113). Prevalence of frailty amongst the various phenotype classifications ranged from 12.7-28.2%. Studies that use this approach are therefore limited in their comparability. Its generalizability outside of an older adult population is also limited. There are advantages of the physiologic assessment measurements used in this approach, and there are methods for mitigation of missing criterion information; however, its application to administrative data studies is limited, as the majority of administrative data do not provide the information to apply the criteria appropriately (113).

2.3.2 The Deficits Accumulation Approach to Frailty and the Frailty Index The deficit accumulation approach focuses on the number of health deficits rather than their characteristics. Deficits may include diseases, signs, symptoms, social issues, and disabilities that collectively influence organ systems (115). This approach suggests that as one accumulates many health deficits, slowly over time, they become frailer and are at greater risk for adverse health outcomes. It can be understood from a stochastic stance. This means there is a relationship between the number of deficits (Nd), degree of environmental stressors (λ) , and the ability to recover (R), written as Nd= λ R (Little's Law). In one person's lifetime, generally, the trajectory of one's environmental stressors and recovery is irregular. Though, on a population level, frailty trajectories are regular and show on average an increase in the frailty index by 10-fold between the age of 20-90 years. This increase is explained by Little's Law, as environmental stress is said to be held relatively constant; thus, it is the recovery time that changes over time (115).

The deficit accumulation approach is applied through the frailty index. The number of health deficits one has, divided by the number of deficits considered, determines the frailty index. It is a score on a continuous scale from zero to one. The higher the index number, the frailer or more vulnerable to adverse health outcomes, one is (115).

The frailty index allows for variations in indices depending on what deficit data are available. Searle et al. (2008) outlined a standard procedure to create a frailty index from an existing health/aging dataset (112). None of the samples that have operationalized the frailty index have used the same deficits. Regardless, the relationship between deficit accumulation and mortality remains similar (112). This reinforces frailty as a real phenomenon and "a property of a biologically complex system" (71 p.1472). It supports that frailty can be measured in various ways and within existing datasets that might not have initially set out to measure frailty. Examples of deficit variables used in this method

include help with ADLs, self-rated health, mini-mental exam scores, body mass index (BMI), grip strength, or diagnoses such as heart attack, stroke, or diabetes (112). For the frailty index to achieve a reliable estimate of frailty, it should include a minimum of 20 and a suggested amount of 30 deficits. The deficits needed to create a frailty index are not always available in administrative data (14).

The phenotypic approach and the deficit accumulation approach aid our understanding of frailty. These algorithms to identify frailty do not work for many administrative databases. This is because some types of data, such as clinical assessment data, are not routinely found in population-based, administrative health databases (112,113,115). Some individuals may also obtain the health data needed to create a frailty index by using databases that contain International Resident Assessment Instrument data. However, these data are not gathered or accessible consistently across Canada (14,116,117). As research expands to measure frailty in other populations, such as cancer survivors, and in the context of health services research, more user-friendly measures for use in administrative data are required.

2.3.3 Rules to Identify Frailty in Administrative Data

Urquhart et al. (2017) addressed this concern by creating rules to identify frailty in administrative data (14). Before developing these rules, only one known algorithm existed to identify frail individuals within population-based administrative data (14). The earlier algorithm examined diagnostic codes from causes of death cited in death certificates (118). The weakness of the prior method is that it does not capture the

complexity of the concept of frailty because it only looks at the specific diagnoses cited on death certificates and includes medical conditions that may not adequately represent frailty (14,118). Further, its use is limited to those who are deceased (14).

While there are many clinical practice tools to screen for frailty, to gain information about frailty on a population level, frailty identification in administrative data is essential. Administrative health databases provide robust, efficient measures of population-based quality of care (119,120). Thus, the ability to identify frail individuals on a population level through administrative data allows researchers and policymakers to efficiently monitor health care utilization across the country, which may help better inform program and policy solutions to address the consequences of frailty (14).

The rules to identify frailty in administrative data create a more comprehensive approach to identify frailty in population-based administrative databases that may be applied to both those living and deceased and be used in a wide variety of settings. The advantage of these rules to identify frailty is that they consider multiple conditions that contribute to frailty, as does the deficit approach, and they can be measured and applied in most administrative health databases (14).

The rules used two sources of administrative data: hospital discharge abstracts (i.e., CIHI DAD) and physicians' claims (i.e., MSI). This method considers the concept of frailty in the context of prior frailty research and expert (geriatrician) consultation (121,122). There are three primary rules to identify frailty. One, the person is considered frail if they

are a long-term care resident. Two, the person is considered frail if they receive palliative care. Three, the person is considered frail if they meet at least two or more domain criteria (cognitive, general health status, incontinence, falls, nutritional issues, functional performance, or targeted health service utilization). Domains are derived from diagnostic codes associated with clinical frailty scales (the Edmonton Frail Scale [EFS] and the Clinical Frailty Scale [CFS]), including service utilization and several suggested events (e.g., falls, incontinence) (14,118,121,122).

These rules were created by the thesis supervisor, are published, and are being used by teams across the country, including Ontario's Institute for Clinical Evaluative Sciences (ICES), to identify frail persons in administrative databases (14,21). To date, at least three studies have been published utilizing the rules (123–125). Urquhart et al. (2017) determined that 5.1-14.7% of living persons aged 65 years and older in their study were frail, varying between provinces (BC, AB, ON, Quebec [QC], and NS) (14). This is comparable to previous literature. For example, studies using the phenotypic approach in a similar age demographic, including community-dwelling persons aged 65 years and older, estimated the prevalence of frailty ranged from 4.0-17.0% (14,126). The rules to identify frailty were used in Canada's first study to examine care for older adults with frailty using administrative data (14,21). The cohort included older persons with frailty in BC, AB, ON, QC, and NS. The study demonstrated the feasibility of using the rules to identify frailty in administrative health data across five Canadian provinces. This thesis used the rules developed by Urquhart et al. (2017) as they allow the use of administrative data, fitting for the study's focus on the health services perspective. The

rules to identify frailty are further described in the methods section of this thesis (Chapter 4).

2.4 Identifying Frailty in Non-cancer Survivor Populations/Areas of Health Care In other (non-cancer specific) areas of care within Canada, frail people often do not receive care that meets their needs (15,16,127,128). There is a need to identify those with frailty, formulate health systems that address their needs, and pay particular attention to their care delivery. Frail individuals have complex care needs that require a unique approach to care delivery (15,16).

In 2016, Giguere et al. conducted a scoping review that described the state of health care delivery and utilization of frail individuals within Canada. They included studies of various methods ranging from qualitative inquiry to randomized trials (127). It concluded with an urgency to restructure care delivery to address the complex health issues faced by frail people appropriately. As older adults use a disproportionate amount of hospital services, their health care costs tend to be high. Reducing these costs is certainly an area of interest for decision-makers. Specifically, within NS, provincial health system decision-makers and health care providers voiced the need for improvement in information sharing, care planning, and, primarily, delivery of care. Canada has been working to improve resources and services to meet the needs of frail older adults; however, coordination of services remains a struggle and impacts both accessibility and efficiency of care (127).

Subsequently, Giguere et al. (2018) explored key stakeholders' views on the quality of care and services available to frail seniors in five Canadian provinces (BC, AB, ON, QC, NS) (15). This qualitative study examined the views of 42 frail older adults, caregivers, and administrative personnel. Stakeholders in frailty care voiced the need for more inclusion of patients and families in care planning, and more integration of care and services across settings, over time, and between providers, to meet frail seniors' needs (15). Given the number and variety of providers that may care for a frail individual, models of care are of interest. Prior research has shown outside of cancer care, that the complexity of frail peoples' needs is often best addressed within multidisciplinary teams (129). Finally, Giguere et al. (2018) called for systematic identification of frail older people to adapt health care systems to the population's needs (15).

Within the context of cancer care, studies have been conducted exploring the prevalence of frailty within newly diagnosed cancer patients and/or those receiving active treatment. Handforth et al. (2015) conducted a systematic review, establishing the prevalence of frailty and prefrailty amongst older cancer patients. With data from 20 observational studies (n=2,916), they found a median prevalence of frailty of 42%, ranging from six to 86%, and a median prevalence of prefrailty of 43%, ranging from 13-79%. Furthermore, treatment complications were more common in frail participants, including intolerance to cancer treatment (adjusted odds ratio [OR] 4.86, 95% confidence interval [CI] 2.19-10.78) and postoperative complications (adjusted 30-day hazard ratio [HR] 3.19, 95% CI 1.68-6.04) (130).

2.5 Identifying Frailty in Cancer Survivors

Frailty within the cancer survivor population is an important area for health services research as many cancer survivors are older, have comorbid conditions, and long-term health issues from their cancer treatment(s). Understanding the prevalence of frailty in cancer survivors would help discern and plan care delivery to improve the burdened cancer care system's efficiency.

The literature examining frailty in cancer survivors has grown in the last five years, yet gaps remain. A recent review published by Ness et al. (2020) discussed the prevalence of frailty in various cancer survivor populations (131). Twenty-three studies were included in the review. However, as the review also aimed to describe how frailty at diagnosis impacts cancer outcomes, some studies in the review assessed frailty status only at cancer diagnosis (131). To our knowledge, there are no studies that examine frailty in cancer survivors using solely administrative data or that estimate the prevalence of frailty amongst a Canadian cancer survivor population.

While a limited number of studies have been conducted on frailty in cancer survivors, two relevant observational studies that used self-reported survey data should be further examined. The first, a US prospective cohort study by Brown et al. (2015), quantified the prognostic value of prefrailty and frailty in an older adult non-skin-related cancer survivor population (132). They used data from the Third National Health and Nutrition Examination Survey to identify frailty using the following criteria: low BMI, slow walking speed, weakness, exhaustion, and low physical activity. Data were linked to the
National Death Index for assessment of the primary outcome, all-cause mortality. Prefrail cancer survivors (HR 1.84, 95% CI 1.28-2.65, p=0.001) and frail cancer survivors (HR 2.79, 95% CI 1.34-5.81, p=0.006) had higher risks of premature mortality than those classified as non-frail. In this study, 37.3% of participants were prefrail, and 9.1% were frail. They concluded that identifying prefrailty and frailty in older adult cancer survivors would help target cancer care and improve survival (132).

In another study, Perez-zepeda et al. (2016) examined the association between a history of cancer (i.e., self-reported cancer between 2001 and 2012) and frailty in a group of Mexican older adults (133). The nested case-control study used data from the Mexican Health and Aging Study. A 55-item frailty index was created using self-reported data. Of the 8,022 older adults with a mean age of 70.6 years of age, the prevalence of a past medical history of cancer was 3.6% (n=288). They concluded that cancer was associated with a higher frailty index, with a potentially relevant role of the time elapsed since cancer diagnosis. They found a significant association between cancer history and incident frailty amongst those recently diagnosed (i.e., less than 10 years) (OR 1.53, 95 % CI 1.04-2.26, p=0.03)]. This association remained when adjusted for age, sex, marital status, years in school, at least one negative event in the last 11 years, smoking status, cognitive decline, chemotherapy, surgery, and radiotherapy interaction (OR 1.74, 95 % CI 1.15-2.61, p=0.008). However, a significant inverse relationship was shown between those with a remote diagnosis (i.e., greater than 10 years from initial cancer diagnosis) and worsening frailty, compared to those without a cancer history (OR 0.56, 95 % CI 0.39-0.8, p=0.002); adjusted model (OR 0.61, 95 % CI 0.38-0.99, p=0.046).

The authors suggested that health professionals be aware of these associations to improve outcomes for older adults who have survived cancer. However, the study had at least two major limitations. One was their definition of a cancer survivor. They did not indicate participants' current cancer status or indicate an assessment of recurrence (133). They used cancer history and cancer survivor synonymously, classifying them into two groups: those recently diagnosed (i.e., less than 10 years) and remotely diagnosed (i.e., greater than 10 years from initial diagnosis) (n=8,022). Secondly, their analysis was inappropriate for their described study design, which may have biased their results (133). They used logistic regression, while conditional logistic regression is recommended for a nested-case control design, and did not explain any alternative appropriate analysis techniques (134).

While most studies have looked at frailty in the context of older adult survivors, the literature has also identified frailty amongst adolescents and young adult (AYA) cancer survivors. In their review, Ness and Wogksch (2020) identified eight studies examining frailty within the AYA population (131). One of those studies, a 2018 cross-sectional study conducted in the US, examined frailty in adult cancer survivors aged 15-39 years using the FRAIL (i.e., fatigue, resistance, ambulation, illness, loss of weight) questionnaire. Within their sample (n=271), 184 were at least one-year post-diagnosis. Ten percent were frail, and 21% were prefrail. They concluded a high prevalence of frailty and comorbidities exist among AYA cancer survivors, suggestive of accelerated aging (135). Furthermore, described in the 2020 review by Ness and Wogksch, although the frailty prevalence amongst adult cancer survivors was higher, ranging from 9.1-59%

(often double that of age, sex and race matched populations), the frailty prevalence amongst AYA cancer survivors was also high. Comparable to frailty found in community-dwelling older adults aged 60-80 years, the prevalence of frailty in childhood cancer survivors in their 30s and 40s ranged from 7.9-47% (131).

In addition to age, frailty in cancer survivors is associated with sex, cancer type, treatment, chronic disease, lifestyle, and access to care (131). Female cancer survivors are more likely to be frail than male cancer survivors. Treatment type also plays a role in susceptibility to frailty. Females receiving treatment affecting estrogen and males receiving androgen deprivation therapy are at increased risk. Receiving radiation to the brain, abdomen, and pelvis are also treatment-related risk factors for frailty amongst cancer survivors. Modifiable risk factors for frailty in cancer survivors include smoking, obesity, and a sedentary lifestyle (131).

These studies point to the importance of identifying frailty in cancer survivors, not only in older adults but also in survivors of all ages. However, there is a need for further studies with more robust methods to strengthen the evidence and determine the prevalence of frailty on a population scale using administrative data. Additionally, further studies in this area will expand our understanding of how frailty varies by survivor characteristics.

2.6 Future Models of Care are Personalized

The growing number of survivors, alongside increased constraints on the cancer care system, has prompted research on alternative care models for this population (136). As a new research area, the term 'personalized models of care' has not been well defined. However, it may be thought of as a way to tailor care structures and delivery to specific groups of cancer survivors with unique needs. For example, some survivors may need more specialized follow-up interventions, such as those with certain complications (e.g., lymphedema), or more support from a multidisciplinary team (i.e., physiotherapy, occupational therapy, career counselling, etc.) (6). Some survivors have severe organ dysfunction and/or are at high risk for recurrence or serious late effects (e.g., cardiovascular and pulmonary disease), causing premature mortality. Alternatively, some survivors may have fewer needs or lower risk of complications and thus, may be effectively managed by PCPs. This gradient of survivor concerns may also be influenced by other comorbid health conditions, access to health care, or psychosocial issues (5). Personalized survivor care models are a way to identify and allocate appropriate followup care resources to different groups of survivors by streamlining care delivery and organization (6).

Some cancer survivorship care models explored internationally include those based on provider type, cancer site, length of follow-up care, and the survivor's role in recovery (137). Models may be further categorized as nurse-led survivorship care models, riskbased pathways (i.e., stratified resource allocation based on unique survivor needs), rehabilitation-focused care models, and self-management models (137). Eleven

randomized controlled trials identified in a 2012 systematic review compared different follow-up care models; none studied needs- or risk-stratified-based care (138). Recently, the National Cancer Survivorship Initiative highlighted issues that cancer survivors face and expressed the need for care delivery to reflect those specific health concerns appropriately. One priority research area identified was the development of risk stratification tools, though it provided little details on this definition or important outcomes to address (20). Risk-based care and coordination between oncologists and PCPs have been identified as important metrics of quality cancer survivorship care (5).

Risk- and needs-based care models fall under the umbrella of personalized models of care. First, let us understand the meaning of risk- and needs-based models. Risk identification refers to events a survivor is more likely to experience in the future, while needs describe issues that are currently present (20). It is hypothesized that levels of risk may help allocate care delivery. For example, survivors at low risk for future health problems may be transitioned to their PCP soon after completion of therapy. A survivor at moderate risk could be cared for primarily by their PCP, with periodic evaluation by their oncology team. The oncology team may follow survivors at high risk for late effects yearly, such as those treated with a stem-cell transplant (6). In practice, some provinces may follow a variation of this risk-stratification through the use of survivorship clinics.

Additionally, Mittmann et al. (2019) recently found that the transition of a group of lowrisk ON breast cancer survivors from oncology to PCP follow-up care was associated with fewer health care costs and similar or better patient outcomes (139). However, there is no consistent practice or tested model of care to facilitate this transition. The assessment of both risk and need is required in the context of cancer survivorship (20). In contrast to risk, (healthcare) need is the capacity to benefit from health care (20). The level of need is a more immediate burden and consideration for health care resource use. However, the current literature does not distinguish specific items that pertain to risk versus need. Instead, they identify a list of cancer-related physical, psychological and social issues that may warrant both risk and needs assessment, including late and long-term physical effects of treatment, depression, anxiety, and financial and employment concerns (5,20).

Further, it is possible to incorporate risk- and needs-based models into a shared-care approach, as a notable aspect of quality survivorship care is having functional interfaces between each stage on the care continuum (6,137,140,141). However, before delineating and testing such models, further work must be done to identify sub-groups of survivors that require different approaches to care. There is an urgency for reliable ways to systematically assess and identify those unique needs through risk stratification and/or needs identification tools (5,6,20).

A growing body of literature suggests that identifying sub-groups of cancer survivors who require different models of care will reduce the burden on the cancer care system (5,6). Personalized models of care aim to target vulnerable high risk and/or needs patients to adequately address their concerns and create a system wherein the patient flow is improved to allow for quicker and easier access for both those identified as high risk and

those who are relatively healthy (i.e., lower risk) individuals. Ideally, this would reduce care costs by organizing care in a more efficient way (139,142).

Identifying frail cancer survivors within administrative data is one way to build upon this growing body of evidence for more personalized survivorship models. With the increasing aging population and the number of older survivors, this large group of survivors will likely require increasingly complex care. Frail survivors may have both high risk (e.g., mortality) and needs (i.e., related to multiple health problems) (132). Consequently, it is essential to understand how aging and frailty apply to future models and cancer follow-up care interventions. Due to the recent shift to encourage PCP involvement in follow-up care and the known complex needs of frail people in other areas of care, frail cancer survivors may also benefit from alternative care models, such as a shared-care approach.

Further research is needed first to determine the prevalence of frailty in cancer survivors and then examine their follow-up care usage patterns to inform improved health service planning. Understanding the trends in frail survivor follow-up care is imperative to creating future models of care that work for this population because of the complex needs of care in frail individuals. The prevalence of frail cancer survivors in NS has not been quantified, nor has their health services utilization been examined. The proposed study would be the first in Canada, if not worldwide, to do both using administrative data. Identifying frailty within the cancer survivor population is one step in gaining insight to inform future models of cancer follow-up care.

Chapter 3: Objectives

The goal of this thesis project was to build upon the body of research aimed at designing more personalized models of care for cancer survivors by identifying frail cancer survivors in NS as one subpopulation who may have differing and/or unique needs/risks after cancer treatment. Prior research outside of cancer tells us frail people are a special population with unique care needs. Our first step to address this issue in cancer care was to quantify frailty in the population. We then described patterns of follow-up care of cancer survivors to inform future model reform. The study's objectives were:

- To estimate the burden of frailty amongst a NS cancer survivor cohort and determine how frailty differs by patient characteristics.
 - 1.1. To estimate the prevalence of frailty amongst a NS cancer survivor cohort and determine how it differs by clinical, demographic, and other characteristics.
 - 1.2. To identify frail cancer survivors' traits (i.e., demographic, clinical, and other characteristics) that differ from those of non-frail cancer survivors.
 - 1.3. To determine the underlying attributes (i.e., physical or cognitive impairment and health service utilization markers) for which cancer survivors are deemed frail.
- To identify patterns of cancer-related follow-up care visits and how they differ between non-frail and frail cancer survivors, by patient characteristics, within a NS cohort.
 - 2.1. To determine whether frail cancer survivors have greater cancer-related followup care use and how this differs by patient characteristics.
 - 2.2. To identify the percentage of cancer survivors' follow-up visits provided by PCPs and determine how this differs by frailty and other patient characteristics.

Chapter 4: Methods

4.1 Data and Study Population

The study design was descriptive in nature and used data from a retrospective populationbased dataset of NS cancer survivors. We used a Research Ethics Board (REB) approved, updated linked administrative dataset from the supervisor's prior work, containing all Nova Scotians diagnosed with invasive breast, colorectal, gynecologic, or prostate cancer between January 01, 2006, and December 31, 2013 (n=20,901). Data from the NSCR were linked, at the patient level, to data obtained from the Oncology Patient Information System (OPIS) and HDNS databases, including MSI Physician Billings, CIHI DAD, Census-Based Small Area Data/Postal Code Conversion File (PCCF+), the Licensed Provider Registry (LPR), and the Insured Patient Registry (IPR). These data provided information on patient characteristics, such as demographic and clinical characteristics, and health services use. Data from NSCR and OPIS spanned from January 01, 2006, to December 31, 2018. Data obtained from HDNS databases began two years before each individual's diagnosis date, thus starting January 01, 2004, and ending December 31, 2018. Data two years before each individual's diagnosis date was required for comorbidity case definitions. Figure 1 visually depicts study timeline information.

From this dataset, a survivor cohort was identified (n=10,176) using the following exclusion criteria: death within one year of diagnosis, stage IV and stage classified as unknown at diagnosis, evidence of metastases or new primary cancer within one year of diagnosis, previous primary cancer diagnosis (i.e., cancer history), and lack of IPR eligibility (i.e., no record of IPR eligibility, or gaps in IPR coverage greater than or equal

to 12 months). Due to our interest in cancer survivors, we included only those with stage I-III disease as the treatment and life expectancy differ for those with unknown and stage IV disease. Additionally, a minimum observation period of greater than or equal to two and a half years during follow-up time was applied. Patients observed for a short followup time may be misclassified as frail due to some event (e.g., death or cancer recurrence). The minimum follow-up time was put in place to balance these potential biases. Patients observed for a longer follow-up period have an increased chance for frailty identification. The maximum length of follow-up was 12 years. Survivors were censored on the evidence of a new primary cancer, cancer recurrence, loss of IPR eligibility (i.e., last date of IPR eligibility), and/or death. Data for censoring was obtained from the NSCR, OPIS, MSI Physician Billings, IPR, and CIHI DAD. Specific MSI Physician Billings and CIHI DAD procedure codes used for censoring can be found in Appendix B. Due to the linked data, the cohort included only those with a valid provincial health card number.

As frailty is not a diagnosis, frail cancer survivors were identified from the NS cancer survivor cohort using the decision rules to identify frail persons from two administrative health databases: CIHI DAD and MSI Physician Billings (14).



2004

Figure 1: Study timeline.

STUDY TIMELINE

4.2 Variables

The following describes the study's variables. All study variables were measured at the individual level except income, which was measured at the neighbourhood level.

4.2.1 Key Measures

Frailty

Frailty is the key measure for objective 1. The rules to identify frailty in administrative data were used to identify frail cancer survivors within the NS cancer survivor cohort. We used the International Classification of Diseases, Ninth and Tenth Revisions (ICD9/10) codes from CIHI DAD and MSI Physician Billings to apply the rules to identify frailty in administrative data for each survivor. These specific codes are described in previously published work (14,127).

The rules to assess frailty are based on the following three criteria. Rule 1, the person was a long-term care resident. Rule 2, the person received palliative care. Rule 3 is derived from diagnoses within the EFS, CFS, and service utilization. To be considered frail via rule 3, the individual had to meet two or more criteria listed in any one of seven domains. These domains included cognitive impairment, general health status (e.g., hospital admissions/emergency department visits), incontinence, falls, nutrition, functional performance, and targeted health service utilization (e.g., geriatrician visits). This process is described in further detail in another paper (14). In this study, frailty was categorized as a binary variable (non-frail/frail). Frailty status was determined for patients in the NS cancer survivor cohort (n=10,176) using data collected beginning 366 days post-cancer diagnosis, ending at five years of follow-up. In reference to the observation of MSI Physician Billings and CIHI DAD ICD9/10 codes, this included at least two hospital admissions or at least two emergency department visits within any 365-day period from 366 days after the index cancer diagnosis to five years of follow-up. This is because active cancer treatment is expected during the first year post-diagnosis and may contribute to a brief deterioration in one's health state (29). Smitherman et al. (2018) used a similar one-year post-diagnosis frailty timeframe (135). Five years is commonly recognized as the end of traditional cancer follow-up, thus why we chose it as a marker for the frailty timeframe end.

Once frail cancer survivors were identified, the prevalence of frailty in the NS cancer survivor cohort was calculated. Frailty was the dependent (outcome) variable of interest in the logistic regression model run to address objective 1.2. Frailty was also used as a key clinical characteristic (independent) variable in objective 2.

Follow-up Care Visits.

The study's second objective focused on follow-up care visits. There is no specific appointment type within OPIS that identifies routine follow-up care. Therefore, Urquhart et al. (2017) made a rule to identify routine follow-up visits using cancer centre data (29) which we used in the present study. The rule uses variables within the existing dataset from NSCR and OPIS (e.g., date of diagnosis, region code, appointment type, resource use, receipt of radiotherapy). It distinguishes routine follow-up care from non-routine

follow-up care (e.g., visits to manage complex late effects or suspicion of recurrence). To determine cancer-related versus non-cancer-related follow-up care within MSI Physician Billings data, we used pre-identified ICD-9/10 billing codes known to be associated with cancer-related care (Appendix C).

Variables were created to describe follow-up care providers (i.e., PCPs, oncologists, and surgeons). Data for the provider type came from OPIS, LPR, and MSI Physician Billings. For this study, a PCP was defined as either a family physician, general, or community medicine practitioner. Additionally, where a provider was suspected to be a general practitioner with training in oncology (i.e., a PCP visit in a cancer center), they were classified as PCP. Oncology included medical and radiation oncologists. Surgeons included general surgeons, urologists, and gynecologists. Although not a surgical speciality, we included gastroenterologists in the surgical category given these specialists perform relevant procedures for colorectal cancer survivors (e.g., colonoscopies). When referring to specialists, this included oncology and surgery physicians.

Total visit count. The outcome variable for objective 2.1. A count of each individual's total cancer-related visits (i.e., PCP, oncology, surgery) during the follow-up period.

Oncology only. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person only visited an oncology provider during the follow-up period or not.

PCP only. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person only visited a PCP during the follow-up period or not.

Surgery only. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person only visited a surgeon during the follow-up period or not.

Oncology and PCP. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person visited both a PCP and an oncology provider during the follow-up period (and not surgery).

Oncology and surgery. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person visited both a surgeon and an oncology provider during the follow-up period (and not primary care).

Surgery and PCP. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person visited both a PCP and a surgeon during the follow-up period (and not oncology).

Oncology, surgery, and PCP. A variable used for objective 2.2 descriptive statistics. It is a binary variable (no/yes) indicating whether the person visited all three, PCP, surgery, and oncology providers during the follow-up period.

Proportion of cancer-related follow-up visits provided by PCPs. The outcome variable for objective 2.2. The count of each individual's visits to PCPs throughout the follow-up period divided by the count of their total cancer-related visits during the follow-up period. This was also categorized into low (i.e., \leq the 25th percentile [\leq 0.30]), medium (i.e., \geq the 25th percentile [>0.30] and < the 75th percentile [<0.73]), and high

(i.e., \geq the 75th percentile [\geq 0.73]) proportion of cancer-related follow-up visits provided by PCPs.

4.2.2 Patient Characteristics

Patient characteristics included clinical characteristics, demographics, and other characteristics. The following describes each patient variable.

Clinical Characteristics

Frailty. Frailty, as defined in objective 1 (non-frail/frail) was used in objective 2. In contrast to objective 1, in objective 2, frailty was not used as an outcome variable. Non-frail was used as the reference category in analyses.

Comorbidity. A comorbidity score was calculated based on the list of comorbid conditions developed by Elixhauser et al. (1998), excluding cancer-related comorbidities (143). The score ranges from zero to 28. This is a baseline comorbidity measure calculated over the two years prior to cancer diagnosis using ICD9/10 codes from hospital discharge records (i.e., from within CIHI DAD). Patients were divided into categories based on their number of comorbidities prior to cancer diagnosis: 0, 1, 2, \geq 3, or no hospitalizations. Zero was used as the reference category in analyses.

Cancer site. The cancer site was defined as the index cancer's anatomical location (i.e., breast, colorectal, gynecologic, or prostate cancer) as recorded within the NSCR. Breast cancer was used as the reference category in analyses.

Cancer stage. Cancer stage at diagnosis was defined using the Collaborative Stage Data Collection System (i.e., stage I, II, or III) identified within the NSCR. Stage I was used as the reference category in analyses. *Chemotherapy*. Receipt of chemotherapy during the treatment period (i.e., from the date of the index cancer diagnosis extending 365 days) was defined as the receipt of neoadjuvant/adjuvant chemotherapy as determined from the NSCR, OPIS, and MSI Physician Billings (e.g., procedure code for administration of IV chemotherapy). Neoadjuvant therapy is defined as treatment given prior to the main treatment, usually surgery, to shrink the tumor beforehand. Adjuvant therapy is an additional cancer treatment given after the primary treatment (144). The categorization of this variable was binary, including no indication of administration of chemotherapy during the treatment period (no), and receipt of neoadjuvant/adjuvant chemotherapy during the treatment period (yes). No chemotherapy was used as the reference category in analyses.

Year of cancer diagnosis. Year of cancer diagnosis was defined as the year an individual was diagnosed with cancer as recorded in the NSCR. Both a continuous and a categorical variable were created. Categories were years 2006-2008, 2009-2010, and 2011-2013.

Demographics

Age. Age represents the age at cancer diagnosis. This variable was created using the date of birth and date of diagnosis from the NSCR. Both a continuous and a categorical variable were created. Age categories included: \leq 40, 41-50, 51-65, 66-79, and \geq 80 years.

Sex. Sex was defined as male or female biological sex as recorded in the NSCR. Male sex was used as the reference category in analyses.

Residence. Urban/rural residence was defined using the patient's postal code at diagnosis and the PCCF+ file. The Statistical Area Classification (SACtype) was used to

determine whether cohort members reside in urban or rural areas based on dichotomization of the Metropolitan Influence Zone (MIZ) classification. MIZ classifies place of patient residence outside Census Metropolitan Areas (CMA) and Census Agglomerations (CA) (145). CMA is defined as an urban population of at least 100,000, and CA includes an urban population of 10,000-99,999. CMA and CA include all neighboring municipalities where 50% or more of the labor force commutes to the urban center. MIZ includes Census subdivisions outside of CMA or CA, which are categorized as strong MIZ, moderate MIZ, weak and no MIZ. Strong MIZ includes a population where 30-49% of the employed workforce commutes to CMA or CA. Moderate and weak MIZ categories include populations where less than 30% and 5%, respectively, of the employed workforce commute to CMA or CA. No MIZ represents a population where none of the employed commute to a CMA or CA, or the employed workforce includes less than 40 people. For this study, urban residence was defined as MIZ categories of CMA, CA, or strong MIZ. Rural residence was defined as MIZ categories of moderate MIZ, weak MIZ, or no MIZ (145). Urban residence was used as the reference category in analyses.

Poverty. Poverty is a proxy measure for individuals' socioeconomic status (SES), a known social determinant of health. The individual-level variables used to measure poverty were enrollment in low-income provincial drug programs, including income assistance (PIA and PIAX), NS family Pharmacare (PNSFP), Pharmacare GIS (PMSIG), and Pharmacare no co-payment (PMSIX). These were from the IPR eligibility file held by HDNS, within the variable program. Poverty was categorized as a binary variable (no/yes) to indicate poverty based on ever having been enrolled in a low-income drug program during the study period; otherwise, they were not considered in poverty. No poverty was used as the reference category in analyses.

Neighbourhood income. Neighbourhood income represents one's median neighbourhood income at the time of cancer diagnosis. It is another proxy measure for SES, though it measures a different aspect than poverty. This income measure was reported at the neighborhood level, as individual household income is not reported in administrative health databases. The patient postal code of residence at diagnosis (from NSCR) was linked to Statistics Canada dissemination areas using the PCCF+. Dissemination areas, used as a proxy for neighbourhoods, are the smallest geographic units for which Statistics Canada reports median household income. One-hundred-andthirty-seven (1.26%) of the study cohort were missing income data. Complete case analysis was used, and those with missing data were excluded from the study. Neighbourhood income, a continuous variable, was also categorized into quintiles representing lowest, low-middle, middle, upper-middle, and highest.

Other Characteristics

Follow-up years. A continuous variable that represents the time in years starting 366 days after the index cancer diagnosis from the NSCR, extending until the study end (i.e., December 31, 2018) or time of censoring. A corresponding categorical variable for descriptive statistics was also created. Categories were: <5 years (i.e., at minimum 2.5 years according to the study criteria, up to but not including 5 years), 5-7 years, and 8-12 years.

Censor reason. A categorical variable that represents the reason for one's end-ofstudy observation. Indications to censor were collected from NSCR, OPIS, MSI Physician Billings, CIHI DAD, and IPR. The initial date and reason indicating a need to censor were determined for each subject. Categories included subsequent cancer (i.e., new primary cancer), cancer recurrence, end of IPR eligibility, death, and study end (i.e., December 31, 2018).

4.3 Analysis

Prior to conducting each regression analyses, the exploratory analysis looked at the mean, median, standard deviation (SD), variance, interquartile range (IQR), boxplots, histograms, etc., of data. The distributions of outcome variables were assessed to assure alignment with the regression type and research question. Multicollinearity between covariates was assessed, including use of the variance inflation factor. Variables included in the final (adjusted/multivariate) models were informed by relevant literature and based on a combination of clinical and statistical significance. All analyses were conducted using Stata/MP 15.0.

4.3.1 Objective 1 Analysis

To address objective 1, we used cohort data from both frail and non-frail cancer survivors (n=10,176) to estimate the burden of frailty amongst the NS cancer survivor cohort, and describe how frailty differs by patient characteristics.

Objective 1.1) To estimate the prevalence of frailty amongst a NS cancer survivor cohort and determine how it differs by clinical, demographic, and other characteristics.

The prevalence of frailty in cancer survivors was calculated via proportions and then described by each characteristic (i.e., comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age, residence, poverty, income, follow-up years, and censor reason) (Table 1). The overall prevalence of frailty within the NS cancer survivor cohort, and prevalence by patient characteristics, were calculated as follows:

prevalence of frailty in cancer survivors

$$= \frac{\# of people in the cancer survivor cohort identified as frail}{total \# of cancer survivors in the cohort} \times 100$$

Objective 1.2) To identify frail cancer survivors' traits (i.e., demographic, clinical, and other characteristics) that differ from those of non-frail cancer survivors.

First, the distribution of each trait was estimated and compared between frail and nonfrail cancer survivors, including clinical characteristics (i.e., comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis), demographics (i.e., sex, age, residence, poverty, income) and other characteristics (i.e., follow-up years, censor reason). Chi-squared (χ^2) for categorical variables and t-tests for continuous variables were used to test statistical differences in the distribution between traits (Table 2).

Second, logistic regression on the binary outcome (non-frail/frail) was used to describe how frailty differs according to the patient characteristics (Table 3). Based upon frailty coding, an OR greater than one indicated higher odds of frailty, and an OR less than one indicated lower odds of frailty. Each patient characteristic variable was run in a univariate logistic model with frailty, and then selected variables were included in a multivariable model. The multivariate model allowed us to observe variations in frailty by important characteristics while adjusting for other characteristics. A priori decisions were made to include some patient characteristics in the multivariate model were based upon known importance in the literature (i.e., age, sex, comorbidity), and clinical importance (i.e., cancer site, cancer stage, chemotherapy). SES has been associated with frailty in the literature. As we had three proxy measures of SES, we included only poverty in the multivariate model as it had a large effect size in the univariate model and is measured at the individual level. Urban/rural residence was excluded as it was not significant at p<0.20 in the univariate model. Assumptions for logistic regression (independence of observations and errors, linearity of continuous variables, absence of multicollinearity, and lack of influential outliers) were met. Sensitivity analysis was conducted including residence in the multivariate model; it remained insignificant, and results were robust. Likelihood-Ratio (LR) tests were used to indicate the significance of categorical variables in the multivariate model. Statistical significance was set to alpha level 0.05.

Objective 1.3) To determine the underlying attributes (i.e., physical or cognitive impairment and health service utilization markers) for which survivors are deemed frail.

Descriptive statistics were used to show the number and percentage of individuals identified as frail within the NS cancer survivor cohort according to each rule to identify frailty, including markers of physical and cognitive impairment, as well as health service utilization (Table 4).

4.3.2 Objective 2 Analysis

The sum of objectives 2.1 and 2.2 address the second overarching objective: To identify patterns of cancer-related follow-up care visits and how they differ between non-frail and frail cancer survivors, by patient characteristics within a NS cohort.

Objective 2.1) To determine whether frail cancer survivors have greater cancer-related follow-up care use and how this differs by patient characteristics.

Median, 25th and 75th percentiles were first estimated, along with the IQR, for the number of total (i.e., PCP, oncology, and surgery) cancer-related visits by frailty and other patient characteristics (Table 5).

Based upon the distribution of the outcome (Appendix D), total visit count, the observation of overdispersion (mean=16.1, variance=171.9), and comparison of Akaike's Information Criteria (AIC) and Bayesian Information Criteria (BIC), we chose to use negative binomial generalized linear modelling (GLM) over Poisson regression. We modelled the annual rate of cancer-related follow-up visits, with the key independent

variable, frailty, adjusting for other patient characteristics (Table 6). An offset function in Stata was used to normalize visit exposure time (that is, the time from the beginning of the follow-up period until censoring or the end of the study, measured in years). Estimated parameters from the model are thus incidence rate ratios (IRRs), which can be interpreted as the average annual rate of total follow-up visits for the level of a variable, relative to the reference category for a variable.

Patient characteristic variables were run in univariate negative binomial GLM models with the outcome, and then selected variables were included in a multivariable model. The key independent variable of interest was frailty. A priori decisions were made to include some other patient characteristics in the multivariate model based on known importance in the literature (i.e., age, sex, comorbidity, year of cancer diagnosis), and clinical importance (i.e., cancer site, cancer stage, chemotherapy). Income, a measure of SES, was associated with routine cancer centre follow-up visits within a similar NS cancer survivor cohort (29). As we had three proxy measures of SES, we included only residence in the multivariate model as it had a more substantial effect size in the univariate model. Poverty and income were excluded as they were not significant at p<0.20 in their univariate models. Sensitivity analyses were conducted excluding the most extreme outliers, and removing comorbidity and age, the results remained robust. LR tests were used to indicate the significance of categorical variables in the multivariate model. Statistical significance was set to alpha level 0.05.

Objective 2.2) To identify the percentage of cancer survivors' follow-up visits provided by PCPs and determine how this differs by frailty and other patient characteristics.

First, descriptive statistics were used to describe, amongst those who had at least one visit, the percentage of all, non-frail, and frail (Table 7) cancer survivors who visited certain combinations of providers for cancer-related care during the follow-up period. Specifically, we looked at who visited specialists (i.e., only oncology, only surgery, or oncology and surgery), who visited only PCPs, and who visited both specialists and PCPs (i.e., oncology and PCP, surgery and PCP, and finally, oncology, surgery, and PCP) for cancer-related care during the follow-up period.

Based upon the distribution (Appendix E) of the continuous variable, proportion of cancer-related follow-up visits provided by PCPs, we categorized the outcome as low, medium, and high (proportion of PCP visits). Next, we described the percentage of cancer-related follow-up visits provided by PCPs for all cancer survivors (low [\leq 30%], medium [>30% <73%], and high [\geq 73%]), according to frailty and patient characteristics. We also described the proportion of cancer survivors who did not have any follow-up visits by patient characteristics (Table 8).

Each patient characteristic variable was run in a univariate multinomial model with the outcome, proportion of PCP visits (low, medium, high) (Table 9). Thus, for each patient characteristic, two models were produced. The first produced estimates of medium-high compared to low proportion of PCP visits (i.e., the odds of being in the medium-high

category compared low). The second produced estimates for being in the high category compared to low-medium (i.e., the odds of being in the high category compared to lowmedium).

We used the Stata package gologit2 (146) to perform partially proportional ordinal logistic regression for multivariate analysis, as the proportional odds assumption was not met for all variables. Gologit2 with autofit allowed us to relax this assumption in instances where the assumption was not met, while applying constraints where the assumption was met, producing a final model that overarchingly did not violate the proportional odds/parallel lines assumption (Table 10). The key independent variable of interest was frailty. The multivariate partial proportional odds model allowed us to observe variations in the estimate of the proportion of visits to PCPs by frailty while adjusting for other patient characteristics. A priori decisions were made to include patient characteristics in the multivariate model based on known importance in the literature related to specialist care and/or health care use (i.e., age, sex, comorbidity, year of cancer diagnosis), and clinical importance (i.e., cancer site, cancer stage, chemotherapy). Not specific to cancer care, SES has shown an association with health care use and specialist visitation (147). As we had three proxy measures of SES, we included poverty in the multivariate model as it had a large effect size in univariate analyses and is measured at the individual level. Residence was also included based on the hypothesis that in NS there are urban and rural differences between seeing specialists or PCPs.

We investigated the appropriateness of the final model compared to regular ordinal logistic regression and a lesser constrained model using LR tests. The tests confirmed that regular ordinal logistic regression was too restrictive (LR test, p<0.001), and that the partial proportional odds model was not too restrictive (LR test, p=0.1902). Results in Table 10 can be interpreted similar to Table 9, however, where the proportional odds assumption was met, estimates in both the three left-hand and right-hand columns are the same. No influential outliers were found. Sensitivity analysis was conducted excluding comorbidity and age, the results remained robust. LR tests were used to indicate the significance of categorical variables in the multivariate model. Statistical significance was set to alpha level 0.05.

4.4 Data Access and Ethics

The proposed research used an updated existing linked administrative dataset approved by the Nova Scotia Health (NSH) REB (REB file#1022782); an amendment to the existing file was submitted and approved. A new data access request (2021-SAM-001) was submitted requesting updated and additional data for the study as a sub-study of the supervisor's original work (2016-RAU-001) and granted by HDNS. Additionally, an amendment to the NSCR/OPIS data was requested through an NSH internal data access request entitled 'Identifying frailty in cancer survivors: Patterns of cancer follow-up care and implications for personalized survivorship models (sub-study of 'Health care utilization for survivors of breast, prostate, colon, rectal, or gynecological cancers in NS'), and granted. Identifying information was not accessed, and access to person-level (de-identified) data was obtained. The linked encrypted health card numbers were removed and replaced with unique study IDs; therefore, as per the Tri-Council Policy Statement Ethical Conduct for Research Involving Humans (TCPS 2), the study did not require individual informed consent (148). The primary ethical concern is the maintenance of confidentiality. The dataset is held on a closed HDNS server and analyses were performed within the NSH firewall. Reported results used aggregated level data with cell counts of five or greater. Per current protocols, data will be kept in a secure file behind the NSH firewall for a minimum of seven years after study completion for audit purposes. Data will be destroyed within seven years of study completion.

Chapter 5: Results

5.1 Participants Eligible for Analyses

Figure 2 outlines participants eligible for analyses. The dataset contained 20,901 Nova Scotians diagnosed with breast, prostate, colorectal, or gynecologic cancer between January 1, 2006, and December 31, 2013. Patients were excluded based on death within one year of index cancer diagnosis (n=2,608), stage IV (n=1,153) and unknown (n=490) cancer, cancer history (n=2,499), new primary cancer within one year of index cancer diagnosis (n=153), no record of IPR eligibility (n=23), gaps in IPR eligibility greater than or equal to one year (n=59), censor date before the beginning of the follow-up period (n=357), and less than two years and a half years of follow-up time (n=3,246). Further, where patient characteristic variables were missing less than 5% of data, complete case analysis was used. Participants with missing median household income (n=137) were excluded. After exclusions, the cohort included 10,176 cancer survivors. The mean age at cancer diagnosis within the cohort was 64 years (range= 7-97 years). Participants had a mean follow-up time of 7.1 years (range= 2.5-12 years). One-hundred-and-seventy-six (1.7%) of patients in the survivor cohort did not have any visits in the follow-up period; these patients were excluded only for certain analyses of objective 2.2 (Tables 7, 9 and 10), leaving 10,000 eligible patients. However, all other analyses used all 10,176 eligible cancer survivors; this difference was highlighted by using bold font in Figure 2.



Cancer Survivor Cohort

Patients remaining after exclusions (n=10,176) were censored on the following: subsequent cancer diagnosis, cancer recurrence, Insured Patient Registry eligibility end, death, and study end (Dec 31, 2018)

The rules to identify frailty in administrative data were used to classify patients as non-frail (n=8,375) or frail (n=1,801)

Patients with no cancer-related follow-up visits (n=176) were excluded only for certain analyses of objective 2.2 (Tables 7, 9 and 10), making 10,000 patients eligible for these analyses. All other analyses were conducted with all 10,176 cancer survivors



5.2 Objective 1 Results

5.2.1 Frailty Prevalence

Overall, 10,176 cancer survivors diagnosed with stage I-III breast, colorectal, gynecologic, or prostate cancer between January 01, 2006, and December 31, 2013, were included in the NS cancer survivor study cohort (Figure 2).

Amongst this cohort (n=10,176), the prevalence of frailty observed in the first five years of follow-up, was 17.7% (n=1,801, 95% CI 16.95%-18.45%). Table 1 shows the prevalence of frailty amongst the NS cancer survivor cohort by patient characteristics. The proportion of frailty increased with the number of comorbidities. More patients with stage II (18.8%, 95% CI 17.63%-19.95%) and III (18.7%, 95% CI 16.74%-20.75%) cancer at diagnosis were frail than those diagnosed at stage I (16.2%, 95% CI 15.12%-17.37%). As expected, more females (18.8%, 95% CI 17.82%-19.90%) than males (16.3%, 95% CI 15.28%-17.43%) were frail, and frailty was higher amongst those with increased age at cancer diagnosis. Further, not surprisingly, there was evidence of socioeconomic inequality. Amongst those in poverty, 23.4% (95% CI 22.06-24.70%) were frail, compared to 14.0% (95% CI 13.15%-14.90%) of people not in poverty. A trend of lower median neighbourhood income and higher frailty compared to those with higher income was also seen. Aligning with prior knowledge, a high percentage of those censored due to death were frail (64.6%, 95% CI 61.63%-67.53%). Similarly, those who were observed for a shorter follow-up period had a higher prevalence of frailty. For further context, amongst frail cancer survivors (n=1,801), 539 (29.9%) died less than five years after being identified as frail, and 262 (14.6%) died five or more years after being

deemed frail. One thousand (55.5%) of those identified as frail did not die during the study period.

Findings for cancer site and chemotherapy were slightly more unpredictable. The proportion of frailty amongst cancer sites was lowest for gynecologic survivors (13.6%, 95% CI 11.51%-15.99%) as anticipated, however, was the highest for colorectal survivors (23.0%, 95% CI 21.23%-24.76%). Breast (17.6%, 95% CI 16.39%-18.93%) and prostate survivors (15.5%, 95% CI 14.30%-16.73%) fell in the middle. Amongst those who had chemotherapy, 11.9% (95% CI 10.38%-13.62%) were frail, while 18.8% (95% CI 17.95%-19.61%) of those with no indication of chemotherapy were frail.

5.2.2 Characteristics of Frail and Non-frail Survivors

Descriptive statistics (i.e., χ^2 tests for categorical variables and t-tests for continuous variables) showed significant differences between non-frail and frail cancer survivors for all patient characteristics (i.e., comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age, residence, poverty, income, follow-up years, and censor reason), except residence (χ^2 test, p=0.364) (Table 2). There were notable differences in magnitude between the percentage of frail and non-frail cancer survivors by sex, age, poverty, neighbourhood income, and those with three or more comorbidities. Patient characteristic variables were used in univariate logistic regression models with the outcome, frailty, and results paralleled those in Table 2, with residence not significantly associated with frailty (OR 1.05, 95% CI 0.94-1.17) (Table 3).

Patient characteristics included in the multivariate logistic regression model for frailty were comorbidity, cancer site, cancer stage, chemotherapy, sex, age, and poverty (Table 3). Participants with three or more comorbidities compared to those with zero comorbidities (OR 1.96, 95% CI 1.52-2.52), older age at diagnosis (OR 1.08, 95% CI 1.07-1.08), stage II or III versus stage I cancer at diagnosis (LR test, p=0.0101), females (OR 1.35, 95% CI 1.09-1.67), and those living in poverty (OR 1.39, 95% CI 1.24-1.55) had higher odds of frailty.

Adjusted for other patient characteristics (i.e., comorbidity, cancer stage, chemotherapy, sex, age, poverty), cancer site was not significantly associated with frailty (LR test, p=0.4717). Chemotherapy was not significantly associated with frailty in the multivariate model (OR 0.91, 95% CI 0.75-1.10) (Table 3). Thus, there are differences between the summary statistics and univariate analyses compared to the *adjusted* odds of frailty for cancer site and chemotherapy.

5.2.3 Frailty Rules

Table 4 presents the results of objective 1.3, indicating the number and percentage of cancer survivors deemed frail by each frailty identification rule, for all NS cancer survivors, and by cancer site. The majority of frailty was identified through rule 3 (15.6%) related to health service use, followed by rule 1 (5.2%) indicating a person was a LTC resident, and rule 2 (2.4%) indicating receipt of palliative care. Ten-point-eight percent of cancer survivors were deemed frail through rule 3 alone. A small proportion of cancer survivors met the criteria through a combination of rules (Table 4). Within rule 3,

the most common reason for frailty was rule 3b, general health status (14.4%), which is specifically related to hospital admissions, emergency department visits, malaise/fatigue, or cachexia. This was followed by rule 3g, targeted health services use (7.7%), indicating geriatrician or in-home provider visits. Finally, 7.1% of survivors were flagged frail due to rule 3a, which represents cognitive impairment (14). Compared with the other cancer sites, colorectal cancer survivors represented the highest proportion of those identified as frail in every category except rule 3c, incontinence, which belonged to breast (4.5%). Furthermore, interestingly, colorectal cancer survivors made up the largest portion (7.4%) identified as LTC residents (rule 1).

5.3 Objective 2 Results

5.3.1 Total Cancer-related Follow-up Care

Descriptive statistics (i.e., median total visits during the follow-up period, the 25th and 75th percentiles, and IQR) described variation in visits for all patient characteristics (i.e., frailty, comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age, residence, poverty, income, follow-up years, and censor reason) (Table 5). The same variables were used in univariate logistic regression models with the outcome. Overall, results paralleled those in Table 5, with variables not significantly associated with a higher annual follow-up visit rate being poverty (IRR 1.03, 95% CI 1.00-1.06) and neighbourhood income (IRR 1.00, 95% CI 0.99-1.00) (Table 6).

Patient characteristics included in the multivariate negative binomial GLM, modelling the annual rate of cancer-related follow-up visits, were frailty, comorbidity, cancer site,

cancer stage, chemotherapy, year of cancer diagnosis, sex, age, and residence (Table 6). Cancer survivors with frailty had a higher annual cancer-related follow-up visit rate (IRR 1.28, 95% CI 1.23-1.33). Comorbidity was a significant predictor (LR test, p<0.001), although visits did not increase with the number of comorbidities. Compared with breast cancer survivors, survivors of colorectal cancer had lower visit use (IRR 0.76, 95% CI 0.72-0.80), as did gynecologic (OR 0.82, 95% CI 0.78-0.87). Chemotherapy received in the treatment period was associated with a 30% increase in the annual rate of follow-up visits (IRR 1.30, 95% CI 1.25-1.36). Each successive diagnosis (calendar) year was associated with a 4% increase in visits (IRR 1.04, 95% CI 1.04-1.05); this likely resulted as those with longer follow-up were diagnosed earlier in the study period. Sex was not significant in the multivariate model (IRR 1.05, 95% CI 0.99-1.12) and rural residence had an annual follow-up visit rate 8% lower than urban dwellers (IRR 0.92, 95% CI 0.90-095).

5.3.2 Proportion of Cancer-related PCP Follow-up Care

Table 7 describes the proportion of all, non-frail, and frail cancer survivors who had at least one follow-up visit and the different combination of providers they visited (i.e., oncology only, surgery only, oncology and surgery, PCP only, oncology and PCP, surgery and PCP, or oncology, surgery, and PCP). A higher proportion of frail cancer survivors (14.2%, 95% CI 12.57%-15.89%) compared with all survivors (8.2%, 95% CI 7.68-8.77%) and non-frail cancer survivors (6.9%, 95% CI 6.40%-7.51%), visited only PCPs for cancer-related follow-up. The most common combination of providers for all cancer survivors to see during the follow-up period was both surgeons and PCPs (54.3%,

95% CI 53.35%-55.31%). Twenty-four-point-one percent of cancer survivors saw all three provider types (i.e., oncologists, surgeons, and PCPs) for cancer-related care during the follow-up period.

Table 8 shows the number and percentage of cancer survivors by each patient characteristic (i.e., frailty, comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age, residence, poverty, income, follow-up years, and censor reason) whose proportion of cancer-related follow-up visits to PCPs was low (i.e., \leq 30%), medium (i.e., \geq 30% and <73%), or high (\geq 73%). A higher proportion of cancer survivors who were non-frail (87.3%), not hospitalized (42.8%), diagnosed later in the study period (44.0%), had prostate cancer (40.9%), were middle-aged (i.e., 51-65 years) (44.6%), males (52.3%), or not living in poverty (64.8%), had a lower proportion of their cancer-related follow-up visits to PCPs. Amongst the cancer survivor cohort, 2,485 (24.4%) were categorized as having low, 5,028 (49.4%) as medium, and 2,487 (24.4%) as having a high proportion of all cancer-related follow-up visits provided by PCPs. Onepoint-seven percent of the survivor cohort (n=176) did not have any follow-up visits. The majority of those with no visits were diagnosed later in the study period (52.8%), nonfrail (75.0%), male (51.7%), urban dwellers (56.2%), living in poverty (53.4%), who never received chemotherapy (96.6%). The percentage of frail cancer survivors classified as low (12.7%), medium (16.1%), or high (25.3%) increased across the categories, respectively. Amongst those who were categorized as high, many were female (59.6%), diagnosed with stage I (45.8%) and/or breast cancer (40.2%). The percentage of poor
cancer survivors classified as low (35.2%), medium (37.2%), or high (47.4%) increased across the categories, respectively.

Table 9 contains results of the univariate multinomial analyses of each patient characteristic with the outcome, proportion of PCP visits (low versus medium-high, and low-medium versus high). Aligning with that described in Table 8, frailty, stage I, earlier year of diagnosis, females, and poverty were associated with a higher proportion of visits to PCPs.

Patient characteristics included in the multivariate partial proportional ordinal logistic model were frailty, comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age, residence, and poverty. Table 10 presents the ORs, 95% CIs and pvalues for the estimates of low compared with medium-high proportion of PCP visits, and the estimates comparing low-medium with the high proportion of PCP visits. Where the proportional odds assumption was met (i.e., for the variables frailty, cancer stage and poverty, and for particular levels of the categorical variables comorbidity and cancer site), estimates remained the same for low (versus medium-high) and low-medium (versus high). For variables (i.e., chemotherapy, year of cancer diagnosis, sex, and age) and levels of categorical variables (i.e., comorbidity, cancer site) where the proportional odds assumption was violated, the second set of estimates were different (Table 10).

Compared to non-frail cancer survivors, frail cancer survivors had 58% greater odds of having a high proportion of PCP visits compared to low-medium, and of having a

medium-high proportion of PCP visits compared to a low proportion (OR 1.58, 95% CI 1.43-1.76). Those diagnosed at stage II or III compared to stage I had lower odds of having a medium-high or high proportion of PCP visits (OR 0.91, 95% CI 0.82-1.00 and OR 0.80, 95% CI 0.70-0.91, respectively). Comorbidity was a significant predictor (LR test, p=0.0019). Those living in poverty had 37% higher odds of having either a medium-high or high proportion of PCP visits compared to those not in poverty (OR 1.37, 95% CI 1.26-1.48). For each successive diagnosis year, the odds of having a higher proportion of visits provided by PCPs decreased; this trend held true for both medium-high (OR 0.93, 95% CI 0.91-0.95) and high (OR 0.97, 95% CI 0.95-0.99) estimates. Females compared to males had increased odds of having a greater proportion of PCP follow-up visits; this trend held true for both medium-high (OR 1.26, 95% CI 1.07-1.49) estimates.

Some trends changed considerably between multivariate analyses of the outcome by different categorization (i.e., low versus medium-high, and low-medium versus high). For example, compared to breast cancer survivors, colorectal cancer survivors had 6% lower odds of having a medium-high versus low proportion of PCP visits, although this was not significant (OR 0.94, 95% CI 0.80-1.11). However, colorectal compared to breast cancer survivors had 34% higher odds of having a high rather than low-medium amount (i.e., proportion) of PCP visits, and this finding was significant (OR 1.34, 95% CI 1.15-1.55). This may be reflective of the fact that of all breast cancer survivors (n=3,522), the majority fell into the medium category (n=1,914 [54.3%]), while 1,000 (28.4%) were categorized as high, compared to 718 (32.1%) of all colorectal cancer survivors being

categorized as high. Similarly, the finding for receipt of chemotherapy was not significant when analyzed as low versus medium-high proportion of PCP visits (OR 1.07, 95% CI 0.92-1.24). However, receipt of chemotherapy was significantly associated with 28% lower odds of having a high proportion of PCP visits (OR 0.72, 95% CI 0.62-0.84). Again, a similar trend was observed for residence, with a statistically insignificant finding for low versus medium-high proportion of PCP visits (OR 0.96, 95% CI 0.87-1.06). Yet, adjusted for other characteristics in the multivariate model, rural residents had 12% greater odds of having a high (as opposed to low-medium) proportion of PCP follow-up visits (OR 1.12, 95% CI 1.01-1.23).

5.4 Tables

5.4.1 Frailty Tables

Table 1	. Prevalence	of frailty	in the ca	ancer su	rvivor	cohort	by p	oatient	charact	eristics
(Objecti	ive 1.1).									

Variable	Ν	N (%) Frail	95% CI
	Denominator		
Clinical characteristics			
Comorbidity			
0	1,304	217 (16.6)	14.66-18.78
1	3,652	685 (18.8)	17.50-20.06
2	664	197 (29.7)	26.22-33.30
≥3	491	209 (42.6)	38.15-47.08
No hospitalization	4,065	493 (12.1)	11.14-13.17
Cancer site			
Breast	3,522	621 (17.6)	16.39-18.93
Colorectal	2,234	513 (23.0)	21.23-24.76
Gynecologic	946	129 (13.6)	11.51-15.99
Prostate	3,474	538 (15.5)	14.30-16.73
Cancer stage			
I	4,228	686 (16.2)	15.12-17.37
II	4,444	834 (18.8)	17.63-19.95
III	1,504	281 (18.7)	16.74-20.75
Chemotherapy		, , , , , , , , , , , , , , , , , , ,	
No	8,583	1,611 (18.8)	17.95-19.61
Yes	1,593	190 (11.9)	10.38-13.62
Year of cancer diagnosis			
2006-2008	3,760	780 (20.7)	19.46-22.08
2009-2010	2,505	482 (19.2)	17.71-20.84
2011-2013	3,911	539 (13.8)	12.72-14.90
Demographics	•	· · · ·	
Sex			
Male	4,635	757 (16.3)	15.28-17.43
Female	5,541	1,044 (18.8)	17.82-19.90
Age			
≤ 40 years	257	12 (4.7)	2.44-8.01
41-50 years	1,028	79 (7.7)	6.13-9.49
51-65 years	4,179	378 (9.1)	8.19-9.96
66-79 years	3,765	840 (22.3)	20.99-23.68
≥80 years	947	492 (52.0)	48.71-55.18
Residence			
Urban	6,474	1,129 (17.4)	16.52-18.39
Rural	3,702	672 (18.2)	16.92-19.43
Poverty	, , , , , , , , , , , , , , , , , , ,		

Variable	Ν	N (%) Frail	95% CI
	Denominator		
No	6,161	863 (14.0)	13.15-14.90
Yes	4,015	938 (23.4)	22.06-24.70
Neighbourhood income			
(quintiles)			
Lowest	1,977	419 (21.2)	19.41-23.06
Low-middle	2,027	381 (18.8)	17.11-20.57
Middle	2,039	353 (17.3)	15.69-19.03
Upper-middle	2,038	350 (17.2)	15.56-18.89
Highest	2,095	298 (14.2)	12.76-15.79
Other characteristics			
Follow-up years			
<5 years	2,444	549 (22.5)	20.82-24.17
5-7 years	4,099	725 (17.7)	16.53-18.89
8-12 years	3,633	527 (14.5)	13.38-15.69
Censor reason			
Study end	7,454	901 (12.1)	11.36-12.85
Subsequent cancer	719	103 (14.3)	11.85-17.10
Cancer recurrence	817	99 (12.1)	9.96-14.55
IPR end	143	24 (16.8)	11.06-23.94
Death	1,043	674 (64.6)	61.63-67.53
Total	10,176	1,801 (17.7)	16.96-18.45

CI= Confidence Interval, IPR= Insured Patient Registry.

Table 2. Patient	characteristics	amongst the cance	r survivor co	hort <i>(Objective</i>	1.2).
		0			

Variable	Total cancer	Non-frail	Frail cancer	p-
	survivor	cancer	survivors	value ^a
	cohort	survivors	N (%)	
	N (%)	N (%)		
Clinical characteristics				
Comorbidity				
0	1,304 (12.8)	1,087 (13.0)	217 (12.1)	
1	3,652 (35.9)	2,967 (35.4)	685 (38.0)	
2	664 (6.5)	467 (5.6)	197 (10.9)	
≥3	491 (4.8)	282 (3.4)	209 (11.6)	
No hospitalizations	4,065 (40.0)	3,572 (42.6)	493 (27.4)	< 0.001
Cancer site				
Breast	3,522 (34.6)	2,901 (34.6)	621 (34.5)	
Colorectal	2,234 (22.0)	1,721 (20.5)	513 (28.5)	
Gynecologic	946 (9.3)	817 (9.8)	129 (7.1)	
Prostate	3,474 (34.1)	2,936 (35.1)	538 (29.9)	< 0.001
Cancer stage				
I	4,228 (41.5)	3,542 (42.3)	686 (38.1)	
II	4,444 (43.7)	3,610 (43.1)	834 (46.3)	

Variable	Total cancer	Non-frail	Frail cancer	р-
	survivor	cancer	survivors	value ^a
	cohort	survivors	N (%)	
	N (%)	N (%)		
III	1,504 (14.8)	1,223 (14.6)	281 (15.6)	0.005
Chemotherapy				
No	8,583 (84.3)	6,972 (83.2)	1,611 (89.5)	
Yes	1,593 (15.7)	1403 (16.8)	190 (10.5)	< 0.001
Year of cancer				
diagnosis,				
mean (SD)	2009.5 (2.3)	2009.6 (2.3)	2009.1 (2.2)	< 0.001
Demographics				
Sex				
Male	4,635 (45.5)	3,878 (46.3)	757 (42.0)	
Female	5,541 (54.5)	4,497 (53.7)	1,044 (58.0)	0.001
Age, mean (SD)	64.2 (11.6)	62.5 (11.0)	71.9 (11.3)	< 0.001
Residence				
Urban	6,474 (63.6)	5,345 (63.8)	1,129 (62.7)	
Rural	3702 (36.4)	3030 (36.2)	672 (37.3)	0.364
Poverty				
No	6,161 (60.5)	5,298 (63.3)	863 (47.9)	
Yes	4,015 (39.5)	3,077 (36.7)	938 (52.1)	< 0.001
Neighbourhood income				
(\$),				
median	55,734	56,387	53,571	
IQR	25,363	25,799	24,610	
(25 th -	(45,259-	(45,426-	(43,543-	
75 th percentile)	70,622)	71,225)	68,153)	< 0.001
Other characteristics				
Follow-up years,				
mean (SD)	7.1 (2.5)	7.2 (2.5)	6.6 (2.5)	< 0.001
Censor reason				
Study end	7,454 (73.3)	6,553 (78.2)	901 (50.0)	
Subsequent cancer	719 (7.1)	616 (7.4)	103 (5.7)	
Cancer recurrence	817 (8.0)	718 (8.6)	99 (5.5)	
IPR end	143 (1.4)	119 (1.4)	24 (1.4)	
Death	1,043 (10.2)	369 (4.4)	674 (37.4)	< 0.001
Total	10,176 (100)	8,375 (100)	1,801 (100)	

^a t-tests for continuous variables, χ^2 test for categorical variables. IPR= Insured Patient Registry, IQR= Interquartile Range, SD= Standard Deviation.

Variable	Frailty					
	Univa	riate		Multivariate ^a		
	OR	95% CI	p-	OR	95% CI	p-
			value			value
Clinical characteristics	1	ſ				1
Comorbidity (reference:						<0.001 ^b
0)	1.16	0.98-1.37	0.089	1.00	0.83-1.20	0.978
1	2.11	1.69-2.64	< 0.001	1.40	1.10-1.78	0.007
2	3.71	2.95-4.68	< 0.001	1.96	1.52-2.52	< 0.001
≥3	0.69	0.58-0.82	< 0.001	0.67	0.55-0.80	< 0.001
No hospitalizations						
Cancer site (reference:						
Breast)						0.4717 ^b
Colorectal	1.39	1.22-1.59	< 0.001	0.90	0.74-1.08	0.244
Gynecologic	0.74	0.60-0.91	0.004	0.87	0.69-1.08	0.202
Prostate	0.86	0.75-0.97	0.016	0.89	0.68-1.16	0.372
Cancer stage (reference:						
Stage I)						0.0101 ^b
Stage II	1.19	1.07-1.33	0.002	1.16	1.01-1.34	0.037
Stage III	1.19	1.02-1.38	0.029	1.32	1.10-1.59	0.003
Chemotherapy	0.59	0.50-0.69	< 0.001	0.91	0.75-1.10	0.338
Year of cancer diagnosis*	0.91	0.89-0.93	< 0.001			
Demographics						
Female sex	1.19	1.07-1.32	0.001	1.35	1.09-1.67	0.007
Age (years)*	1.09	1.08-1.09	< 0.001	1.08	1.07-1.08	< 0.001
Rural residence	1.05	0.94-1.17	0.364			
Poverty	1.87	1.69-2.07	< 0.001	1.39	1.24-1.55	< 0.001
Neighbourhood income	0.99	0.99-1.00	< 0.001			
(\$)*						
Other characteristics						
Follow-up years*	0.90	0.88-0.92	< 0.001			
Censor reason (reference:						
Study end)						
Subsequent cancer	1.22	0.98-1.51	0.081			
Cancer recurrence	1.00	0.80-1.25	0.980			
IPR end	1.47	0.94-2.29	0.091			
Death	13.38	11.49-	< 0.001			
		15.35				

Table 3. Logistic regression for the association between patient characteristics and frailty (Objective 1.2).

^a Multivariate analysis adjusted for comorbidity, cancer site, cancer stage,

chemotherapy, sex, age at cancer diagnosis, and poverty.
^b Overall p-value for variables with >2 categories.
* Continuous variable.

CI= Confidence Interval, Insured Patient Registry= IPR, OR= Odds Ratio.

Frailty identification rule ^a	e ^a Cancer site				
·	Breast N=3,552	Colorec -tal N=2,234	Gynec- ologic N=946	Prostate N=3,474	Total N=10,176
	Ν	Ν	Ν	Ν	Ν
	%	%	%	%	%
1. Person was a LTC resident	200	166	25	135	526
	5.7	7.4	2.6	3.9	5.2
2. Person received palliative	59	82	16	78	235
care	1.7	3.7	1.7	2.3	2.3
3. Person met at least 2 of the	553	442	114	481	1,590
listed domains 3a to 3g	15.7	19.8	12.1	13.9	15.6
3a. Cognitive impairment	244	218	37	224	723
	6.9	9.8	3.9	6.5	7.1
3b. General health status	502	411	104	448	1,465
	14.3	18.4	11.0	12.9	14.4
3c. Incontinence (urinary or	157	72	35	124	388
fecal)	4.5	3.2	3.7	3.6	3.8
3d. Falls (with hospitalization)	98	92	19	73	282
	2.8	4.1	2.0	2.1	2.8
3e. Nutrition issues	suppress	17	suppress	23	68
	suppress	0.8	suppress	0.7	0.7
3f. Functional performance	74	72	26	73	245
	2.1	3.2	2.8	2.1	2.4
3g. Targeted health services	260	241	49	232	782
utilization	7.4	10.8	5.2	6.7	7.7
Total persons identified	621	513	129	538	1,801
_	17.6	23.0	13.6	15.5	17.7
Patients identified by each con	nbination o	of identifica	ntion rules	5	
Rule 1 only	49	36	6	26	117
	1.4	1.6	0.6	0.8	1.2
Rule 2 only	17	31	7	30	85
	0.5	1.4	0.7	0.9	0.8
Rule 3 only	379	286	91	338	1,094
	10.8	12.8	9.6	9.7	10.8
Rules 1 and 2	17	21	suppress	suppress	54
	0.5	0.9	suppress	suppress	0.5
Rules 1 and 3	149	126	17	108	400
	4.2	5.6	1.8	3.1	3.9
Rules 2 and 3	40	47	7	47	141
	1.1	2.1	0.7	1.4	1.4

Table 4. Frailty identification amongst cancer survivors by specific rule (Objective 1.3).

Frailty identification rule ^a	Cancer site							
	Breast	Colorec	Gynec-	Prostate	Total			
	N=3,552	-tal	ologic	N=3,474	N=10,176			
		N=2,234	N=946					
	Ν	Ν	Ν	Ν	Ν			
	%	%	%	%	%			
Rules 1, 2, and 3	suppress	17	suppress	12	45			
	suppress	0.8	suppress	0.4	0.4			

^a Patients with frailty identification by each individual rule are not mutually exclusive.

suppress Numbers suppressed due to small cell counts.

5.4.2 Health Care Use Tables

Table 5. Median number of total cancer-related follow-up visits during the follow-up period by patient characteristics *(Objective 2.1)*.

Variable	Total cancer-related follow-up visits					
	Median	25th; 75th	IQR			
	number visits	percentile				
Clinical characteristics	1	1				
Frailty						
Non-frail	13	8; 21	13			
Frail	14	7; 25	18			
Comorbidity						
0	15	9; 23	14			
1	15	9; 23	14			
2	11	5.5; 19	13.5			
≥3	11	5; 18	13			
No hospitalizations	12	7; 19	12			
Cancer site						
Breast	16	9; 24	15			
Colorectal	11	6; 18	12			
Gynecologic	11	7; 18	11			
Prostate	13	8; 20	12			
Cancer stage						
I	13	7; 20	13			
II	13	8; 21	13			
III	14	8; 23	15			
Chemotherapy						
No	12	7; 20	13			
Yes	18	11; 27	16			
Year of cancer diagnosis						
2006-2008	15	9; 24	15			

Variable	Total cancer-related follow-up visits					
	Median	25th; 75th	IQR			
	number visits	percentile				
2009-2010	14	8; 23	15			
2011-2013	11	6; 17	11			
Demographics			•			
Sex						
Male	12	7; 20	13			
Female	14	8; 22	14			
Age						
≤ 40 years	16	10; 27	17			
41-50 years	17	10; 27	17			
51-65 years	13	8;21	13			
66-79 years	13	7;21	14			
≥ 80 years	10	4; 17	13			
Residence						
Urban	14	8: 22	14			
Rural	12	7:20	13			
Poverty						
No	13	8:21	13			
Yes	13	7:21	14			
Neighbourhood income		,,				
(quintiles)						
Lowest	13	7:21	14			
Low-middle	12	7:20	13			
Middle	14	8:23	15			
Upper-middle	13	8; 21	13			
Highest	13	8; 21	13			
Other characteristics	L					
Follow-up years						
<5 years	10	5; 15	10			
5-7 years	13	8; 20	12			
8-12 years	17	10; 26	16			
Censor reason						
Study end	14	8; 22	14			
Subsequent cancer	12	7; 19	12			
Cancer recurrence	13	8; 21	13			
IPR end	9	6; 15	9			
Death	10	5; 19	14			
Total	13	8; 21	13			

IPR= Insured Patient Registry, IQR= Interquartile Range.

Variable	Annual rate of total cancer-related follow-up visits ^a						
	Univa	riate		Multivariate^b			
	IRR	95% CI	p-	IRR	95% CI	p-value	
			value				
Clinical characteristics							
Frail	1 20	1 15-1 24	< 0.001	1.28	1 23-1 33	< 0.001	
Comorbidity (reference: 0)	1.20	1.15 1.24	\$0.001	1.20	1.25 1.55	<0.001 <0.001°	
	0.95	0.91-0.99	0.026	0.98	0 94-1 02	0 384	
2	0.95	0.80-0.92	<0.020	0.90	0.86-0.98	0.015	
>3	0.87	0.80-0.94	< 0.001	0.92	0.86-1.00	0.019	
No hospitalizations	0.94	0.90-0.99	0.013	0.90	0.86-0.94	< 0.001	
Cancer site (reference:	0.5		01010	0.50		0.001	
Breast)						<0.001°	
Colorectal	0.77	0.74-0.80	< 0.001	0.76	0.72-0.80	< 0.001	
Gynecologic	0.82	0.78-0.86	< 0.001	0.82	0.78-0.87	< 0.001	
Prostate	0.85	0.82-0.88	< 0.001	0.95	0.88-1.02	0.163	
Cancer stage (reference:							
Stage I)						<0.001°	
Stage II	1.03	1.00-1.06	0.076	1.06	1.02-1.09	0.004	
Stage III	1.17	1.12-1.22	< 0.001	1.18	1.13-1.24	< 0.001	
Chemotherapy	1.38	1.33-1.44	< 0.001	1.30	1.25-1.36	< 0.001	
Year of cancer diagnosis*	1.03	1.02-1.03	< 0.001	1.04	1.04-1.05	< 0.001	
Demographics		•					
Female sex	1.13	1.10-1.16	< 0.001	1.05	0.99-1.12	0.096	
Age (years)*	0.99	0.99-0.99	< 0.001	0.99	0.99-0.99	< 0.001	
Rural residence	0.91	0.88-0.93	< 0.001	0.92	0.90-0.95	< 0.001	
Poverty	1.03	1.00-1.06	0.028				
Neighbourhood income	1.00	0.99-1.00	0.505				
(\$)*							
Other characteristics							
Censor reason (reference:							
Study end)							
Subsequent cancer	1.17	1.10-1.23	< 0.001				
Cancer recurrence	1.55	1.47-1.63	< 0.001				
IPR end	1.02	0.90-1.15	0.744				
Death	1.11	1.06-1.16	< 0.001				

Table 6. Negative binomial (GLM) regression for the association between patient characteristics and the annual rate of cancer-related follow-up visits (*Objective 2.1*).

^a Regression models were run with an offset, the log of the annual follow-up time (in years). Total cancer-related visits included oncology, primary care, and surgery follow-up visits.

^b Multivariate analysis adjusted for frailty, comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age at cancer diagnosis, and residence.

- ^c Overall p-value for variables with >2 categories.
- * Continuous variable.

CI= Confidence Interval, GLM= Generalized Linear Model, IPR= Insured Patient Registry, IRR= Incidence Rate Ratio.

Table 7. Provider type seen by all, non-frail, and frail cancer survivors during the follow-up period *(Objective 2.2).*

Provider type seen	Cancer survivors with visits during the follow-up period							
	All survivors	Non-frail	Frail survivors					
	N=10,000	survivors	N=1,757					
		N=8,243						
	% Patients with	% Patients with	% Patients with					
	≥1 visit (95% CI)	≥1 visit (95% CI)	≥1 visit (95% CI)					
Specialist								
Oncology only	0.5 (0.38-0.67)	0.5 (0.38-0.70)	0.5 (0.20-0.90)					
Surgery only	5.7 (5.27-6.19)	6.1 (5.57-6.62)	4.0 (3.17-5.07)					
Oncology and	0.8 (0.62-0.97)	0.9 (0.68-1.10)	0.3 (0.13-0.74)					
Surgery								
РСР								
PCP only	8.2 (7.68-8.77)	6.9 (6.40-7.51)	14.2 (12.57-15.89)					
Specialist and PCP								
Oncology and PCP	6.3 (0.59-0.68)	6.5 (6.00-7.10)	5.5 (4.45-6.63)					
Surgery and PCP	54.3 (53.35-55.31)	54.4 (53.28-55.44)	54.2 (51.82-56.53)					
Oncology, Surgery, and PCP	24.1 (23.27-24.96)	24.7 (23.77-25.65)	21.3 (19.45-23.33)					

CI= Confidence Interval, PCP= Primary care provider.

Table 8. The percentage of cancer survivors' cancer-related follow-up care attributable to PCPs (*Objective 2.2*).

Variable	% of car	No follow-up		
	Low Medium ⊥ (≤30%) (>30% <73%) (High (≥73%)	visits
	N (%)	N (%)	N (%)	N (%)
Clinical characteristics		-		-
Frailty				
Non-frail	2,169 (87.3)	4,216 (83.9)	1,858 (74.7)	132 (75.0)
Frail	316 (12.7)	812 (16.1)	629 (25.3)	44 (25.0)
Comorbidity				
0	290 (11.7)	684 (13.6)	315 (12.7)	15 (8.5)
1	835 (33.6)	1,864 (37.1)	900 (36.2)	53 (30.1)
2	194 (7.8)	262 (5.2)	184 (7.4)	24 (13.6)
≥3	103 (4.1)	196 (3.9)	182 (7.3)	10 (5.7)

Variable	% of car	No		
		provided by PCP	S	follow-up
	Low	Medium	High	visits
	(≤30%)	(>30% <73%)	(≥73%)	
	N (%)	N (%)	N (%)	N (%)
No hospitalizations	1,063 (42.8)	2,022 (40.2)	906 (36.4)	74 (42.1)
Cancer site				
Breast	565 (22.7)	1,914 (38.1)	1,000 (40.2)	43 (24.4)
Colorectal	473 (19.0)	977 (19.4)	718 (28.9)	66 (37.5)
Gynecologic	431 (17.4)	402 (8.0)	101 (4.0)	12 (6.8)
Prostate	1,016 (40.9)	1,735 (34.5)	668 (26.9)	55 (31.3)
Cancer stage				
Ι	999 (40.2)	2,014 (40.1)	1,139 (45.8)	76 (43.2)
II	1,095 (44.1)	2,249 (44.7)	1,029 (41.4)	71 (40.3)
III	391 (15.7)	765 (15.2)	319 (12.8)	29 (16.5)
Chemotherapy				
No	2,140 (86.1)	4,110 (81.7)	2,163 (87.0)	170 (96.6)
Yes	345 (13.9)	918 (18.3)	324 (13.0)	6 (3.4)
Year of cancer diagnosis				
2006-2008	793 (31.9)	1,939 (38.6)	976 (39.2)	52 (29.6)
2009-2010	599 (24.1)	1,237 (24.6)	638 (25.7)	31 (17.6)
2011-2013	1,093 (44.0)	1,852 (36.8)	873 (35.1)	93 (52.8)
Demographics	· · · · · · · · · · · · · · · · · · ·		· · · · · ·	· · · · · ·
Sex				
Male	1,300 (52.3)	2,240 (44.5)	1,004 (40.4)	91 (51.7)
Female	1,185 (47.7)	2,788 (55.5)	1,483 (59.6)	85 (48.3)
Age				
≤ 40 years	37 (1.5)	151 (3.0)	63 (2.5)	6 (3.4)
41-50 years	192 (7.7)	575 (11.4)	249 (10.0)	12 (6.8)
51-65 years	1,107 (44.6)	2,139 (42.5)	873 (35.1)	60 (34.1)
66-79 years	969 (39.0)	1,812 (36.1)	924 (37.2)	60 (34.1)
≥80 years	180 (7.2)	351 (7.0)	378 (15.2)	38 (21.6)
Residence				
Urban	1,583 (63.7)	3,281 (65.3)	1,511 (60.8)	99 (56.2)
Rural	902 (36.3)	1,747 (34.7)	976 (39.2)	77 (43.8)
Poverty				
No	1,611 (64.8)	3,159 (62.8)	1,309 (52.6)	82 (46.6)
Yes	874 (35.2)	1,869 (37.2)	1,178 (47.4)	94 (53.4)
Neighbourhood income				
(quintiles)				
Lowest	438 (17.6)	950 (18.9)	551 (22.2)	38 (21.6)
Low-middle	516 (20.8)	957 (19.0)	519 (20.9)	35 (19.9)
Middle	443 (17.8)	1,069 (21.3)	491 (19.7)	36 (20.4)
Upper-middle	554 (22.3)	980 (19.5)	463 (18.6)	41 (23.3)
Highest	534 (21.5)	1,072 (21.3)	463 (18.6)	26 (14.8)
Other characteristics				/

Variable	% of car	No		
		follow-up		
	Low Medium I		High	visits
	(≤30%)	(>30% <73%)	(≥73%)	
	N (%)	N (%)	N (%)	N (%)
Follow-up years				
<5 years	707 (28.4)	1,072 (21.3)	589 (23.7)	76 (43.2)
5-7 years	1,033 (41.6)	2,032 (40.4)	967 (38.9)	67 (38.1)
8-12 years	745 (30.0)	1,924 (38.3)	931 (37.4)	33 (18.7)
Censor reason				
Study end	1,799 (72.4)	3,767 (74.9)	1,780 (71.6)	108 (61.4)
Subsequent cancer	181 (7.3)	378 (7.5)	153 (6.1)	7 (4.0)
Cancer recurrence	271 (10.9)	413 (8.2)	126 (5.1)	suppress
IPR end	36 (1.4)	70 (1.4)	33 (1.3)	suppress
Death	198 (8.0)	400 (8.0)	395 (15.9)	50 (28.4)
Total	2,485 (100)	5,028 (100)	2,487 (100)	176 (100)

IPR= Insured Patient Registry, PCP= Primary care provider. ^{suppress} Numbers suppressed due to small cell counts.

Table 9. Univariate multinomial models for the association between patient characteristics and the proportion of PCP visits (low, medium, high) *(Objective 2.2)*.

Variable	Univariate analysis ^a							
	Low (reference) vs.			Low-Medium (reference)				
	Medium-High PCP Visits			vs. High PCP Visits				
	OR	95% CI	p-value	OR	95% CI	p-value		
Clinical characteristics								
Frail	1.63	1.43-1.86	< 0.001	1.92	1.72-2.14	< 0.001		
Comorbidity (reference: 0)								
1	0.96	0.83-1.12	0.607	1.03	0.89-1.20	0.685		
2	0.67	0.54-0.83	< 0.001	1.25	1.01-1.54	0.042		
≥3	1.07	0.83-1.37	0.625	1.88	1.50-2.35	< 0.001		
No hospitalizations	0.80	0.69-0.93	0.003	0.91	0.78-1.05	0.199		
Cancer site (reference:								
Breast)								
Colorectal	0.69	0.61-0.80	< 0.001	1.23	1.09-1.38	0.001		
Gynecologic	0.23	0.19-0.26	< 0.001	0.30	0.24-0.37	< 0.001		
Prostate	0.46	0.41-0.52	< 0.001	0.60	0.54-0.67	< 0.001		
Cancer stage (reference:								
Stage I)								
Stage II	0.95	0.86-1.05	0.294	0.81	0.74-0.90	< 0.001		
Stage III	0.88	0.77-1.01	0.061	0.73	0.63-0.84	< 0.001		
Chemotherapy	1.23	1.08-1.40	0.002	0.74	0.65-0.85	< 0.001		
Year of cancer diagnosis*	0.92	0.91-0.94	< 0.001	0.96	0.95-0.98	< 0.001		
Demographics	Demographics							

Variable	Univariate analysis ^a					
	Low (reference) vs.			Low-Medium (reference)		
	Medium-High PCP Visits			vs. High PCP Visits		
	OR 95% CI p-value		OR	95% CI	p-value	
Female sex	1.44	1.32-1.58	< 0.001	1.32	1.20-1.44	< 0.001
Age (years)*	0.99	0.99-1.00	0.076	1.02	1.01-1.02	< 0.001
Rural residence	1.00	0.91-1.10	0.954	1.19	1.08-1.30	< 0.001
Poverty	1.26	1.14-1.38	< 0.001	1.56	1.43-1.72	< 0.001
Income (\$)*	0.99	0.99-0.99	0.011	0.99	0.99-0.99	< 0.001
Other characteristics						
Follow-up years*	1.09	1.07-1.11	< 0.001	1.02	0.99-1.04	0.082
Censor reason (reference:						
Study end)						
Subsequent cancer	0.95	0.80-1.14	0.581	0.86	0.71-1.03	0.102
Cancer recurrence	0.65	0.55-0.75	< 0.001	0.58	0.47-0.70	< 0.001
IPR end	0.93	0.63-1.36	0.702	0.97	0.66-1.44	0.894
Death	1.30	1.10-1.54	0.002	2.07	1.80-2.37	< 0.001

^a Univariate models are presented in this table. Multivariate analysis for the same outcome can be found in Table 10.

* Continuous variable.

CI= Confidence Interval, OR= Odds Ratio, PCP= Primary care provider.

Table 10. Multivariate partial proportional odds for the association between patient characteristics and the proportion of PCP visits (low, medium, high) *(Objective 2.2)*.

Variable	Multivariate analysis ^a					
	Low (reference) vs.			Low-Medium (reference)		
	Medium-High PCP Visits			vs. Hi	gh PCP Vis	its
	OR	95% CI	p-value	OR	95% CI	p-value
Clinical characteristics						
Frail ^b	1.58	1.43-1.76	< 0.001	1.58	1.43-1.76	< 0.001
Comorbidity (reference: 0)			0.0019 ^c			0.0019 ^c
1 ^b	0.95	0.83-1.07	0.388	0.95	0.83-1.07	0.388
2	0.72	0.58-0.88	0.002	0.98	0.79-1.21	0.826
≥3 ^b	1.23	0.99-1.52	0.052	1.23	0.99-1.52	0.052
No hospitalizations ^b	1.00	0.89-1.14	0.954	1.00	0.89-1.14	0.954
Cancer site (reference:						
Breast)			< 0.001°			< 0.001°
Colorectal	0.94	0.80-1.11	0.453	1.34	1.15-1.55	< 0.001
Gynecologic	0.22	0.19-0.26	< 0.001	0.29	0.25-0.37	< 0.001
Prostate ^b	0.74	0.61-0.89	0.002	0.74	0.61-0.89	0.002
Cancer stage (reference:						
Stage I)			0.0031 ^c			0.0031 ^c
Stage II ^b	0.91	0.82-1.00	0.052	0.91	0.82-1.00	0.052

Variable	Multivariate analysis ^a					
	Low (reference) vs.			Low-	Medium (re	ference)
	Medium-High PCP Visits			vs. High PCP Visits		
	OR	95% CI	p-value	OR	95% CI	p-value
Stage III ^b	0.80	0.70-0.91	0.001	0.80	0.70-0.91	0.001
Chemotherapy	1.07	0.92-1.24	0.356	0.72	0.62-0.84	< 0.001
Year of cancer diagnosis*	0.93	0.91-0.95	< 0.001	0.97	0.95-0.99	0.006
Demographics						
Female sex	1.44	1.21-1.72	< 0.001	1.26	1.07-1.49	0.006
Age (years)*	0.99	0.98-0.99	< 0.001	1.00	1.00-1.01	0.035
Rural residence	0.96	0.87-1.06	0.451	1.12	1.01-1.23	0.025
Poverty ^b	1.37	1.26-1.48	< 0.001	1.37	1.26-1.48	< 0.001
Neighbourhood income						
(\$)*						
Other characteristics						
Follow-up years*						
Censor reason (reference:						
Study end)						
Subsequent cancer						
Cancer recurrence						
IPR end						
Death						

^a Multivariate analysis adjusted for frailty, comorbidity, cancer site, cancer stage, chemotherapy, year of cancer diagnosis, sex, age at cancer diagnosis, residence, and poverty.

^b Indicates where the proportional odds assumption was met.

^c Overall p-value for variables with >2 categories.

* Continuous variable.

CI= Confidence Interval, IPR= Insured Patient Registry, OR= Odds Ratio, PCP= Primary care provider.

Chapter 6: Discussion

This study was the first to describe frailty in a cancer survivor population and examine their health services using administrative data. We found that 17.7% of breast, colorectal, gynecologic, and prostate cancer survivors in the study's NS population-based cohort were frail in their first five years of the follow-up period. Some clinical characteristics (i.e., more comorbidities and later cancer stage at diagnosis) were significantly associated with higher odds of frailty. Although, demographic characteristics were particularly important, as female sex, older age, and poverty were associated with greater odds of frailty. These findings were not surprising and are supported by prior literature.

Interestingly, when adjusted for other patient characteristics in the multivariate model, cancer site was not significantly associated with frailty. Regardless, the prevalence of frailty was highest amongst colorectal cancer survivors. Overall, the findings suggest frailty is prevalent amongst NS cancer survivors, at least within the study cohort, and highlight certain characteristics of patients who may be more susceptible to frailty. Further, frail cancer survivors had high health care usage in the follow-up period with a high proportion of visits to PCPs, which may suggest increased needs related to the late and long-term effects of cancer. From a health system planning perspective, these findings could increase awareness regarding the types of patients that may require additional support during follow-up and may benefit from a multidisciplinary model of follow-up care. There may be a need to direct adequate resources to community providers and potentially build multidisciplinary primary care teams to meet the complex needs of frail cancer survivors.

6.1 Comparing Findings to Literature

Studies that have estimated the prevalence of frailty in cancer survivor populations range from approximately 7.9-59% (149). This study found 17.7% of its cancer survivor cohort to be frail within the five-year follow-up period. One cross-sectional US survey of community-dwelling older adults 50 years and older found 16.1% of cancer survivors in their sample to be frail. Their sample included individuals who had been diagnosed with cancers of the breast, colon, prostate, or lung, lymphoma, or leukemia (150). Another cross-sectional US study, which used clinical measures and questionnaires, found that 18% of breast cancer survivors aged 53-87 years, with a mean of 5-7 years posttreatment, were frail (151). Our study was not limited to older adults; it included cancer survivors ranging from age 7-97 years. The proportion of frail cancer survivors in our study 40 years or younger was 4.7%. Frailty is not limited to older cancer survivors. Hayek et al. (2020) conducted a retrospective cohort study in the US using clinical data and questionnaires. They found frailty in childhood cancer survivors who were five or more years post-diagnosis to be three times higher compared with siblings (152). Five years is the period of typical follow-up care. This period may represent a time when needs are higher, and survivors are more vulnerable. While our study estimates frailty prevalence only within the five-year follow-up window, there are mixed findings regarding the prevalence of frailty amongst cancer survivors for longer follow-up periods. For example, one study found an inverse association between frailty and greater than 10 years elapsed since cancer diagnosis (133). Our findings reinforce the need to support cancer survivors and monitor for frailty, particularly in the five-year follow-up period.

Sociodemographic factors associated with frailty have been described in the literature. Our study demonstrated some of these expected findings, such as increased odds of frailty in the female sex, which has been previously described in the literature as related to estrogen deficiency (149,152). We also found higher odds of frailty amongst those of lower SES (i.e., poverty). While evidence is lacking specifically related to frailty in cancer survivors and SES, outside of this realm, as described in a recent systematic review, frailty is associated with lower SES (153). The associations between frailty, increasing age, and comorbidities have been well documented, including within cancer survivor populations (150,152); this study aligns with those findings.

The variation in frailty by clinical characteristics was not as conspicuous compared to its variation by demographic characteristics. Although, as expected, those with stage III cancer at diagnosis had higher odds of frailty than those with stage I. Smitherman et al. (2018) showed a similar finding (135). However, compared to breast cancer survivors, survivors of colorectal, gynecologic, and prostate had lesser odds of frailty when adjusted for other patient characteristics, though the associations were non-significant. Similarly, two studies that included survivors of various cancers did not find cancer site to be significantly associated with higher frailty (133,135). Even among studies conducted with single cancer sites, estimates of prevalence have varied. For example, amongst two studies with breast cancer survivors, one found a frailty prevalence of 5.1% (154) and the other 18% (151). Similarly, a study by Winters-Stone et al. (2017) identified a prevalence of frailty greater than 20% for prostate cancer survivors (155), contrasting with the findings of Bylow et al. (2011) who identified a frailty prevalence of less than 10% for

prostate survivors (156). Such variation suggests that using cancer site alone is likely not a great way to predict survivors who may become frail. Although descriptive data and univariate analysis showed a higher proportion of frailty amongst colorectal cancer survivors, this finding was dissolute in the multivariate model after accounting for other characteristics. It may be that other demographic factors found amongst colorectal survivors were contributing to frailty within this group.

Receipt of chemotherapy was associated with lower odds of frailty, though this finding was non-significant in the multivariate model. Within the population-based cohort, 15.7% of participants received chemotherapy during the treatment period. This number aligns with the cancer sites included in this study (e.g., chemotherapy is not standard of care for low-risk prostate cancer patients or for early stage [stage I-II], low-risk colon cancer patients); however, it is likely an underestimation of chemotherapy receipt due to missing data (i.e., no information on oral chemotherapy) (157). Some other studies have found cancer treatments to be associated with frailty (152,158), while others have not (151,159). The possible underestimation of chemotherapy and/or the particular cancer sites of survivors represented in this study may have contributed to the lack of association between chemotherapy and frailty. These findings suggest that sociodemographic factors rather than disease-specific characteristics are important for frailty identification, although those with more severe disease certainly have higher odds of frailty. Further research should be conducted to explore frailty amongst various cancer sites and treatment modalities.

Frail cancer survivors in the study's cohort were most commonly identified through rule 3 for identification of frailty in administrative data. Urquhart et al. (2017) found that rule 2 identified the largest proportion of frailty within an NS community-dwelling older adult decedent population; they were unable to calculate frailty for NS living persons due to issues with data availability (14). The health care usage that flagged frailty according to rule 3 is likely related to the late and long-term effects survivors seek care for in the follow-up period. Seven-point-one percent of cancer survivors were flagged as frail related to cognitive impairment. While the data does not contain this level of detail, it is possible this is related to 'brain or chemo' fog following cancer treatment. Related, Mandleblatt et al. (2021) described a self-reported decline in cognition as cancer survivors grew frail (160). To our knowledge, this study was the first to use these rules specifically within a cancer survivor population. Future research could use these rules amongst other Canadian cancer survivor populations.

We found that frail cancer survivors compared to non-frail cancer survivors in the study cohort had a higher annual cancer-related follow-up visit rate. This finding relates to existing literature, which states the needs of frail individuals' often go unmet (15), consequently, they continue to seek health care. Similarly, a 2019 US retrospective study of cancer patients after urologic surgery found that increasing frailty was associated with increasing health care use after surgery, including prolonged hospital stay, referrals for continuing care, and hospital re-admission (161).

This study found that each successive year of cancer diagnosis, higher stage and younger age at cancer diagnosis were significantly associated with more follow-up visits, aligning with the findings of follow-up visits examined in a similar cohort of NS cancer survivors (29). Differing from findings of this similar NS cohort of cancer survivors, neighbourhood income was not a significant predictor and rural residence was significantly associated with less follow-up visits in our study. Further, we found that compared to breast cancer survivors, survivors of gynecologic cancer had significantly less, not more visits (29). However, there were notable differences between these two cohorts that likely explain the variation. Firstly, our study used a minimum follow-up observation inclusion criterion of 2.5 years, due to our frailty assessment. Secondly, we examined not only cancer centre visits, but cancer-related visits, including those to PCP. Finally, we had a longer period of follow-up data and thus, observed visits over a longer time period.

In this study, we established that a high percentage of non-frail and frail NS cancer survivors saw both a surgeon and a PCP for follow-up care, although it was higher for those who were non-frail. Another NS study found breast cancer survivors had greater surgeon follow-up compared to other Canadian provinces (30). We also found that frail compared to non-frail cancer survivors more commonly saw only PCPs. Furthermore, frailty was associated with a higher proportion of PCP visits. Similarly, prior literature not specific to cancer care, found frailty to be associated with increased PCP use (162).

Other patient characteristics within our study cohort associated with a medium-high or a high proportion of PCP visits included female sex, poverty, breast and colorectal cancer survivors, and stage I cancer at diagnosis. Urguhart et al. (2020) examined factors associated with primary care use during follow-up care within a similar NS cancer survivor cohort (163). They found female sex significantly associated with increased annual PCP use when analyzed for all PCP visits, but not for cancer-specific PCP visits. In their study, neighbourhood income was significant at a null value. Our study found poverty was significantly associated with higher odds of a medium-high or high proportion of all cancer-related follow-up visits provided by PCPs. Urguhart et al. (2020) found stage II or III compared to stage I cancer was associated with a higher annual cancer-specific PCP visit rate. They also found that prostate cancer survivors had more cancer-specific PCP visits per year, and breast cancer survivors had more annual PCP visits for all reasons, compared to colorectal and gynecologic survivors (163). Our study found that considering all cancer-related follow-up visits, patients diagnosed with stage I cancer, or with breast or colorectal cancer, had higher odds of having a high proportion of those visits provided by PCPs. While our study found that each unit increase in the year of cancer diagnosis was associated with lower odds of a higher proportion of PCP visits, Urquhart et al. (2020) found each successive diagnosis year associated with a higher annual PCP visit rate (163). This is likely reflective of the different outcomes measured; in our study, the proportion of PCP visits, and in Urquhart et al. (2020), the annual PCP visit rate. Thus, our result likely reflects that those diagnosed later in the study period were in a more immediate follow-up period, likely still seeing specialists more often than those diagnosed earlier (i.e., where more time had passed). Patients diagnosed earlier

likely needed fewer specialist visits as time went on and/or were discharged to primary care for remaining follow-up.

6.2 Limitations

The study is not without limitations. The following discussion of the study's limitations centres around challenges using administrative data, including data availability and measurement limitations of variables.

Administrative data are not captured for research purposes. While they represent a powerful and efficient means to study healthcare utilization, there are known limitations to the use of administrative data. MSI Physician Billings data exists to pay physicians and provide accountability for their work; therefore, it is expected that these data are accurate descriptions of physicians' services rendered. However, all diagnostic codes may not have been recorded, and there is no way to tell what codes, if any, are missing. All physician visits may not be represented. Shadow billing has often been a concern with the use of physician claims data. Although we acknowledge this may have under-reported services provided, at least two recent Canadian studies, from MB and Alberta (AB), have reported high levels of accuracy in reported physician claims data (164,165). Not having a family doctor is a likely barrier to discharge from the cancer clinic. As such, those patients may be more likely to receive ongoing specialist care; this may create a high estimate of specialist care use. Our data are limited because we were unable to identify those who did not have a PCP or those who lost their PCP or specialist provider during follow-up. However, the proportion of patients in our cohort who did not have at least

one PCP visit was small (8.6%), suggesting most patients found a way to access PCP services whether or not they had a usual PCP.

We only examined cancer-related visits provided by medical oncologists, radiation oncologists, PCPs, general surgeons, obstetrician/gynecologists, gastroenterologists, and urologists. Therefore, we did not capture health services used from other specialty visits. However, this was done purposefully to achieve a narrow focus on cancer-related followup. We used a known set of diagnostic codes to determine cancer-related visits within MSI Physician Billings (Appendix C) and a previously used decision-rule for OPIS (29). However, we did include other cancer-related and mental health codes that represent issues commonly experienced by survivors in follow-up (e.g., depression, nausea and vomiting), though there is no way to tell if these particular codes were certainly related to a cancer diagnosis (Appendix C). Despite the possibility of overestimating cancer-related visits, the act of specifying these codes is one that few follow-up studies have done. In order to maintain a relatively narrow focus and due to feasibility, we did not explore noncancer-related visits. Due to the complexities of care and cancer-related issues that may be addressed by other providers for frail and/or non-frail cancer survivors, it may be beneficial for future studies to explore cancer survivor patterns in non-cancer-related visits, for example, to geriatricians, other specialists, or mental health providers.

There is no gold standard to measure frailty and measuring frailty in administrative data poses some challenges. Although the rules to identify frailty have been tested and have face validity from expert geriatricians, they do not replace clinical information.

Moreover, this measure of frailty captures physical and cognitive aspects but not psychosocial factors. The rules to identify frailty thus require further validation. Further, individuals may experience varying degrees of frailty (e.g., prefrail, moderate, severe), which we were not able to delineate. However, I would argue that for population-level frailty identification, the categorization of non-frail and frail is sufficient as a starting point for health services insight. Certainly, these decision rules do identify a prevalence of frailty similar to that reported by others. Furthermore, the rules of frailty and data presented allow the reader to see in which domains an individual was deemed frail (Table 4). Additionally, the identification rules focus on specificity over sensitivity; therefore, some frail individuals may not have been identified (14).

The estimate of frailty prevalence in this study represents a period prevalence of NS breast, colorectal, gynecologic, and prostate cancer survivors diagnosed between January 01, 2006, and December 2013, during the first five years of follow-up. However, our calculation is based upon specific inclusion/exclusion criteria and does not include a group of long-term survivors in NS diagnosed before the study period who were still survivors at that time. Furthermore, it is important to note when interpreting findings, results are limited in their comparison to today's current patients, as less chemotherapy, less invasive surgeries, and more immunotherapy and targeted treatments are being utilized. Additionally, the cohort does not include those with stage IV disease. Although small numbers, this means we did not identify some stage IV patients who may have converted to curable status and had long-term survival. Therefore, the prevalence of

frailty may be underestimated, and readers should not extrapolate the findings far beyond the study cohort.

The study design does not fit the traditional definition of a retrospective cohort study or a cross-sectional study due to the follow-up data. Although a cross-sectional design is ideal for determining prevalence, this study is population-based, and thus includes NS cancer survivors of stage I-III breast, colorectal, gynecologic, and prostate cancers during the study period. For analysis purposes, data were treated as cross-sectional. Being cross-sectional, consequently, temporality between variables cannot be determined; thus, no causal inferences can be made. Finally, a critique of estimating frailty using administrative data is the overall risk for overestimation. However, I feel the study took steps to mitigate this (i.e., the rules favoured specificity over sensitivity, we used a minimum observation period, and described the proportion of those who died less than five or five or more years after becoming frail). In an effort to counteract the risk of overestimation, we may have underestimated the prevalence of frailty. However, if this is the case, the frailty prevalence of 17.7% is still a considerable proportion, given the prudent measurement.

As the measure of comorbidity used was from hospital discharge records (i.e., CIHI DAD) and no outpatient records, bias was introduced. The amount of comorbidity is likely underestimated as we are missing those with indicators of comorbidity found in outpatient records. Forty percent of individuals had no hospitalization records; thus, comorbidity results must be interpreted with caution. We removed patients with missing

data for the neighbourhood income variable (1.26%). However, complete case analysis wherein less than 5% of independent variable data is missing has been cited as having a low risk for bias when there are no clear systematic differences in those missing (166).

There are other variables or components of concepts that we may not have captured due to the nature of administrative data and/or availability within our specific datasets. For example, it would have been beneficial to observe more treatment information, such as hormone therapy, radiation, surgery, immunotherapy and targeted treatments, and drug information (e.g., oral chemotherapy). For example, codes indicating the type of surgery a patient had for treatment of their cancer would have been useful. Some surgical procedures are less invasive (e.g., polypectomy for early stage I colon cancer) compared to others (e.g., partial colectomy for stage III colon cancer). Recovering from an invasive surgery is likely more taxing on the body, and thus this information could have been particularly useful for the examination of frailty within this study. More specific cancer type information (e.g., triple-negative breast cancer) could have provided more context, especially related to health care use. Given there are racial disparities in cancer-related outcomes along the cancer care continuum (167), race would have been an important covariate to explore. However, race-based data are not available in existing administrative health data.

6.3 Strengths, Research and Policy Implications

Despite its limitations, the study has several strengths. They include the benefits of using administrative data, its novelty, the period for examination of follow-up care, the measure

of routine follow-up visits, and the rules to identify frailty within administrative data. Further, this systematic identification of frailty is useful from a health systems perspective for resource planning and future care models.

The benefit of using a population-based administrative dataset is that the study did not have to rely on self-reported data or introduce recall or selection bias associated with that type of data. Using administrative data allowed the linkage of various data sources and information, and a larger study population than would be otherwise possible. To our knowledge, this study represents the first-ever systematic identification of frailty using solely administrative data in a Canadian cancer survivor population. Given the overall prevalence of frailty (17.7%) found within the study cohort and its spread across cancer sites, it would be beneficial to ensure oncology providers are aware of this information, to prompt further clinical assessment of frailty and referrals for support amongst patients as needed, and to achieve early identification of frailty within the follow-up period. It will also be important to ensure that providers caring for this population have training in frailty care.

This study described patterns of follow-up care in a time period after Choosing Wisely Canada made recommendations (in 2015) to use PCPs where possible for survivorship follow-up care (1). Research following this recommendation is important to see if those efforts have made an impact in practice. This may prompt decision-makers to re-evaluate care pathways. Additionally, this study provides a more precise measure of follow-up visits. Unlike many prior studies, it differentiated between cancer-related and non-cancer-

related visits and included oncology and PCP follow-up visits, capturing a broader range of visits. This study shows that cancer survivors in NS are frequenting PCPs, though many continue to see specialists, oncologists, and particularly surgeons. However, frail cancer survivors do have less specialist cancer follow-up than non-frail survivors. Additionally, a higher proportion of frail cancer survivors compared to non-frail survivors visited only PCPs for cancer-related care during the follow-up period. This points to movement from specialist-led to primary-focused cancer follow-up. However, due to the volume of visits, particularly amongst frail survivors, it may point to a need for ensuring robust structures and care models equipped to care for patients with increased and complex needs. Primary care has been associated with reduced hospitalizations and increased satisfaction of care in frail patients. Primary care models tested in frail populations include, integrated care, shared care, home-based care, and family medicine speciality clinics (168). While within this study sufficient data was not available to examine the role of Advanced Practice Nurses (APNs) in follow-up care, their role in oncology care has been discussed in the literature. APNs have begun to aid in the care of older adult cancer patients across the cancer continuum. When practicing to their full scope, APNs could aid in enhancing communication across care transitions, survivorship care planning, and contribute to Comprehensive Geriatric Assessment, which would help the care team identify and manage frailty (169). Key stakeholders (i.e., decision and policy-makers, PCPs, researchers, and oncology specialists) may want to explore how community oncology care or multidisciplinary primary care models could benefit this group.

The rules to identify frailty in administrative data incorporate health service utilization in addition to physical and cognitive aspects of frailty. The rules use ICD9/10 codes and were purposefully built using only primary databases (CIHI DAD and MSI Physician Billings); thus, these data are widely available in other Canadian provinces, which makes this study easily replicable in other Canadian provinces.

6.4 Future Research Directions

This study lays a foundation on which to further investigate frailty within cancer survivors, as well as their health care use, both in the NS and Canadian contexts. Future studies should use the rules to identify frail survivors in other provinces as followup care practices vary by province (29).

As a large proportion of frail survivors utilized PCPs during follow-up, future studies may benefit from recruiting cancer survivors through PCP clinics for clinical frailty assessment. This could be coupled with qualitative interviews to identify what supports and interventions frail survivors would find beneficial or examine the differences in needs voiced by frail and non-frail individuals. This may also help tease out differences in needs between people with different cancer sites. Furthermore, it is unknown how information is actually being shared between providers. It would be beneficial to capture the continuity of care both in administrative data and qualitative interviews (e.g., patient perspectives of their continuity of care and transition from or between care of PCPs and specialists). Researchers should engage key stakeholders in cancer and primary care, including patients, to determine their perspectives on the model of care best suited for

frail cancer survivors, along with its potential challenges; these data could inform the selection of a care model to test within this population. The study's findings demonstrate differences between frail and non-frail cancer survivors, suggesting frailty may be useful in personalizing follow-up care. Given the complex needs frail people often have, multidisciplinary primary care teams could benefit frail survivors (e.g., social workers to aid with socioeconomic challenges, physiotherapists to rebuild muscle strength, occupational therapists to work on cognitive function, etc.). Researchers should involve oncology specialists and PCPs in planning and testing such transitions in care.

6.5 Conclusion

This study was the first to quantify frailty in a Canadian cancer survivor population. In the NS breast, colorectal, gynecologic, and prostate cancer survivor cohort, frailty was prevalent at 17.7%. While disease severity is relevant, sociodemographic factors remain particularly important for frailty assessment. Compared to non-frail survivors, frail survivors had high usage of follow-up care and a higher proportion of cancer-related follow-up visits to PCPs. SES differences amongst survivors were observed in health care use, particularly in regard to higher visit use and proportion of PCP visits. The findings suggest frail cancer survivors are being cared for by PCPs, but they require a high amount of support (visits). Moving forward, more research should be conducted with cancer survivors and key stakeholders in primary and oncology care to find and test alternative care models well suited to meet the needs of this population.

6.6 Knowledge Translation

Prior to data analysis, the initial research plan was presented in the form of a poster presentation at the Department of Community Health and Epidemiology 2020 Research Day, the 2021 Canadian Centre for Applied Research in Cancer Control conference, and the 2021 NSH Department of Medicine Research Day. I have written lay summaries to explain the research proposal to the general public for awarding funding agencies, the Maritime SPOR SUPPORT Unit, the Canadian Institutes of Health Research, and the Canadian Centre for Applied Research in Cancer Control.

Moving forward for this single study, the goal of the end-of-grant knowledge translation plan is to disseminate the findings to interested audiences. I will work with the research supervisor to identify opportunities to disseminate the research findings in summaries/abstracts and/or conferences that will reach oncology providers, researchers, and stakeholders in both cancer and frailty care. Additionally, I will prepare and submit a manuscript to a relevant cancer care journal.

References

- 1. Mitera G, Earle C, Latosinsky S, Booth C, Bezjak A, Desbiens C, et al. Choosing wisely Canada cancer list: ten low-value or harmful practices that should be avoided in cancer care. J Oncol Pract. 2015;11(3):e296–303.
- 2. Smith BD, Smith GL, Hurria A, Hortobagyi GN, Buchholz TA. Future of cancer incidence in the United States: burdens upon an aging, changing nation. J Clin Oncol. 2009;27(17):2758–65.
- Smith L, Cancer Society C, John S, Ryan Woods L, Brenner D, Bryan S, et al. Canadian cancer statistics 2019. Toronto, ON: The Canadian Cancer Society; 2019. 1-92 p.
- 4. Siegel R, DeSantis C, Virgo K, Stein K, Mariotto A, Smith T, et al. Cancer treatment and survivorship statistics, 2012. CA Cancer J Clin. 2012;62(4):220–41.
- 5. McCabe MS, Partridge AH, Grunfeld E, Hudson MM. Risk-based health care, the cancer survivor, the oncologist, and the primary care physician. Semin Oncol. 2013;40(6):804–12.
- 6. Oeffinger KC, McCabe MS. Models for delivering survivorship care. J Clin Oncol. 2006;24(32):5117–24.
- 7. Hewitt M, Greenfield S, Stovall E. From cancer patient to cancer survivor.: lost in transition. Washington, DC: The National Academies Press; 2006. 1–506 p.
- 8. Grunfeld E, Levine MN, Julian JA, Coyle D, Szechtman B, Mirsky D, et al. Randomized trial of long-term follow-up for early-stage breast cancer: a comparison of family physician versus specialist care. J Clin Oncol. 2006;24(6):848–55.
- 9. Wattchow DA, Weller DP, Esterman A, Pilotto LS, Mcgorm K, Hammett Z, et al. General practice vs surgical-based follow-up for patients with colon cancer: randomised controlled trial. Br J Cancer. 2006;94:1116–21.
- Grunfeld E, Mant D, Yudkin P, Adewuyi-Dalton R, Cole D, Stewart J, et al. Routine follow up of breast cancer in primary care: randomised trial. BMJ. 1996;313(7058):665–9.
- 11. Murchie P, Nicolson MC, Hannaford PC, Raja EA, Lee AJ, Campbell NC. Patient satisfaction with gp-led melanoma follow-up: a randomised controlled trial. Br J Cancer. 2010;102:1447–55.
- 12. Emery JD, Jefford M, King M, Hayne D, Martin A, Doorey J, et al. Procare trial: a phase II randomized controlled trial of shared care for follow-up of men with

prostate cancer. BJU Int. 2017;119(3):381-9.

- 13. Erikson C, Salsberg E, Forte G, Bruinooge S, Goldstein M. Future supply and demand for oncologists : challenges to assuring access to oncology services. J Oncol Pract. 2007;3(2):79–86.
- 14. Urquhart R, Giguere AC, Lawson B, Kendell C, Holroyd-Leduc JM, Puyat JH, et al. Rules to identify persons with frailty in administrative health databases. Can J Aging / La Rev Can du Vieil. 2017;36(4):514–21.
- 15. Giguere AC, Farmanova E, Holroyd-Leduc JM, Straus SE, Urquhart R, Carnovale V, et al. Key stakeholders' views on the quality of care and services available to frail seniors in Canada. BMC Geriatr. 2018;18(1):290.
- Boutette M, Hoffer A, Plant J, Robert B, Sinden D. Establishing an integrated model of subacute care for the frail elderly. Healthc Manag Forum. 2018;31(4):133–6.
- 17. Kent EE, Arora NK, Rowland JH, Bellizzi KM, Forsythe LP, Hamilton AS, et al. Health information needs and health-related quality of life in a diverse population of long-term cancer survivors. Patient Educ Couns. 2012;89(2):345–52.
- 18. Kenzik KM. Health care use during cancer survivorship: review of 5 years of evidence. Cancer. 2019;125(5):673–80.
- 19. Grunfeld E. Looking beyond survival: how are we looking at survivorship? J Clin Oncol. 2006;24(32):5166–9.
- 20. Watson EK, Rose PW, Neal RD, Hulbert-williams N, Donnelly P, Hubbard G, et al. Personalised cancer follow-up: risk stratification, needs assessment or both? Br J Cancer; London. 2012;106(1):1–5.
- 21. Kendell C, Lawson B, Puyat JH, Urquhart R, Kazanjian A, Johnston G, et al. Assessing the quality of care provided to older persons with frailty in five Canadian provinces, using administrative data. Can J Aging. 2020;39(1):52–68.
- 22. Brenner DR, Poirier A, Woods RR, Ellison LF, Billette JM, Demers AA, et al. Projected estimates of cancer in Canada in 2022. CMAJ. 2022;194(17):E601–7.
- 23. The National Cancer Institute. Risk factors for cancer [internet]. The National Cancer Institute; 2015 [cited 2020 Jun 11]. Available from: https://www.cancer.gov/about-cancer/causes-prevention/risk
- 24. Canadian Cancer Society. Cancer statistics at a glance [internet]. www.cancer.ca. 2020 [cited 2020 Jun 11]. Available from: https://www.cancer.ca:443/en/cancer-in formation/cancer-101/cancer-statistics-at-a-glance/?region=on

- 25. Chaput G, Giudice ME Del, Kucharski E. Cancer screening in Canada: what's in, what's out, what's coming. Can Fam Physician. 2021;67(1):27.
- 26. Canadian Partnership Against Cancer. Canada leading developed countries in survival for lung and colon cancer [internet]. Canadian Partnership Against Cancer; 2022 [cited 2022 Nov 22]. Available from: https://www.partnershipagainst cancer.ca/news-events/news/article/survival-rates-canada/
- 27. National Comprehensive Cancer Network. Managing cancer as a chronic condition [internet]. National Comprehensive Cancer Network; 2020 [cited 2020 Aug 28]. Available from: https://www.nccn.org/patients/resources/life_after_cancer/managing.aspx
- 28. Cancer.Net. What is survivorship? [internet]. American Society of Clinical Oncology; 2010 [cited 2020 Jun 11]. Available from: https://www.cancer.net/surv ivorship/what-survivorship
- 29. Urquhart R, Lethbridge L, Porter GA. Patterns of cancer centre follow-up care for survivors of breast, colorectal, gynecologic, and prostate cancer. Curr Oncol. 2017;24(6):360–6.
- 30. Kendell C, Decker KM, Groome PA, McBride ML, Jiang L, Krzyzanowska MK, et al. Use of physician services during the survivorship phase: a multi-province study of women diagnosed with breast cancer. Curr Oncol. 2017;24(2):81–9.
- 31. Urquhart R, Folkes A, Porter G, Kendell C, Cox M, Dewar R, et al. Populationbased longitudinal study of follow-up care for patients with colorectal cancer in Nova Scotia. J Oncol Pract. 2012;8(4):246–52.
- 32. Sullivan J, Thornton Snider J, van Eijndhoven E, Okoro T, Batt K, DeLeire P. T. The well-being of long-term cancer survivors. AJMC. 2018;24(4):188–95.
- Stein KD, Syrjala KL, Andrykowski MA. Physical and psychological long-term and late effects of cancer [Internet]. Vol. 112, Cancer. John Wiley & Sons, Ltd; 2008 [cited 2020 Oct 20]. p. 2577–92. Available from: https://acsjournals-onlineli brary-wiley-com.ezproxy.library.dal.ca/doi/full/10.1002/cncr.23448
- National Cancer Institute. Coping late effects [internet]. National Cancer Institute;
 2021 [cited 2022 Nov 22]. Available from: https://www.cancer.gov/about-cancer/ coping/survivorship/late-effects
- 35. Livestrong. Second cancers [internet]. Livestrong; n.d. [cited 2022 Dec 8]. Available from: https://www.livestrong.org/we-can-help/healthy-living-after-treat ment/second-cancers
- Costanzo ES, Ryff CD, Singer BH. Psychosocial adjustment among cancer survivors: findings from a national survey of health and well-being. Heal Psychol. 2009;28(2):147–56.
- 37. Ussher JM, Wong WKT, Perz J. A qualitative analysis of changes in relationship dynamics and roles between people with cancer and their primary informal carer. Health (Irvine Calif). 2011;15(6):650–67.
- 38. Arrington MI. "She's right behind me all the way": an analysis of prostate cancer narratives and changes in family relationships. J Fam Commun. 2005;5(2):141–62.
- 39. Nova Scotia Health Cancer Care Program. Cancer surveillance guidelines [internet]. Nova Scotia Health; n.d. [cited 2021 Oct 28]. Available from: http://www.cdha.nshealth.ca/nova-scotia-cancer-care-program-4
- 40. National Comprehensive Cancer Network. Treatment by cancer type [internet]. National Comprehensive Cancer Network; 2021 [cited 2021 Sep 28]. Available from: https://www.nccn.org/guidelines/category_1
- 41. Gradishar W, Moran M, Abraham J, Aft R, Agnese D, Allison KH, et al. Nccn clinical practice guidelines in oncology breast cancer nccn evidence blocks tm nccn guidelines version 8.2021 [internet]; 2021 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/breast_blocks.pdf
- 42. Benson AB, Venook AP, Al-Hawary MM, Azad N, Chen Y-J, Ciombor KK, et al. Nccn clinical practice guidelines in oncology colon cancer nccn evidence blocks tm nccn guidelines version 3.2021 [internet]; 2021 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/colon_blocks.pdf
- 43. Benson AB, Venook AP, Al-Hawary MM, Azad N, Chen Y-J, Ciombor KK, et al. Nccn clinical practice guidelines in oncology rectal cancer nccn evidence blocks tm nccn guidelines version 2.2021 [internet]; 2021 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/rectal_blocks.pdf
- 44. Armstrong DK, Alvarez RD, Bakkum-Gamez JN, Barroilhet L, Behbakht A, Chen L, et al. Nccn clinical practice guidelines in oncology ovarian cancer including fallopian tube cancer and primary peritoneal cancer nccn evidence blocks tm nccn guidelines version 3.2021 [internet]; 2021 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/ovarian_blocks.pdf
- 45. Abu-Rustum N, Yashar C, Bradley K, Brooks R, Campos S, Chino J, et al. Nccn clinical practice guidelines in oncology uterine neoplasms nccn evidence blocks tm nccn guidelines version 4.2021 [internet]; 2021 [cited 2021 Sep 27]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/uterine_blocks.pdf

- 46. Abu-Rustum NR, Yashar CM, Bradley K, Campos SM, Sook Chon H, Chu C, et al. Nccn clinical practice guidelines in oncology cervical cancer nccn evidence blocks tm nccn guidelines version 1.2021 [internet]; 2020 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/cervical_bl ocks.pdf
- Abu-Rustum NR, Yashar CM, Bradley K, Campos SM, Sook Chon H, Chu C, et al. Nccn clinical practice guidelines in oncology vulvar cancer (squamous cell carcinoma) nccn evidence blocks tm nccn guidelines version 3.2021 [internet]; 2021 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/ph ysician_gls/pdf/vulvar_blocks.pdf
- 48. Schaeffer E, Srinivas S, Antonarakis E, Armstrong AJ, Cheng HH, Victor AD, et al. Nccn clinical practice guidelines in oncology prostate cancer nccn evidence blocks tm nccn guidelines version 1.2022 [internet]; 2021 [cited 2021 Sep 27]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/prostate_bl ocks.pdf
- 49. Sanft T, Tevaarwerk A, Denlinger C, Ansbaugh S, Armenian S, Scott Baker K, et al. Nccn clinical practice guidelines in oncology survivorship tm nccn guidelines version 3.2021 [internet]; 2021 [cited 2021 Sep 26]. Available from: https://www.nccn.org/professionals/physician_gls/pdf/survivorship.pdf
- 50. The American Society of Clinical Oncology. Guidelines, tools, & resources [internet]. The American Society of Clinical Oncology; 2021 [cited 2021 Sep 28]. Available from: https://www.asco.org/practice-patients/guidelines
- 51. Runowicz CD, Leach CR, Henry NL, Henry KS, Mackey HT, Cowens-Alvarado RL, et al. American cancer society/American society of clinical oncology breast cancer survivorship care guideline. CA Cancer J Clin. 2016;66(1):43–73. Available from: https://onlinelibrary.wiley.com/doi/full/10.3322/caac.21319
- 52. Khatcheressian JL, Hurley P, Bantug E, Esserman LJ, Grunfeld E, Halberg F, et al. Breast cancer follow-up and management after primary treatment: American society of clinical oncology clinical practice guideline update. J Clin Oncol. 2013;31(7):961–5.
- 53. Meyerhardt JA, Mangu PB, Flynn PJ, Korde L, Loprinzi CL, Minsky BD, et al. Follow-up care, surveillance protocol, and secondary prevention measures for survivors of colorectal cancer: American society of clinical oncology clinical practice guideline endorsement. J Clin Oncol. 2013;31(35):4465–70.
- 54. Resnick MJ, Lacchetti C, Bergman J, Hauke RJ, Hoffman KE, Kungel TM, et al. Prostate cancer survivorship care guideline: American society of clinical oncology clinical practice guideline endorsement. J Clin Oncol. 2015;33(9):1078–85.

- Alberta Health Services. Cancer guidelines [internet]. Alberta Health Services;
 2021 [cited 2021 Sep 28]. Available from: https://www.albertahealthservices.ca/in fo/cancerguidelines.aspx
- 56. Alberta Health Services. Alberta health services follow-up care for early-stage breast cancer clinical practice guideline br-013 version 2 [internet]. Alberta Health Services; 2015 [cited 2021 Sep 27]. Available from: https://www.albertahealthserv ices.ca/assets/info/hp/cancer/if-hp-cancer-guide-br013-early-stage-follow-up.pdf
- 57. Alberta Health Services. Alberta health services algorithm for the management of undifferentiated endometrial sarcoma and recurrent sarcoma of the uterus gynecologic oncology provincial treatment pathways (gyne-007) [internet]. Alberta Health Services; 2013 [cited 2021 Sep 27]. Available from http://www.alb ertahealthservices.ca/assets/info/hp/cancer/if-hp-cancer-algorithm-gyne007uterine-sarcoma.pdf
- 58. Alberta Health Services. Alberta health services local prostate cancer clinical practice guideline gu-012-version 3 [internet]. Alberta Health Services; 2020 [cited 2021 Sep 27]. Available from: https://www.albertahealthservices.ca/assets/info/hp /cancer/if-hp-cancer-guide-gu012-local-prostate.pdf
- 59. Albaerta Health Services. Alberta health services advanced/metastatic prostate cancer clinical practice guideline gu-010-version 2 [internet]. Alberta Health Services; 2020 [cited 2021 Sep 27]. Available from: https://www.albertahealthser vices.ca/assets/info/hp/cancer/if-hp-cancer-guide-gu010-met-prostate.pdf
- 60. Alberta Health Services. Alberta health services colorectal cancer surveillance (stages I, II, and III) clinical practice guideline gi-002 version 7 [internet]. Alberta Health Services; 2019 [cited 2021 Sep 27]. Available from: https://www.albertahe althservices.ca/assets/info/hp/cancer/if-hp-cancer-guide-gi002-colon-surveillance.pdf
- 61. Alberta Health Services. Alberta health services algorithm for the management of early stage cancer of the uterine cervix gynecologic oncology provincial treatment pathways (gyne-004) [internet]. Alberta Health Services; 2015 [cited 2021 Sep 27]. Available from: https://www.albertahealthservices.ca/assets/info/hp/cancer/if-hp-cancer-guide-gyne004-algorithm-cervical.pdf
- 62. Alberta Health Services. Alberta health services algorithm for the management of advanced stage and recurrent/persistent cancer of the uterine cervix gynecologic oncology provincial treatment pathways (gyne-004) [internet]. Alberta Health Services; 2013 [cited 2021 Sep 27]. Available from: https://www.albertahealthserv ices.ca/assets/info/hp/cancer/if-hp-cancer-guide-gyne004-algorithm-cervical.pdf

- 63. Alberta Health Services. Alberta health services algorithm for the management of endometrial carcinoma gynecologic oncology provincial treatment pathways (gyne-002) [internet]. Alberta Health Services; 2016 [cited 2021 Sep 27]. Available from: https://www.albertahealthservices.ca/assets/info/hp/cancer/if-hp-cancer-guide-gyne002-algorithm-endometrial.pdf
- 64. Alberta Health Services. Alberta health services algorithm for the management of endometrial serous, clear cell, and carcinosarcoma gynecologic oncology provincial treatment pathways (gyne-002) [internet]. Alberta Health Services; 2016 [cited 2021 Sep 27]. Available from https://www.albertahealthservices.ca/as sets/info/hp/ cancer/if-hp-cancer-guide-gyne002-endometrial.pdf
- 65. Alberta Health Services. Alberta health services algorithm for the management of early stage (stages I, IIA) epithelial ovarian, fallopian tube and primary peritoneal cancer gynecologic oncology provincial treatment pathways (gyne-005) [internet]. Alberta Health Services; 2013 [cited 2021 Sep 27]. Available from: https://www.al bertahealthservices.ca/assets/info/hp/cancer/if-hp-cancer-guide-gyne005-algorith m-epithelialovarian.pdf
- 66. Alberta Health Services. Alberta health services algorithm for the management of intermediate (stage IIB, IIC)/ advanced (stages III, IV) and recurrent epithelial ovarian, fallopian tube and primary peritoneal cancer gynecologic oncology provincial treatment pathways (gyne-005) [internet]. Alberta Health Services; 2013 [cited 2021 Sep 27]. Available from https://www.albertahealthservices.ca/ass ets/info/hp/cancer/if-hp-cancer-guide-gyne005-algorithm-epithelialovarian.pdf
- 67. Alberta Health Services. Alberta health services algorithm for the management of uterine leiomyosarcoma, adenosarcoma & endometrial stromal sarcoma gynecologic oncology provincial treatment pathways (gyne-007) [internet]. Alberta Health Services; 2013 [cited 2021 Sep 27]. Available from: https://www.a lbertahealthservices.ca/assets/info/hp/cancer/if-hp-cancer-algorithm-gyne007-uterine-sarcoma.pdf
- 68. Cancer Care Ontario. Displaying guidelines [internet]. Cancer Care Ontario; 2020. [cited 2021 Sep 28]. Available from: https://www.cancercareontario.ca/en/guideli nes-advice/cancer-continuum/survivorship
- Eisen A, Grunfeld E, Muradali D, Trudeau M. Cco's position on guidelines for breast cancer well follow-up care [internet]. Cancer Care Ontario; 2012 [cited 2021 Sep 28]. Available from: https://www.cancercareontario.ca/en/guidelines-ad vice/types-of-cancer/37786

- 70. Kennedy E, Zwaal C, Asmis T, Cho C, Galica J, Ginty A, et al. Cco follow-up care, surveillance protocol, and secondary prevention measures for survivors of colorectal cancer guideline 26-2 version 3 [internet]. Cancer Care Ontario; 2021 [cited 2021 Sep 26]. Available from: https://www.cancercareontario.ca/en/guideli nes-advice/types-of-cancer/256
- Elit L, Kennedy EB, Fyles A, Metser U. Cco follow-up for cervical cancer guideline 4-16 version 2 [internet]. Cancer Care Ontario; 2015 [cited 2021 Sep 27]. Available from: https://www.cancercareontario.ca/en/guidelines-advice/typesof-cancer/476
- 72. Fung-Kee-Fung M, Dodge J, Elit L, Lukka H, Chambers A, Oliver T, et al. Cco follow-up after primary therapy for endometrial cancer evidence-based series 4-9 version 2 [internet]. Cancer Care Ontario; 2017 [cited 2021 Sep 27]. Available from: https://www.cancercareontario.ca/en/guidelines-advice/types-of-cancer/616
- 73. Matthew A, Souter LH, Breau RH, Canil C, Haider M, Jamnicky L, et al. Cco follow-up care and psychosocial needs of survivors of prostate cancer guideline 26-4 [internet]. 2015 [cited 2021 Sep 27]. Available from: https://www.cancercare ontario.ca/en/guidelines-advice/types-of-cancer/266
- 74. Sussman J, Beglaryan H, Payne A. Cco follow-up model of care for cancer survivors- recommendations for the delivery of follow-up care for cancer survivors in ontario [internet]. Cancer Care Ontario; 2019 [cited 2021 Sep 26]. Available from: https://www.cancercareontario.ca/en/guidelines-advice/types-of-cancer/587 36
- 75. Sussman J, Souter LH, Grunfeld E, Howell D, Gage C, Keller-Olaman S, et al. Cco models of care for cancer survivorship evidence-based series 26-1 version 2 [internet]. Cancer Care Ontario; 2017 [cited 2021 Sep 27]. Available from: https:// www.cancercareontario.ca/en/guidelines-advice/types-of-cancer/246
- 76. Saskatchewan Cancer Agency. Follow up guidelines [internet]. Saskatchewan Cancer Agency; n.d. [cited 2021 Sep 28]. Available from: http://www.saskcancer. ca/health-professionals-article/follow-up-guidelines
- 77. Saskatchewan Cancer Agency. Breast cancer criteria for discharge [internet]. Saskatchewan Cancer Agency; 2020 [cited 2021 Sep 28]. Available from http://www.saskcancer.ca/images/pdfs/health_professionals/clinical_resources/foll ow-up-guidelines/Breast_Cancer_Criteria_for_Discharge_08-2020.pdf
- 78. Saskatchewan Cancer Agency. Gastro-intestinal cancer criteria for discharge [internet]. Saskatchewan Cancer Agency; 2020 [cited 2021 Sep 28]. Available from: http://www.saskcancer.ca/images/pdfs/health_professionals/clinical_resou rces/follow-up-guidelines/Gastro-Esophageal_Junction_Cancer_Criteria_for_Disc harge_09-2020.pdf

- 79. Saskatchewan Cancer Agency. Cervical cancer follow-up guidelines [internet]. Saskatchewan Cancer Agency; 2021 [cited 2021 Sep 28]. Available from http://www.saskcancer.ca/images/pdfs/health_professionals/clinical_resources/foll ow-up-guidelines/Cervial_Cancer_Follow_Up_-_March_2021.pdf.
- 80. Saskatchewan Cancer Agency. Endometrial cancer follow-up guidelines [internet]. Saskatchewan Cancer Agency; 2021 [cited 2021 Sep 28]. Available from: http://w ww.saskcancer.ca/images/pdfs/health_professionals/clinical_resources/follow-upguidelines/Endometrial_Cancer_Follow_Up_-_March_2021.pdf
- 81. Saskatchewan Cancer Agency. Ovarian cancer follow-up guidelines [internet]. Saskatchewan Cancer Agency; 2021 [cited 2021 Sep 28[. Available from: http://w ww.saskcancer.ca/images/pdfs/health_professionals/clinical_resources/follow-upguidelines/Ovarian_Cancer_Follow_Up_-_March_2021.pdf
- 82. Saskatchewan Cancer Agency. Vulvar cancer follow-up guidelines [internet]. Saskatchewan Cancer Agency; 2021 [cited 2021 Sep 28]. Available from: http://w ww.saskcancer.ca/images/pdfs/health_professionals/clinical_resources/follow-upguidelines/Vulvar_Cancer_Follow_Up_Guidelines_-_April_2021.pdf
- 83. Saskatchewan Cancer Agency. Genitourinary discharge care pathway [internet]. Saskatchewan Cancer Agency; 2020 [cited 28 Sep 2021]. Available from: http://w ww.saskcancer.ca/images/pdfs/health_professionals/clinical_resources/follow-upguidelines/GU_Discharge_Care_Pathway_12-2020.pdf
- 84. Miedema B, MacDonald I, Tatemichi S. Cancer follow-up care patients' perspectives. Can Fam Physician. 2003;49(7).
- 85. Grunfeld E, Levine MN, Julian J, Folkes A, Pond GR, Maunsell E. Breast cancer survivors perception of family physician (fp) or specialist as principal provider of routine follow-up care. J Clin Oncol. 2010;28(15_suppl):9090–9090.
- Cheung WY, Neville BA, Cameron DB, Cook EF, Earle CC. Comparisons of patient and physician expectations for cancer survivorship care. J Clin Oncol. 2009;27(15):2489–95.
- 87. Chapman K, Wiernikowski J. Adult cancer survivorship a self-learning resource for nurses [internet]. Canadian Association of Nurses in Oncology; 2011 [cited 2020 Oct 20]. Available from: www.cano-acio.ca
- de Oliveira C, Weir S, Rangrej J, Krahn MD, Mittmann N, Hoch JS, et al. The economic burden of cancer care in Canada: a population-based cost study. C Open. 2018;6(1):E1–10.

- 89. Fundytus A, Hopman WM, Hammad N, Biagi JJ, Sullivan R, Vanderpuye V, et al. Medical oncology workload in Canada: infrastructure, supports, and delivery of clinical care. Curr Oncol. 2018;25(3):206–12.
- 90. Panagioti M, Geraghty K, Johnson J, Zhou A, Panagopoulou E, Chew-Graham C, et al. Association between physician burnout and patient safety, professionalism, and patient satisfaction. JAMA Intern Med. 2018;178(10):1317.
- 91. Sussman J, Evans DB and WK. Towards integrating primary care with cancer care: a regional study of current gaps and opportunities in Canada. Healthcare Policy. 2017;12(3):50-65.
- 92. Choosing Wisely Canada. About [internet]. Choosing Wisely Canada; 2020. [cited 2020 Jun 11]. Available from: https://choosingwiselycanada.org/about/
- 93. Grunfeld E, Julian JA, Pond G, Maunsell E, Coyle D, Folkes A, et al. Evaluating survivorship care plans: results of a randomized, clinical trial of patients with breast cancer. J Clin Oncol. 2011;29(36):4755–62.
- 94. Tan J, Muir J, Coburn N, Singh S, Hodgson D, Saskin R, et al. Surveillance patterns after curative-intent colorectal cancer surgery in Ontario. Can J Gastroenterol Hepatol. 2014;28(8):427–33.
- 95. Grunfeld E, Hodgson DC, Del Giudice ME, Moineddin R. Population-based longitudinal study of follow-up care for breast cancer survivors. J Oncol Pract. 2010;6(4):174–81.
- 96. Hodgson DC, Grunfeld E, Gunraj N, Del Giudice L. A population-based study of follow-up care for hodgkin lymphoma survivors: opportunities to improve surveillance for relapse and late effects. Cancer. 2010;116(14):3417–25.
- 97. Sisler JJ, Seo B, Katz A, Shu E, Chateau D, Czaykowski P, et al. Concordance with asco guidelines for surveillance after colorectal cancer treatment: a population-based analysis. J Oncol Pract. 2012;8(4):e69–79.
- 98. Hodgson DC, Grunfeld E, Gunraj N, Giudice L Del. A population-based study of follow-up care for hodgkin lymphoma survivors. Cancer. 2010;116(14):3417–25.
- 99. Kwon JS, Elit L, Saskin R, Hodgson D, Grunfeld E. secondary cancer prevention during follow-up for endometrial cancer: Obstet Gynecol. 2009;113(4):790–5.
- 100. Jiang L, Lofters A, Moineddin R, Decker K, Groome P, Kendell C, et al. Primary care physician use across the breast cancer care continuum recherche le rôle des médecins de première ligne tout au long du traitement du cancer du sein. Can Fam physician / Med Fam Can. 2016;62:589–98.

- Corkum M, Urquhart R, Kephart G, Hayden JA, Porter G. Breast and cervical cancer screening behaviours among colorectal cancer survivors in Nova Scotia. Curr Oncol. 2014;21(5):e670–7.
- 102. Cheung WY, Aziz N, Noone A-M, Rowland JH, Potosky AL, Ayanian JZ, et al. Physician preferences and attitudes regarding different models of cancer survivorship care: a comparison of primary care providers and oncologists. J Cancer Surviv. 2013;7(3):343–54.
- 103. Potosky AL, Han PKJ, Rowland J, Klabunde CN, Smith T, Aziz N, et al. Differences between primary care physicians' and oncologists' knowledge, attitudes and practices regarding the care of cancer survivors. J Gen Intern Med. 2011;26(12):1403–10.
- 104. Kantsiper M, McDonald EL, Geller G, Shockney L, Snyder C, Wolff AC. Transitioning to breast cancer survivorship: perspectives of patients, cancer specialists, and primary care providers. J Gen Intern Med. 2009;24(2):459.
- 105. Audisio RA, Robertson C. Colorectal cancer follow-up: perspectives for future studies. Eur J Surg Oncol. 2000;26(4):329–37.
- 106. Huibertse LJ, van Eenbergen M, de Rooij BH, Bastiaens MT, Fossion LL, de la Fuente RB, et al. Cancer survivors' preference for follow-up care providers: a cross-sectional study from the population-based profiles-registry. Acta Oncol (Madr). 2017;56(2):278–87.
- Grava-Gubins I, Safarov A, Eriksson J. 2010 national physician survey : workload patterns of Canadian family physicians. [PowerPoint presentation]. National Physician Survey. 2012 [cited 15 Nov 2020].
- 108. Statistics Canada. Seniors [internet]. Statistics Canada; 2018 [cited 2020 Jul 7]. Available from: https://www150.statcan.gc.ca/n1/pub/11-402-x/2011000/chap/ sen iors-aines/seniors-aines-eng.htm
- 109. Buckinx F, Rolland Y, Reginster JY, Ricour C, Petermans J, Bruyère O. Burden of frailty in the elderly population: perspectives for a public health challenge. Archives of Public Health. 2015;73(1):1-7.
- 110. Mulasso A, Roppolo M, Giannotta F, Rabaglietti E. Associations of frailty and psychosocial factors with autonomy in daily activities: a cross-sectional study in italian community-dwelling older adults. Clin Interv Aging. 2016;11:37–45.
- 111. Fried LP, Ferrucci L, Darer J, Williamson JD, Anderson G. Untangling the concepts of disability, frailty, and comorbidity: implications for improved targeting and care. Journals Gerontol Ser A. 2004;59(3):M255–63.

- 112. Searle SD, Mitnitski A, Gahbauer EA, Gill TM, Rockwood K. A standard procedure for creating a frailty index. BMC Geriatr. 2008;8(1):1–10.
- 113. Theou O, Cann L, Blodgett J, Wallace LMK, Brothers TD, Rockwood K. Modifications to the frailty phenotype criteria: systematic review of the current literature and investigation of 262 frailty phenotypes in the survey of health, ageing, and retirement in europe. Ageing Research Reviews. 2015;21:78-94.
- 114. Fried LP, Tangen CM, Walston J, Newman AB, Hirsch C, Gottdiener J, et al. Frailty in older adults: evidence for a phenotype. Journals Gerontol - Ser A Biol Sci Med Sci. 2001;56(3):146–57.
- Theou O, Walston J, Rockwood K. Operationalizing frailty using the frailty phenotype and deficit accumulation approaches. Interdiscip Top Gerontol Geriatr. 2015;41:66–73.
- 116. Davidoff AJ, Zuckerman IH, Pandya N, Hendrick F, Ke X, Hurria A, et al. A novel approach to improve health status measurement in observational claimsbased studies of cancer treatment and outcomes. J Geriatr Oncol. 2013;4(2):157– 65.
- 117. Rosen A, Wu J, Chang BH, Berlowitz D, Rakovski C, Ash A, et al. Risk adjustment for measuring health outcomes: an application in va long-term care. Am J Med Qual. 2001;16(4):118–27.
- 118. Fassbender K, Fainsinger RL, Carson M, Finegan BA. Cost trajectories at the end of life: the Canadian experience. J Pain Symptom Manage. 2009;38(1):75–80.
- 119. Earle CC, Park ER, Lai B, Weeks JC, Ayanian JZ, Block S. Identifying potential indicators of the quality of end-of-life cancer care from administrative data. J Clin Oncol. 2003;21(6):1133–8.
- 120. Iezzoni LI. Assessing quality using administrative data. In: Annals of Internal Medicine. American College of Physicians; 1997;127(8II SUPPL):666–73.
- Rockwood K, Song X, Macknight C, Bergman H. A global clinical measure of fitness and frailty in elderly people. Canadian Medical Association Journal. 2005;30(5):489.
- 122. Rolfson DB, Majumdar SR, Tsuyuki RT, Tahir A, Rockwood K. Validity and reliability of the Edmonton frail scale. Age and Ageing . 2006;35(5):526-29.
- 123. Langton JM, Wong ST, Burge F, Choi A, Ghaseminejad-Tafreshi N, Johnston S, et al. Population segments as a tool for health care performance reporting: an exploratory study in the Canadian province of British Columbia. BMC Fam Pract. 2020;21(1).

- 124. Gershon AS, Chung H, Porter J, Campitelli MA, Buchan SA, Schwartz KL, et al. Influenza vaccine effectiveness in preventing hospitalizations in older patients with chronic obstructive pulmonary disease. J Infect Dis. 2020;22(1):42-52.
- 125. Wong ST, Katz A, Williamson T, Singer A, Peterson S, Taylor C, et al. Can linked electronic medical record and administrative data help us identify those living with frailty? Int J Popul Data Sci. 2020;5(1).
- 126. Collard RM, Boter H, Schoevers RA, Oude Voshaar RC. Prevalence of frailty in community-dwelling older persons: a systematic review. Journal of the American Geriatrics Society. 2012;60(8):1487–92.
- 127. Giguere AC. Environmental scan to describe the current care received by frail seniors in Canada [preliminary report]. Technology in the Elderly Network. 2016.
- 128. Drubbel I, de Wit NJ, Bleijenberg N, Eijkemans RJC, Schuurmans MJ, Numans ME. Prediction of adverse health outcomes in older people using a frailty index based on routine primary care data. Journals Gerontol Ser A. 2013;68(3):301–8.
- 129. Featherstone A. Developing a holistic, multidisciplinary community service for frail older people. Nurs Older People. 2018;30(7):34–40.
- Handforth C, Clegg A, Young C, Simpkins S, Seymour MT, Selby PJ, et al. The prevalence and outcomes of frailty in older cancer patients: a systematic review. Ann Oncol. 2015;26(6):1091–101.
- 131. Ness KK, Wogksch MD. Frailty and aging in cancer survivors. Transl Res. 2020;221:65.
- 132. Brown JC, Harhay MO, Harhay MN. The prognostic importance of frailty in cancer survivors. J Am Geriatr Soc. 2015;63(12):2538–43.
- 133. Pérez-zepeda MU, Cárdenas-cárdenas E, Cesari M, Navarrete-reyes AP, Gutiérrezrobledo LM. Cancer and frailty in older adults: a nested case-control study of the Mexican health and aging study. J Cancer Surviv New York. 2016;10(4):736–42.
- 134. Lesser known facts about nested case-control designs [Internet]. JSCI Med Central; n.d. [cited 2020 Sep 6]. Available from: http://www.jscimedcentral.com/ TranslationalMedicine/Articles/translationalmedicine-1-1007.php
- 135. Smitherman AB, Anderson C, Lund JL, Bensen JT, Rosenstein DL, Nichols HB. Frailty and comorbidities among survivors of adolescent and young adult cancer: a cross-sectional examination of a hospital-based survivorship cohort. J Adolesc. 2018;7(3):374–83.

- 136. Shulman LN, Jacobs LA, Greenfield S, Jones B, McCabe MS, Syrjala K, et al. Cancer care and cancer survivorship care in the United States: will we be able to care for these patients in the future? J Oncol Pract. 2016;5(3):119–23.
- Halpern MT, McCabe MS, Burg MA. The cancer survivorship journey: models of care, disparities, barriers, and future directions. Am Soc Clin Oncol Educ B. 2018;36:231-9.
- 138. Sussman J, Souter LH, Grunfeld E, Howell D, Gage C, Keller- Olaman S et al. Models of care for cancer survivorship - full report. 2012;26–1(2):1–58.
- 139. Mittmann N, Beglaryan H, Liu N, Seung SJ, Rahman F, Gilbert J, et al. Examination of health system resources and costs associated with transitioning cancer survivors to primary care: a propensity-score-matched cohort study. J Oncol Pract. 2018;14(11):e653–64.
- 140. Halpern MT, Argenbright KE. Evaluation of effectiveness of survivorship programmes: how to measure success? Lancet Oncol. 2017;18(1):e51–9.
- Taplin SH, Rodgers AB. Toward improving the quality of cancer care: addressing the interfaces of primary and oncology-related subspecialty care. JNCI Monogr. 2010;2010(40):3–10.
- 142. Mayer DK, Alfano CM. Personalized risk-stratified cancer follow-up care: its potential for healthier survivors, happier clinicians, and lower costs. Journal of the National Cancer Institute. 2019;111:442–8.
- 143. Elixhauser A, Steiner C, Harris D CR. Comorbidity measures for use with administrative data. Med Care. 1998;36(1):8–27.
- 144. The National Cancer Institute. Definition of neoadjuvant therapy nci dictionary of cancer terms [internet]. The National Cancer Institute; n.d. [cited 2020 Sep 6]. Available from: https://www.cancer.gov/publications/dictionaries/cancer-terms/def /neoadjuvant-therapy
- 145. Statistics Canada. Dictionary, census of population, 2016 census metropolitan influenced zone (miz) [internet]. Statistics Canada; 2016 [cited 2022 Dec 2]. Available from: https://www12.statcan.gc.ca/census-recensement/2016 /ref/dict/g eo010-eng.cfm
- 146. Williams R. Generalized ordered logit/partial proportional odds models for ordinal dependent variables. Sage. 2006;6(1):58–82.

- 147. Lueckmann SL, Hoebel J, Roick J, Markert J, Spallek J, von dem Knesebeck O, et al. Socioeconomic inequalities in primary-care and specialist physician visits: a systematic review. Int J Equity Health. 2021;20(1):1–19.
- 148. Canadian Institutes of Health Research, Natural Sciences and Engineering Research Council of Canada. Tri-council policy statement: ethical conduct for research involving humans – tcps2 2018 – chapter 5: privacy and confidentiality. Ottawa, ON: Government of Canada; 2018.
- 149. Ness KK, Wogksch MD. Frailty and aging in cancer survivors. Transl Res. 2020;221:65–82.
- 150. Koll TT, Semin JN, Brodsky R, Keehn D, Fisher AL, High R, et al. Health-related and sociodemographic factors associated with physical frailty among older cancer survivors. J Geriatr Oncol. 2021;12(1):96–101.
- Bennett JA, Winters-Stone KM, Dobek J, Nail LM. Frailty in older breast cancer survivors: age, prevalence, and associated factors. Oncol Nurs Forum. 2013;40(3):E126.
- 152. Hayek S, Gibson TM, Leisenring WM, Guida JL, Gramatges MM, Lupo PJ, et al. Prevalence and predictors of frailty in childhood cancer survivors and siblings: a report from the childhood cancer survivor study. J Clin Oncol. 2020;38(3):232–47.
- 153. Wang J, Hulme C. Frailty and socioeconomic status: a systematic review. J Public health Res. 2021;10(3):2036.
- Mandelblatt JS, Clapp JD, Luta G, Faul LA, Tallarico MD, McClendon TD, et al. Long-term trajectories of self-reported cognitive function in a cohort of older survivors of breast cancer: calgb 369901 (alliance). Cancer. 2016;122(22):3555– 63.
- 155. Winters-Stone KM, Moe E, Graff JN, Dieckmann NF, Stoyles S, Borsch C, et al. Falls and frailty in prostate cancer survivors: current, past, and never users of androgen deprivation therapy. J Am Geriatr Soc. 2017;65(7):1414–9.
- 156. Bylow K, Hemmerich J, Mohile SG, Stadler WM, Sajid S, Dale W. Obese frailty, physical performance deficits, and falls in older men with biochemical recurrence of prostate cancer on androgen deprivation therapy: a case-control study. Urology. 2011;77(4):934–40.
- 157. Urquhart R, Rayson D, Porter GA, Grunfeld E. Quantifying limitations in chemotherapy data in administrative health databases: implications for measuring the quality of colorectal cancer care. Healthc Policy. 2011;7(1):32.

- 158. Ness KK, Krull KR, Jones KE, Mulrooney DA, Armstrong GT, Green DM, et al. Physiologic frailty as a sign of accelerated aging among adult survivors of childhood cancer: a report from the st jude lifetime cohort study. J Clin Oncol. 2013;31(36):4496–503.
- 159. Pérez-Campdepadrós M, Castellano-Tejedor C, Sábado-Álvarez C, Gros-Subías L, Capdevila L, Blasco-Blasco T. Type of tumour, gender and time since diagnosis affect differently health-related quality of life in adolescent survivors. Eur J Cancer Care (Engl). 2015;24(5):635–41.
- 160. Mandelblatt JS, Zhou X, Small BJ, Ahn J, Zhai W, Ahles T, et al. Deficit accumulation frailty trajectories of older breast cancer survivors and non-cancer controls: the thinking and living with cancer study. JNCI J Natl Cancer Inst. 2021;113(8):1053–64.
- Taylor BL, Xia L, Guzzo TJ, Scherr DS, Hu JC. Frailty and greater health care resource utilization following major urologic oncology surgery. Eur Urol Oncol. 2019;2(1):21–7.
- 162. Ambagtsheer RC, Moussa RK. Association of frailty with health service utilisation and health care expenditure in Sub-Saharan Africa: evidence from Côte d'Ivoire. BMC Geriatr. 2021;21(1).
- 163. Urquhart R, Lethbridge L. Primary care use after cancer treatment: an analysis of linked administrative data. Curr Oncol. 2020;27(6):590-595.
- 164. Cunningham CT, Jette N, Li B, Dhanoa RR, Hemmelgarn B, Noseworthy T, et al. Effect of physician specialist alternative payment plans on administrative health data in Calgary: a validation study. C Open. 2015;3(4):E406–12.
- 165. Lix LM, Kuwornu JP, Kroeker K, Kephart G, Sikdar KC, Smith M, et al. Estimating the completeness of physician billing claims for diabetes case ascertainment using population-based prescription drug data. Heal Promot Chronic Dis Prev Canada. 2016;36(3):54–60.
- 166. Bartlett J. When is complete case analysis unbiased? [internet]. The Stats Geeks; 2013 [cited 2022 Sep 29]. Available from: https://thestatsgeek.com/2013/07/06/w hen-is-complete-case-analysis-unbiased/
- 167. Ellis KR, Black KZ, Baker S, Cothern C, Davis K, Doost K, et al. Racial differences in the influence of healthcare system factors on informal support for cancer care among black and white breast and lung cancer survivors. Fam Community Health. 2020;43(3):200.

- 168. Frank C, Wilson CR. Models of primary care for frail patients. Can Fam Physician. 2015;61(7):601-6.
- 169. Morgan B, Tarbi E. The role of the advanced practice nurse in geriatric oncology care. Semin Oncol Nurs. 2016;32(1):33–43.
- 170. The American Society of Clinical Oncology. Patient and survivor care [internet]. The American Society of Clinical Oncology; n.d. [cited 2020 Aug 29]. Available from: https://www.asco.org/research-guidelines/quality-guidelines/guidelines/pati ent-and-survivor-care
- 171. National Comprehensive Cancer Network. Nccn survivorship guidelines [internet]. National Comprehensive Cancer Network; 2011 [cited 2020 Aug 29]. Available from: https://www.nccn.org/professionals/physician_gls/default.aspx
- 172. Alberta Health Services. Cancer guidelines [internet]. Alberta Health Services; n.d. [cited 2020 Aug 29]. Available from: https://www.albertahealthservices.ca/info/ca ncerguidelines.aspx
- 173. Saskatchewan Cancer Agency. Follow up guidelines [internet]. Saskatchewan Cancer Agency; n.d. [cited 2020 Aug 29]. Available from: http://www.saskcancer. ca/health-professionals-article/follow-up-guidelines

Appendix A: Follow-up Guideline Chart

The table below describes recommended models of care/provider type by cancer site according to select organizations with cancer survivor follow-up care guidelines (68,170–173).

Organization	Cancer site recommendation ^a			
	Breast	Colorectal	Gynecologic	Prostate
Alberta	May transition	Follow-up care	Survivors of	No provider
Health	early-stage	may be provided	early stage and	type
Services	survivors to	by the general	advanced stage	recommendatio
	family	practitioner,	recurrent/persist	n found for
	physician or	nurse	ent cancer of	follow-up care
	nurse	practitioner, or a	the uterine	(58,59).
	practitioner	medical/	cervix should	
	(56).	radiation	initially be	
		oncologist (60).	followed by a	
			physician	
			experienced in	
			surveillance of	
			cancer patients.	
			After the first	
			two years, the	
			patient can be	
			discharged to	
			the primary	
			care physician	
			(61,62).	
			For survivors of	
			endometrioid	
			carcinoma and	
			of endometrial	
			serous, clear	
			cell, and	
			carcinosarcoma	
			, follow-up by	
			the treating	
			gynecologic	
			oncologist,	
			general	
			oncologist, or	
			general	

Organization	Cancer site recommendation ^a			
	Breast	Colorectal	Gynecologic	Prostate
			Cynecologicpractitionercould be basedon risk ofrecurrence(63,64).The algorithmformanagement ofepithelialovarian,fallopian tube,and primaryperitonealcancer does notspecify whichcare providershould beresponsible forfollow-up(65,66).For follow-upof uterinesarcomasurvivors, noprovider typerecommendation was found forfollow-up care	
American Society of Clinical Oncology	Early stage (i.e., tumor less than five centimeters and less than four positive nodes) may be followed solely by a primary care physician one- year post-	Recommendatio ns for colorectal cancer survivors, stages II and III (not stage I or resected metastatic disease, both of which have minimal data to provide guidance)	(57,67). No guidelines found.	May transition survivors to a primary care clinician for follow-up (54).

Organization	Cancer site recommendation ^a			
U	Breast	Colorectal	Gynecologic	Prostate
	diagnosis	include		
	(52).	transitioning		
		survivors who		
	They note a	have completed		
	lack of	all treatment to		
	evidence, but	community-		
	state it seems	based family		
	likely that	physician		
	history,	coordinated care		
	physical and	or institution-		
	breast exams	based nurse		
	may be	coordinated care		
	conducted by	(53).		
	experienced			
	non-physician			
	providers			
	(e.g., nurse			
	practitioners,			
	(51)			
Cancer Care	Most	May be	For cervical	Follow-up
Ontario	survivors can	discharged to	cancer	surveillance
Onturio	be safely	community-	survivors.	may be
	transitioned to	based family	follow-up by a	provided by the
	primary care	physician-led.	physician	treating
	physicians	Transition to	experienced in	oncologist,
	(69).	nurse-led care	surveillance of	urologist,
	Community-	within an	patients with	family
	based family	institution may	cancer. May be	physician,
	physician-led,	be a reasonable	followed by a	nurse
	or nurse-led	option (70).	primary care	practitioner, or
	care within an	They say	physician after	hospital-based
	institution are	although	five years of	nurses (73).
	reasonable	specialist	recurrence-free	
	options. States	follow-up is	follow-up (71).	Patients may
	hormonal	currently the		still be on
	therapy may	most common	For survivors of	hormone
	still be	practice, their	endometrial	therapy.
	ongoing (75).	2019 follow-up	cancer, 11 a	Transition to
		model of care	patient is	nursing-led
		Ior cancer	initially	care within an
		survivor	Iollowed by a	institution is a
	1	guidelines	specialist, they	reasonable

Organization	Cancer site recommendation ^a			
-	Breast	Colorectal	Gynecologic	Prostate
	Breast	Colorectal recommend primary care follow-up with specific handover pieces and supports in place from the specialist team, for survivors of all cancer types. This organization planned to implement the recommendation s across ON in breast and colorectal cancer site groups and expanded to other cancer sites over time	Gynecologic may be followed by a qualified general practitioner after three to five years of recurrence-free follow-up (72). No recommendatio ns can be made about models of care of other gynecologic disease types based on the currently available published literature (75).	Prostate option. There is not enough data to say discharge to primary care is equivalent to specialist care, but, based on the disease trajectory, the expert opinion is that this is a reasonable option (75).
National Comprehensi ve Cancer Network	Shared coordination between the primary care provider, oncology and subspeciality care providers is encouraged. Transition of care to primary care provider may be done when clinically appropriate (41,49).	(74). Care may be provided by the oncologist and the primary care provider. Roles should be clearly defined. A prescription for survivorship and transfer to the primary care physician should be written (42,43).	Unclear recommendatio ns. Varies by specific gynecologic cancer type. But does emphasize communication and coordination with all clinicians involved in the care of survivors, including primary care clinicians. However, does	No clear recommendatio n. Does mention primary care as an option for patients' follow-up care (48).

Organization	Cancer site recommendation ^a			
	Breast	Colorectal	Gynecologic	Prostate
			not specifically	
			mention	
			transfer of	
			follow-up care	
			to primary	
			care/other (44–	
			47).	
Nova Scotia	Indicates	Follow-up	No guidelines	No guidelines
Health Cancer	patients on	surveillance	available.	available.
Care Program	routine	guidelines exist		
	surveillance,	for both colon		
	discharged	and rectal		
	from the	cancers (39).		
	cancer centre			
	are followed-	For both rectal		
	up by the	and colon		
	primary care	cancers, patients		
	provider	on routine		
	(where	surveillance,		
	appropriate in	discharged from		
	partnership	the cancer centre		
	with the	are followed-up		
	surgeon).	by the primary		
	Patients also	care provider		
	receive a copy	(where		
	of the follow-	appropriate in		
	up	partnership with		
	recommendati	the surgeon).		
	ons, so they	Patients also		
	are aware of	receive a copy		
	the	of the follow-up		
	appropriate	recommendation		
	schedule.	s, so they are		
	Primary care	aware of the		
	providers/surg	appropriate		
	eons can	Schedule.		
	patient's	providers/surges		
	treating	providers/surgeo		
	oncologist at	the nationt's		
	any time for	treating		
	re-referral due	oncologist at		
	to significant	any time for re-		
	treatment_	referral due to		
	oncologist at any time for re-referral due to significant treatment-	the patient's treating oncologist at any time for re- referral due to		

Organization	Cancer site recommendation ^a			
C	Breast	Colorectal	Gynecologic	Prostate
	related side	significant		
	effects or	treatment-		
	suspected	related side		
	recurrence	effects or		
	(39).	suspected		
		recurrence (39).		
		For colon		
		cancers,		
		colonoscopy		
		follow-up is to		
		be coordinated		
		with the		
		attending		
		surgeon. For		
		rectal cancers		
		recto		
		sigmoidoscopy		
		and colonoscopy		
		are to be		
		coordinated with		
		the attending		
		surgeon (39).		
Saskatchewan	May transition	May transition	Cervical cancer	A risk-stratified
Cancer	early-stage	stage I, II or low	survivors are to	approach for
Agency	survivors to	risk stage II	be followed by	follow-up is
	the .	survivors to the	a gynecologic	described
	community	family doctor.	oncologist for	below.
	physician.	High risk stage	five years and	. .
	Although,	III and all stage	then discharge	For survivors
	those on	IV, unless	to the family	with localized
	adjuvant	otherwise	physician (79).	prostate cancer
	zoledronic	determined and	En demestriel	and very low or
	acid or a	fallowed at the	Endometrial	IOW TISK
	hormona	ionowed at the	cancer	If on active
	normone	Dadiation	Survivors (stage	II on active
	hormona	radiation	IA allu IB;	survemance-
	agonist	follows these	adenocarainam	urologist If
	agoinst	who have had	allow right) are	noiogist. II
	follow up at	dofinitivo	to be fellowed	post
	the concer	chemoradiathara	by a family	ofter consult
	(110 cancel)	ny (e.g. anal	by a failing	with radiation
		py (c.g., allal	gynecologist	oncologist can
		cancer) or long	gynecologist	oncologist can

Organization	Cancer site recommendation ^a			
U	Breast Colorectal Gynecologic Pro		Prostate	
Organization	Breast	Cancer site red Colorectal course chemoradiothera py (e.g., rectal cancer) wherein radiation is the primary modality. Medical oncology follows those who only had	commendation*Gynecologicfor five years,then dischargedfromendometrialcancer follow-up (80).Endometrialcancersurvivors,patients with	Prostate be discharged to family physician or urologist. If post external beam radiation therapy/ brachytherapy, radiation oncology to follow until
		who only had chemotherapy or had chemotherapy following concurrent chemoradiothera py (78).	patients with stage IA confined to the endometrium/n o myometrial invasion, to be discharged from post-op visit with no need for regular follow-up (no designated provider assigned follow-up) (80). All other endometrial cancer survivors	follow until resolution of toxicities and then discharge to family physician/urolo gist (83). For survivors in the favourable intermediate risk group: If on active surveillance- care by urologist. If post prostatectomy, after consult to radiation
			(advanced stages, high- grade histology/ high risk) should be followed (where possible) by a gynecologic oncologist for 5 years of follow- up, then discharged if no relapse (80).	oncology can be discharged to family physician and or urologist. If post external beam radiation therapy /brachytherapy /androgen deprivation therapy, radiation oncology to

Organization	Cancer site recommendation ^a			
-	Breast	Colorectal	Gynecologic	Prostate
				follow until
			Ovarian cancer	resolution of
			survivors to be	toxicities and
			followed by a	duration of
			gynecologic	androgen
			oncologist for a	deprivation
			total of 10 years	therapy. Then,
			from diagnosis	may be
			(82).	discharged to
				family
			Vulvar cancer	physician and
			survivors (stage	or urologist
			I and II) no	(83).
			comment on a	
			specific	For survivors
			provider type	with
			for follow-up	intermediate
			(82).	unfavourable or
				high-risk
			Vulvar cancer	disease:
			survivors	Radiation
			presenting with	oncology to
			more advanced	follow post
			disease will be	radiation
			followed at the	therapy until
			cancer center	resolution of
			and not	toxicities and
			discharged	completion of
			before five	androgen
			years of relapse	deprivation
			free follow-up	therapy (83).
			(82).	

^a Recommendations are based on patients who have completed active treatment, have no evidence of metastatic disease, or treatment complications not already addressed by an oncologist, unless exceptions are listed. Studies with nurse-led follow-up were performed in hospital institutions, not community-based settings. As most guidelines did not define what they meant by provider type (e.g., what constitutes a primary care provider), the original verbiage used for such terms was left as stated in the guideline. Generally, those enrolled in clinical trials are expected to be followed in the cancer centre. Most guidelines state clear care direction from the discharging physician (i.e., oncology specialist) to the receiving physician (e.g., primary care provider) is required for care transition. There is a distinction between nurse coordinated care and the most responsible provider. Nurse coordinated care does not mean the nurse is the most responsible provider.

Appendix B: MSI Physician Billings/CIHI DAD Censoring Procedure Codes

The below MSI Physician Billings and CIHI DAD procedure codes represent some of the

data which was used to censor individuals in the study cohort.

Cancer site	Procedures indicating cancer recurrence				
	MSI Physician Billings	CIHI DAD			
Breast	MASG 97.12 (Unilateral) Complete	1.YM.89.^^ Excision			
	Mastectomy	total, breast			
	MASG 97.13 Bilateral Complete	1.YM.90.^^ Excision			
	Mastectomy	total with			
	MASG 97.14 (Unilateral) Extended Simple	reconstruction, breast			
	Mastectomy	1.YM.91. ^{^^} Excision			
	MASG 97.15 Bilateral Extended Simple	radical, breast			
	Mastectomy	1.YM.92.^^ Excision			
	MASG 97.27A Quadrant resection,	radical with			
	lumpectomy, radical mastectomy with	reconstruction, breast			
	axillary dissection (regions required)				
Colorectal	MASG 44.3 Segmental resection of lung	1.GR.87.^^ Excision			
	(basilar) (superior)	partial, lobe of lung			
	MASG 44.4 Lobectomy of lung (regions	1.GR.89.^^ Excision			
	required)	total, lobe of lung			
	MASG 44.5 Complete pneumonectomy	1.GR.91. ^{^^} Excision			
	MASG 62.12 Partial hepatectomy-local	radical, lobe of lung			
	excision of lesion	1.OA.87.^^ Excision			
	MASG 62.2 Lobectomy of liver	partial, liver			
Gynecologic	No relevant surgical procedures	No relevant surgical			
		procedures			
Prostate	MASG 72.4A Prostatectomy with	1.QT.87.^^ Excision			
	vesiculectomy includes deep	partial, prostate			
	lymphadenectomy	1.QT.91. [^] Excision			
	MASG 72.4B Prostatectomy-radical	radical, prostate			
	including deep pelvic lymphadenectomy				
All	VADT 13.55 Injection or infusion of	Not applicable			
	cancer chemotherapeutic substance NEC				

CIHI DAD= Canadian Institutes for Health Information Discharge Abstracts Database, MSI= Medical Services Insurance Physicians' Billings.

Appendix C: MSI Physician Billings Diagnostic Codes Cancer-related Follow-up

Visits

The diagnostic codes listed below were used to identify follow-up care visits that were

related to the individual's cancer diagnosis. For each cancer site, diagnosis codes

considered relevant to the individual's cancer diagnosis were identified (some codes are

applicable to more than one or even all sites).

Category	MSI Physician Billings diagnostic code(s)				
Breast cance	r				
Malignant	174 Malignant neoplasm of female breast (includes 174.0, 174.1, 174.2,				
neoplasms	174.3, 174.4 174.5, 174.6, 174.8, 174.9)				
	175 Malignant neoplasm of male breast (includes 175.0 and 175.9)				
Related and	195.1 Malignant neoplasm of thorax (includes: axilla, chest wall not				
benign	otherwise specified)				
neoplasms	217 Benign neoplasm of breast				
	238.3 Neoplasm of uncertain behaviour of breast				
	239.3 Neoplasm of unspecified nature of breast				
Carcinoma	233.0 Carcinoma in situ of breast				
in situ					
Other	173.5 Unspecified malignant neoplasm of skin of trunk, except scrotum				
related	(includes: axillary fold and skin of breast) (1735.0, 173.51, 173.52,				
	173.59)				
	216.5 Benign neoplasm of skin of trunk, excluding scrotum (includes:				
	axillary fold and skin of breast)				
	232.5 Carcinoma in situ of skin of trunk (includes: axillary fold and skin				
	of breast)				
	451.89 Phlebitis and thrombophlebitis of other sites (includes:				
	thrombophlebitis of breast [Mondor's disease])				
	610 Benign mammary dysplasia (includes 610.0, 610.1, 610.2, 610.3,				
	610.4, 610.8, 610.9)				
	611 Other disorders of breast (includes 611.0, 611.1, 611.2, 611.3,				
	611.4, 611.5, 611.6, 611.7[611.71, 611.72, 611.79], 611.8[611.81,				
	611.82, 611.83, 611.89])				
	612 Deformity and disproportion of reconstructed breast (includes				
	612.0, 612.1)				
	793.8 Abnormal radiographic or other findings of the breast (i.e.,				
	abnormal or inconclusive mammogram) (includes 793.80, 793.81,				
	793.82, 793.89)				
	879.0 Open wound of breast without mention of complication (in				
	chapter 'internal injuries of thorax, abdomen, and pelvis')				
	879.1 Open wound of breast, complicated (from same chapter as above)				

Category	MSI Physician Billings diagnostic code(s)
	V10.3 Personal history of malignant neoplasm of the breast
	V45.71Acquired absence of breast and nipple
	V50.41 Prophylactic removal of breast
	V51 Aftercare involving the use of plastic surgery (e.g., breast
	reconstruction) (includes V51.0, V51.8)
	V76.1 Special screening for malignant neoplasms of breast (includes
	V76.10, V76.11, V76.12, V76.19)
	V86 Estrogen receptor status (includes V86.0, V86.1)
	V87.43 Personal history of estrogen therapy
Colorectal ca	ncer
Malignant	153 Malignant neoplasm of colon (includes 153.0, 153.1, 153.2, 153.3,
neoplasms	153.4, 153.5, 153.6, 153.7, 153.8, 153.9)
	154 Malignant neonlasm of rectum rectosigmoid junction and anus
	(includes 154.0, 154.1, 154.2, 154.3, 154.8)
Related and	195.2 Malignant neoplasm of abdomen (includes: intra-abdominal not
benign	otherwise specified)
neonlasms	211.4 Benign neonlasm of rectum and anal canal
neoplasmo	211.9 Benign neoplasm of other and unspecified site (includes:
	alimentary tract digestive system gastrointestinal tract intestinal tract
	intestine spleen not otherwise specified)
	235.2 Neonlasm of uncertain behavior of stomach intestines and
	rectum
	230 0 Neonlasm of unspecified nature of digestive system
Carainama	239.0 Neoplasm of unspectified nature of digestive system
in situ	250.5 Colon (includes: appendix, cecum, neocecal value, large intestine
III Situ	230 4 Rectum
	230.5 Anal canal
	230.6 Anus unspecified
	230.7 Other and unspecified parts of intesting
	230.7 Other and unspecified digestive argans (includes: digestive organ
	230.9 Other and unspective organs (includes, digestive organ
Othor	and gastronnestinal fact not otherwise specified, paneteas, spicell)
other valatad	$560 \pm 560 = 2560 \pm 20, 560 \pm 21, 560 \pm 20, 560 \pm 1, 560 \pm 0, 560$
related	500.1, 500.2, 500.50, 500.51, 500.52, 500.59, 500.81, 500.89, 500.9
	562 Diverticula of intestine (includes $562.00, 562.01, 562.02, 562.05, 562.10, 562.11, 562, 12, 562.12)$
	502.10, 502.11, 502, 12, 502.15)
	564 Functional digestive disorders, not elsewhere classified (includes $5(4,0), 5(4,0), 5(4,0), 5(4,0), 5(4,0)$
	504.0, 504.00, 504.01, 504.02, 504.09, 504.1, 504.2, 504.3, 504.4,
	564.5, 564.6, 564.7, 564.8, 564.81, 564.89, 564.9)
	507 reritonitis and retroperitoneal infections (includes $567.0, 567.1, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.2, 567.$
	50/.2, 50/.21, 50/.22, 50/.23, 50/.29, 50/.3, 50/.31, 50/.38, 50/.39,
	50/.8, 50/.81, 50/.82, 50/.89, 56/.9)
	568 Other disorders of peritoneum (includes $568.0, 568.8, 568.81, 568.82, 568.80, 568.81, 568.82, 568.81, 568.82, 568.81, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82, 568.82,$
	508.82, 508.89, 568.9)
	569 Other disorders of intestine (includes 569.0, 569.1, 569.2, 569.3,
	569.4, 569.41, 569.42, 569.43, 569.44, 569.49, 569.5, 569.6, 569.60,

Category	MSI Physician Billings diagnostic code(s)
	569.61, 569.62, 569.69, 569.7, 569.71, 569.79, 569.8, 569.81, 569.82,
	569.83, 569.84, 569.85, 569.86, 569.87, 569.89, 569.9)
	579.2 Blind loop syndrome (i.e., excessive intestinal bacteria, possible
	result of surgical complication)
	579.3 Other and unspecified postsurgical non-absorption
	579.8 Other specified intestinal malabsorption
	579.9 Unspecified intestinal malabsorption
	680.5 Infections of skin and subcutaneous tissue of the buttock
	(includes: anus, gluteal region)
	783.9 Other symptoms concerning nutrition, metabolism, and
	development (i.e., hypometabolism, dehydration, disorders of
	electrolyte and acid-base imbalance)
	787.1 Heartburn
	787.2 Dysphagia (i.e., difficulty swallowing) (includes 787.20, 787.21,
	787.22, 787.23, 787.24, 787.29)
	787.3 Flatulence, eructation, and gas pain
	787.4 Visible peristalsis
	787.5 Abnormal bowel sounds (includes: absent or hyperactive bowel
	sounds)
	787.6 Incontinence of feces (includes 7876.0, 7876.1, 7876.2, 7876.3)
	787.7 Abnormal feces
	787.9 Other symptoms involving digestive system (includes: diarrhea,
	change in bowel movements, cramping rectal pain) (includes 7879.1,
	7879.9)
	789 Other symptoms involving abdomen and pelvis (All 789 except
	789.7 Colic (i.e., in babies) (includes 789.00,789.01, 789.02, 789.03,
	789.04, 789.05, 789.06, 789.07, 789.09, 789.1, 789.2, 789.30, 789.31,
	789.32, 789.33, 789.34, 789.35, 789.36, 789.37, 789.39, 789.40, 789.41,
	789.42, 789.43, 789.44, 789.45, 789.46, 789.47, 789.49, 7895, 789.60,
	789.61, 789.62, 789.63, 789.64, 789.65, 789.66, 789.67, 789.69, 789.9)
	792.1 Abnormal findings in stool contents
	793.4 Abnormal radiographic or other findings of the gastrointestinal
	tract
	793.6 Abnormal radiographic or other findings of the abdominal area,
	including retroperitoneum
	799.4 Cachexia
	V10.0 Personal history of malignant neoplasm of gastrointestinal tract
	(includes V10.00, V10.04, V10.05, V10.06, V10.09)
	V44.2 Presence of ileostomy
	V44.3 Presence of colostomy
	V44.4 Presence of other artificial opening of gastrointestinal tract
	V45.72 Acquired absence of small/large intestine
	V55.2 Encounter for aftercare or procedures related to ileostomy
	V55.3 Encounter for aftercare or procedures related to colostomy

Category	MSI Physician Billings diagnostic code(s)
	V55.4 Encounter for aftercare or procedures related to other artificial
	opening of digestive tract
	V76.41 Special screening for malignant neoplasms of rectum
	V76.5 Special screening for malignant neoplasms of intestine (includes:
	intestine and colon; excludes: rectum) (includes V76.50, V76.51,
	V76.52)
	E943 Adverse effects related to therapeutic use of agents primarily
	affecting gastrointestinal system (includes: emetics, emollients, anti-
	diarrheal drugs, others) (includes E943.0, E943.1, E943.2, E943.3,
	E943.4, E943.5, E943.6, E943.8, E943.9)
Gynecologic	cancer
Malignant	179 Malignant neoplasm of uterus, part unspecified
neoplasm	180 Malignant neoplasm of cervix uteri (includes 180.0, 180.1, 180.8,
1	180.9)
	181 Malignant neoplasm of placenta
	182 Malignant neoplasm of body of uterus (includes 182.0, 182.1,
	182.8)
	183 Malignant neoplasm of ovary and other uterine adnexa (includes
	183.0, 183.2, 183.3, 183.4, 183.5, 183.8, 183.9)
	184 Malignant neoplasm of other and unspecified female genital organs
	(includes 184.0, 184.1, 184.2, 184.3, 184.4, 184.8, 184.9)
Related and	195.3 Pelvis (includes: groin, inguinal region not otherwise specified,
benign	sacrococcygeal regions, sites overlapping within pelvis such as
neoplasm	rectovaginal (septum) and rectovesical (septum)
	218 Uterine leiomyoma (includes 218.0, 218.1, 218.2, 218.9)
	219 Other benign neoplasm of uterus (includes 219.0, 219.1, 219.8,
	219.9)
	220 Benign neoplasm of ovary
	221 Benign neoplasm of other female genital organs (includes: fallopian
	tubes, vagina, vulva, site NOS) (includes 221.0, 221.1, 221.2, 221.8,
	221.9)
	236.0 Neoplasm of uncertain behavior of uterus
	236.1 Neoplasm of uncertain behavior of placenta
	236.2 Neoplasm of uncertain behavior of ovary
	236.3 Neoplasm of uncertain behavior of other and unspecified female
	genital organs
	239.5 Neoplasm of unspecified nature of other genitourinary organs
Carcinoma	233.1 Cervix uteri
ın sıtu	233.2 Other and unspecified parts of uterus
	233.3 Other and unspecified female genital organs (includes 233.30,
	233.31, 233.32, 233.39)
Other	614 Inflammatory disease of ovary, fallopian tube, pelvic cellular tissue,
related	and peritoneum (includes 614.0, 614.1, 614.2, 614.3, 614.4, 614.5,
	614.6, 614.7, 614.8, 614.9)

Category	MSI Physician Billings diagnostic code(s)
	615 Inflammatory diseases of uterus, except cervix (includes 615.0,
	615.1, 615.9)
	616 Inflammatory disease of cervix, vagina, and vulva (includes 616.0,
	616.10, 616.11, 616.2, 616.3, 616.4, 616.50, 616.51, 616.8, 616.9
	618 Genital prolapsed (includes 618.0, 618.1, 618.2, 618.3, 618.4,
	618.5, 618.6, 618.7, 618.8, 618.9)
	619 Fistula involving female genital tract (includes 619.0, 619.1, 619.2,
	619.8, 619.9)
	620 Noninflammatory disorders of ovary, fallopian tube, and broad
	ligament Includes 620.0, 620.1, 620.2, 620.3, 620.4, 620.5, 620.6,
	620.7, 620.8, 620.9)
	621 Disorders of uterus, not elsewhere classified (includes 621.0, 621.1,
	621.2, 621.3, 621.4, 621.5, 621.6, 621.7, 621.8, 621.9)
	622 Noninflammatory disorders of cervix (622.0, 622.1, 622.2, 622.3,
	622.4, 622.5, 622.6, 622.7, 622.8, 622.9)
	623 Noninflammatory disorders of vagina (includes 623.0, 623.1, 623.2,
	623.3, 623.4, 623.5, 623.6, 623.7, 623.8, 623.9)
	624 Noninflammatory disorders of vulva and perineum (includes 624.0,
	624.1, 624.2, 624.3, 624.4, 624.5, 624.6, 624.8, 624.9)
	625 Pain and other symptoms associated with female genital organs
	(includes 625.0, 625.1, 625.2, 625.3, 625.4, 625.5, 625.6, 625.7, 625.8,
	625.9)
	626 Disorders of menstruation and other abnormal bleeding from female
	genital tract (All 626 except 626.3 Puberty bleeding) (includes 626.0,
	626.1, 626.2, 626.4, 626.5, 626.6, 626.7, 626.8, 626.9)
	627 Menopausal and postmenopausal disorders (includes 627.0, 627.1,
	627.2, 627.3, 627.4, 627.8, 627.9
	628 Infertility, female (includes $628.0, 628.1, 628.2, 628.3, 628.4, 628.0, 628.1, 628.2, 628.3, 628.4, 628.0, 628.1, 628.2, 628.3, 628.4, 628.0, 628.1, 628.2, 628.3, 628.4, 628.2, 628.2, 628.3, 628.4, 628.2, 628.2, 628.3, 628.4, 628.2, 628.2, 628.2, 628.3, 628.4, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2, 628.2,$
	(28.8, 628.9)
	629 Other disorders of female genital organs (All 629 except 629.2
	Female genital mutilation status) (includes $629.0, 629.1, 629.3, 629.8, 620.0)$
	029.9)
	/ 88 Symptoms involving the unnary system (Air / 88 except / 88.0 renar
	$\begin{array}{c} \text{conc} (\text{includes } / 80.1, / 80.20, / 80.21, / 80.29, / 80.30, / 80.31, / 80.32, \\ 788 22 788 24 788 25 788 26 788 27 788 20 788 41 788 42 788 42 \\ \end{array}$
	788 5 788 61 788 62 788 63 788 64 788 65 788 60 788 7 788 8
	788.0)
	705.0 Abnormal Pananicalaou smear of cervix and cervical human
	papillomavirus (includes 795.0)
	795 1 Abnormal Pananicolaou smear of vagina and vaginal human
	napillomavirus (includes 795.1)
	V10.40 Personal history of malignant neonlasm of female genital organ
	unspecified
	V10.41 Personal history of malignant neoplasm of cervix uteri
	V10.42 Personal history of malignant neoplasm of other parts of uterus

Category	MSI Physician Billings diagnostic code(s)		
	V10.43 Personal history of malignant neoplasm of ovary		
	V10.44 Personal history of malignant neoplasm of other female genital		
	organs		
	V45.77 Acquired absence of genital organs (excludes: cervix and uterus,		
	genital mutilation)		
	V47.5 Other genital problems		
	V50.42 Prophylactic removal of ovary		
	V72.3 Gynecological examination		
	V76.2 Special screening for malignant neoplasms of cervix		
	V76.46 Special screening for malignant neoplasms of ovary		
	V76.47 Special screening for malignant neoplasms of vagina (pap smear		
	status-post hysterectomy for non-malignant condition)		
	V88.0 Acquired absence of cervix and uterus		
	E932.2 Adverse effect related to therapeutic use of ovarian hormones		
	and synthetic substitutes (includes: oral contraceptives, estrogens,		
	estrogens and progestogens combined, progestogens)		
Prostate can	Prostate cancer		
Malignant	185 Malignant neoplasm of prostate		
neoplasms			
Related and	195.3 Pelvis (includes: groin, inguinal region not otherwise specified,		
benign	sacrococcygeal regions, sites overlapping within pelvis such as		
neoplasm	rectovaginal (septum) and rectovesical (septum)		
	222.2 Benign neoplasm of prostate		
	236.5 Neoplasm of uncertain behavior of prostate		
	239.5 Neoplasm of unspecified nature of other genitourinary organs		
Carcinoma	233.4 Prostate		
in situ			
Other	600 Hyperplasia of prostate		
related	601 Inflammatory diseases of prostate (includes 601.0, 601.1, 601.2,		
	601.3, 601.4, 601.8, 601.9)		
	602 Other disorders of prostate (includes 602.0, 602.1, 602.2, 602.3,		
	602.8, 602.9)		
	603 Hydrocele (i.e., accumulation of fluid in the spermatic cord, testis,		
	or tunica vaginalis) (includes $603.0, 603.1, 603.8, 603.9$)		
	604 Orchitis and epididymitis (i.e., inflammation of the testicles or		
	epididymis) (includes 604.0, 604.90, 604.91, 604.99)		
	(07 Discussions of the maximum (includes 606.0, 606.1, 606.8, 606.9))		
	607 Disorders of the penis (includes $607.0, 607.1, 607.2, 607.3, 607.81, 607.82, 607.82, 607.84, 607.85, 607.80, 607.0)$		
	00/.82, 00/.83, 00/.84, 00/.83, 00/.89, 00/.9)		
	608 Other disorders of the male genital organs (includes $608.0, 608.1, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.2, 608.$		
	000.20, 000.21, 000.22, 000.23, 000.24, 000.5, 000.4, 000.01, 000.82, 000 92 600 94 600 95 600 96 600 97 600 90 600 0)		
	000.03, 000.04, 000.03, 000.00, 000.07, 000.09, 000.9)		
	(onlice) (includes 788 1, 788 20, 788 21, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788 20, 788		
	788 33 788 34 788 35 788 36 788 37 788 30 788 41 788 42 788 42		
	/00.33, /00.34, /00.33, /00.30, /00.37, /00.39, /00.41, /00.42, /00.43,		

Category	MSI Physician Billings diagnostic code(s)	
	788.5, 788.61, 788.62, 788.63, 788.64, 788.65, 788.69, 788.7, 788.8,	
	788.9)	
	792.2 Abnormal findings in semen	
	V10.45 Personal history of malignant neoplasm of male genital organ,	
	unspecified	
	V10.46 Personal history of malignant neoplasm of prostate	
	V47.4 Other urinary problems	
	V47.5 Other genital problems	
	V76.44 Special screening for malignant neoplasms of prostate	
	E932.1 Adverse effects related to therapeutic use of androgens and	
	anabolic congeners (includes: nandrolone phenpropionate,	
	oxymetholone, testosterone and preparations)	
All cancer sites		
Malignant	199.0 Malignant neoplasm without specification of site, disseminated	
neoplasms	199.1 Malignant neoplasm without specification of site, other	
Related and	229.0 Benign neoplasm of other and unspecified sites, lymph nodes	
benign	234.9 Carcinoma in situ of other and unspecified sites, site unspecified	
neoplasms	(includes: carcinoma in situ not otherwise specified)	
Other	683 Acute lymphadenitis	
related	780.5 Sleep disturbances (includes 780.50, 780.51, 780.52, 780.53,	
	780.54, 780.55, 780.56, 780.57, 780.58, 780.59)	
	783.0 Anorexia	
	783.1 Abnormal weight gain	
	783.2 Abnormal loss of weight and underweight	
	787.0 Nausea and vomiting (includes 787.01, 787.02, 787.03, 787.04)	
	790.0 Abnormality of red blood cells (includes: anemia, hemoglobin	
	disorders, polycythemia)	
	792.9 Other nonspecific abnormal findings in body substances	
	(includes: peritoneal, pleural, synovial, and vaginal fluids)	
	795.4 Other nonspecific abnormal histological findings	
	795.8 Abnormal tumor markers (includes: elevated tumour associated	
	antigens and tumour specific antigens)	
	990 Effects of radiation, unspecified (includes: complication of	
	radiation therapy, radiation sickness)	
	999.3 Other infection related to surgical or medical care (includes:	
	infection due to port-o-cath)	
	V15.3 Personal history of irradiation	
	V41.7 Problems with sexual function	
	V58.0 Encounter for aftercare or procedures related to radiotherapy	
	V58.1 Encounter for aftercare or procedures related to chemotherapy	
	and immunotherapy for neoplastic conditions	
	V67.0 Follow-up examination following surgery	
	V67.1 Follow-up examination following radiotherapy	
	V67.2 Follow-up examination following chemotherapy (i.e., cancer	
	chemotherapy follow-up)	

Category	MSI Physician Billings diagnostic code(s)
	V87.41 Personal history of antineoplastic chemotherapy
	V87.42 Personal history of monoclonal drug therapy
Mental	300.0 Anxiety states (includes 300.00, 300.01, 300.02, 300.09)
health and	300.4 Dysthymic disorder (includes: anxiety depression, depression
related	with anxiety, other types of depression)
	300.5 Neurasthenia (includes: fatigue neurosis, nervous debility,
	psychogenic asthenia, psychogenic general fatigue)
	308 Acute reaction to stress (includes 308.0, 308.1, 308.2, 308.3, 308.4,
	308.9)
	309 Adjustment reaction (includes 309.0, 309.1, 309.21, 309.22, 309.23,
	309.24, 309.28, 309.29, 309.3, 309.4, 309.81, 309.82, 309.83, 309.89,
	309.9)
	311 Depressive disorder, not elsewhere classified (includes: depressive
	disorder not otherwise specified, depressive state not otherwise
	specified, depression)
	799.2 Signs and symptoms involving emotional state (excludes: anxiety
	and depression; includes: nervousness, irritability, impulsiveness,
	emotional lability, demoralization and apathy, other)

MSI= Medical Services Insurance Physicians' Billings.









Cancer Survivor Cohort